

# Local transfer for quasi-split classical groups and congruences mod l

Alberto Mínguez, Vincent Sécherre

### ▶ To cite this version:

Alberto Mínguez, Vincent Sécherre. Local transfer for quasi-split classical groups and congruences mod l. 2024. hal-03979594v2

## HAL Id: hal-03979594 https://hal.science/hal-03979594v2

Preprint submitted on 5 Nov 2024

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

# LOCAL TRANSFER FOR QUASI-SPLIT CLASSICAL GROUPS AND CONGRUENCES MOD $\ell$

by

Alberto Mínguez & Vincent Sécherre

Abstract. — Let G be the group of rational points of a quasi-split p-adic special orthogonal, symplectic or unitary group for some odd prime number p. Following Arthur and Mok, there are an integer  $N \ge 1$ , a p-adic field E and a local functorial transfer from isomorphism classes of irreducible smooth complex representations of G to those of  $\operatorname{GL}_N(E)$ . By fixing a prime number  $\ell$  different from p and an isomorphism between the field of complex numbers and an algebraic closure of the field of  $\ell$ -adic numbers, we obtain a transfer map between representations with  $\ell$ -adic coefficients. Now consider a cuspidal irreducible  $\ell$ -adic representation  $\pi$  of G: we can define its reduction mod  $\ell$ , which is a semi-simple smooth representation of G of finite length, with coefficients in a field of characteristic  $\ell$ . Let  $\pi'$  be a cuspidal irreducible  $\ell$ -adic representation of G whose reduction mod  $\ell$  is isomorphic to that of  $\pi$ . We prove that the transfers of  $\pi$  and  $\pi'$  have reductions mod  $\ell$  which may not be isomorphic, but which have isomorphic supercuspidal supports. When G is not the split special orthogonal group SO<sub>2</sub>, we further prove that the reductions mod  $\ell$  of the transfers of  $\pi$  and  $\pi'$  share a unique common generic component.

2010 Mathematics Subject Classification: 22E50, 11F70 Keywords and Phrases: Automorphic representation, Classical group, Congruences mod  $\ell$ , Cuspidal representation, Functorial transfer

#### 1. Introduction

#### 1.1.

Let F be a p-adic field for some odd prime number p and G be the group of rational points of a quasi-split special orthogonal, unitary or symplectic group defined over F. In the case where G is unitary, let E be the quadratic extension of F with respect to which G is defined; otherwise, let E be equal to F. According to Arthur [2] for special orthogonal and symplectic groups, and to Mok [47] for unitary groups, there is a positive integer N = N(G) and a map from isomorphism classes of irreducible (smooth) complex representations of G to those of the general linear group  $\operatorname{GL}_N(E)$ , called the local *transfer* or *base change*, which we will denote by **t**. 1.2.

Let us fix a prime number  $\ell$  different from p and an isomorphism of fields  $\iota$  between  $\mathbb{C}$  and an algebraic closure  $\overline{\mathbb{Q}}_{\ell}$  of the field of  $\ell$ -adic numbers. Replacing formally  $\mathbb{C}$  by  $\overline{\mathbb{Q}}_{\ell}$  thanks to  $\iota$ , we get a local transfer between isomorphism classes of irreducible smooth  $\overline{\mathbb{Q}}_{\ell}$ -representations, denoted  $\mathbf{t}_{\ell}$ . (We describe the dependency of  $\mathbf{t}_{\ell}$  in the choice of  $\iota$  – or equivalently the behavior of  $\mathbf{t}$  with respect to automorphisms of  $\mathbb{C}$ : see Paragraph 6.4 for unramified representations, and Paragraph 9.2 for discrete series representations, of G.)

We can now consider irreducible  $\overline{\mathbb{Q}}_{\ell}$ -representations which are *integral* – that is, which carry a stable  $\overline{\mathbb{Z}}_{\ell}$ -lattice, where  $\overline{\mathbb{Z}}_{\ell}$  denotes the ring of integers of  $\overline{\mathbb{Q}}_{\ell}$ . Given such a representation  $\pi$ , one can define its reduction mod  $\ell$ : this is the semi-simplification of the reduction of any of its stable  $\overline{\mathbb{Z}}_{\ell}$ -lattices modulo the maximal ideal of  $\overline{\mathbb{Z}}_{\ell}$ . This is a smooth representation of finite length with coefficients in  $\overline{\mathbb{F}}_{\ell}$ , the residue field of  $\overline{\mathbb{Z}}_{\ell}$ , denoted  $\mathbf{r}_{\ell}(\pi)$ . One then can ask whether the map  $\mathbf{t}_{\ell}$  preserves the fact of being integral, and how it behaves with respect to congruences mod  $\ell$ .

Similar questions have already been answered for other local correspondences: see [66, 16, 12] for the local Langlands correspondence for  $GL_n$ , as well as [17, 41] for the local Jacquet-Langlands correspondence between inner forms of  $GL_n$ , for  $n \ge 1$ . (See also Paragraph 1.7 below and Appendix A, where we discuss the case of the cyclic local base change for  $GL_n$ .) In this paper, we prove the following theorem.

**Theorem 1.1.** — Let  $\pi_1$ ,  $\pi_2$  be integral cuspidal irreducible  $\overline{\mathbb{Q}}_{\ell}$ -representations of G, and assume that

(1.1) 
$$\mathbf{r}_{\ell}(\pi_1) \leqslant \mathbf{r}_{\ell}(\pi_2)$$

that is,  $\mathbf{r}_{\ell}(\pi_1)$  is contained in  $\mathbf{r}_{\ell}(\pi_2)$  as semi-simple  $\overline{\mathbb{F}}_{\ell}$ -representations of G. Then

(1) The local transfer  $\mathbf{t}_{\ell}(\pi_i)$  is an integral  $\overline{\mathbb{Q}}_{\ell}$ -representation of  $\mathrm{GL}_N(E)$  for each i = 1, 2.

(2) The irreducible components of the semi-simple  $\overline{\mathbb{F}}_{\ell}$ -representation  $\mathbf{r}_{\ell}(\mathbf{t}_{\ell}(\pi_1)) \oplus \mathbf{r}_{\ell}(\mathbf{t}_{\ell}(\pi_2))$  all have the same supercuspidal support (see below for a definition).

(3) Assume that G is not isomorphic to the split special orthogonal group  $SO_2(F) \simeq F^{\times}$ . The semi-simple  $\overline{\mathbb{F}}_{\ell}$ -representations  $\mathbf{r}_{\ell}(\mathbf{t}_{\ell}(\pi_1))$  and  $\mathbf{r}_{\ell}(\mathbf{t}_{\ell}(\pi_2))$  have a unique generic irreducible component in common.

As in the case of complex coefficients, an irreducible representation of  $\operatorname{GL}_N(E)$  on an  $\overline{\mathbb{F}}_{\ell}$ -vector space V is said to be *generic* if V carries a non-zero  $\overline{\mathbb{F}}_{\ell}$ -linear form  $\Lambda$  such that  $\Lambda(\pi(u)v) = \theta(u)v$ for all  $v \in V$  and all upper triangular unipotent matrices u of  $\operatorname{GL}_N(E)$ , where  $\theta$  is the  $\overline{\mathbb{F}}_{\ell}$ -character

$$u \mapsto \psi(u_{1,2} + \dots + u_{n-1,n})$$

and  $\psi$  is a non-trivial  $\overline{\mathbb{F}}_{\ell}$ -character of F.

An irreducible  $\overline{\mathbb{F}}_{\ell}$ -representation of  $\operatorname{GL}_N(E)$  is supercuspidal if it does not occur as a subquotient of any representation parabolically induced from a proper Levi subgroup. A supercuspidal support of an irreducible  $\overline{\mathbb{F}}_{\ell}$ -representation  $\pi$  of  $\operatorname{GL}_N(E)$  is a pair  $(M, \rho)$  made of a Levi subgroup of  $\operatorname{GL}_N(E)$  and a supercuspidal representation  $\rho$  of M such that  $\pi$  occurs as a subquotient of the normalized parabolic induction of  $\rho$ . It is uniquely determined up to conjugacy ([67] V.4, [39] Théorème 8.16).

 $\mathbf{2}$ 

Note that, unless G is the split special orthogonal group  $SO_2(F) \simeq F^{\times}$ , the centre of G is compact. When this is the case, any cuspidal irreducible  $\overline{\mathbb{Q}}_{\ell}$ -representation of G is integral. We will discuss the case of the split  $SO_2(F)$  in detail in Paragraph 9.3.

Also note that, if G is not isomorphic to  $SO_2(F) \simeq F^{\times}$ , then (3) implies (2), since all irreducible components of the reduction mod  $\ell$  of an integral irreducible  $\overline{\mathbb{Q}}_{\ell}$ -representation of  $GL_N(E)$  have the same supercuspidal support [66] §1.3.

Before discussing the other assumptions of Theorem 1.1 (in Paragraph 1.6), let us explain how we prove it. The general strategy goes back to Khare [**32**] and Vignéras [**66**] who study the congruence properties of the local Langlands correspondence for  $\operatorname{GL}_n(F)$  with  $n \ge 1$ .

#### 1.3.

The first step is to pass from our given local situation to the following global situation (which is the purpose of Sections 2 to 4).

First, k is a totally real number field, l is either k or a totally imaginary quadratic extension of k and w is a finite place of k above p, inert in l, such that  $k_w = F$  and  $l_w = E$  (see §2.4).

Next,  ${\bf G}$  is a connected reductive group defined over k such that

(1) the group  $\mathbf{G}(F)$  naturally identifies with G,

(2) the group  $\mathbf{G}(k_v)$  is compact for any real place v and quasi-split for any finite place v,

(3) the k-group **G** is an inner form of a quasi-split special orthogonal, unitary or symplectic group  $\mathbf{G}^*$ .

The existence of such a group  $\mathbf{G}$  is proved in Section 2 (see Theorem 2.1).

Finally,  $\Pi_1$  and  $\Pi_2$  are irreducible automorphic representations of  $\mathbf{G}(\mathbb{A}_k)$ , where  $\mathbb{A}_k$  denotes the ring of adèles of k, such that

(1)  $\Pi_{1,w} \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_{\ell}$  is isomorphic to  $\pi_1$  and  $\Pi_{2,w} \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_{\ell}$  is isomorphic to  $\pi_2$ ,

(2) the representations  $\Pi_{1,v}$  and  $\Pi_{2,v}$  are trivial for any real place v,

(3) for a given finite place  $u \neq w$  of k, the representations  $\Pi_{1,u}$  and  $\Pi_{2,u}$  are both isomorphic to some cuspidal irreducible unitary representation  $\rho$  of  $\mathbf{G}(k_u)$  which is compactly induced from a compact mod centre, open subgroup of  $\mathbf{G}(k_u)$ ,

(4) there is a finite set S of places of k, containing all real places, such that for all  $v \notin S$ :

(a) the group **G** is unramified over  $k_v$ ,

(b) the representations  $\Pi_{1,v} \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_{\ell}$  and  $\Pi_{2,v} \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_{\ell}$  are unramified with respect to some hyperspecial maximal compact subgroup of  $\mathbf{G}(k_v)$ ,

(c) their Satake parameters (in the sense of Paragraph 3.4) are integral and congruent mod the maximal ideal of  $\overline{\mathbb{Z}}_{\ell}$ ,

where all tensor products are taken with respect to  $\iota$ . We construct such  $\Pi_1$  and  $\Pi_2$  in Sections 3 and 4.

#### 1.4.

The next step – which is the purpose of Section 5 – is to associate to  $\Pi_1$  and  $\Pi_2$  two cuspidal irreducible automorphic representations  $\Pi_1$  and  $\Pi_2$  of  $\operatorname{GL}_N(\mathbb{A}_l)$  such that, for any finite place v, the local transfer of  $\Pi_{i,v}$  is isomorphic to  $\Pi_{i,v}$ , for i = 1, 2. For this, we use the results of Taïbi [63] if  $\mathbf{G}^*$  is symplectic or special orthogonal, and Labesse [36] if  $\mathbf{G}^*$  is unitary. Namely, let  $\Pi_i$  be

- the Arthur parameter associated with  $\Pi_i$  if  $\mathbf{G}^*$  is symplectic or special othogonal ([63]),
- the stable base change of  $\Pi_i$  to  $\operatorname{GL}_N(\mathbb{A}_l)$  if  $\mathbf{G}^*$  is unitary ([36]).

In both cases,  $\widetilde{\Pi}_i$  is algebraic regular and [**63**, **36**] provide  $\widetilde{\Pi}_i$  with certain local-global compatibilities at all finite places. In order to ensure that these local-global compatibilities are what we want, namely, that the local transfer of  $\Pi_{i,v}$  is isomorphic to  $\widetilde{\Pi}_{i,v}$  at all finite v, and in prevision of the next step, we need  $\widetilde{\Pi}_i$  to be cuspidal.

In order to choose  $\Pi_1$ ,  $\Pi_2$  so that  $\Pi_1$ ,  $\Pi_2$  are cuspidal, we use the auxiliary cuspidal representation  $\rho$  of Paragraph 1.3. More precisely, we prove the following result (see Lemma 9.1).

**Proposition 1.2.** — Given k, w and **G** as in Paragraph 1.3, the finite place u of k and the representation  $\rho$  of  $\mathbf{G}(k_u)$  can be chosen so that the local transfer of  $\rho$  is cuspidal.

If  $\mathbf{G}^*$  is unitary, it suffices to choose u so that  $\mathbf{G}$  is split over  $k_u$ . In the symplectic and special orthogonal cases, this is the purpose of Appendices B and C. (In particular, the place u has to divide 2 in the symplectic case.)

#### 1.5.

We now have two algebraic regular, cuspidal irreducible automorphic representations  $\Pi_1$  and  $\Pi_2$  of  $\operatorname{GL}_N(\mathbb{A}_l)$  such that, for i = 1, 2 and all finite places v, the transfer of  $\Pi_{i,v}$  is isomorphic to  $\Pi_{i,v}$ . Besides, it follows from the properties of the transfer from  $\mathbf{G}(\mathbb{A}_k)$  to  $\operatorname{GL}_N(\mathbb{A}_l)$  that the conjugate of the contragredient of  $\Pi_i$  by the generator c of  $\operatorname{Gal}(l/k)$  is isomorphic to  $\Pi_i$ .

From the properties of  $\Pi_1$  and  $\Pi_2$  at all places  $v \notin S$ , and from the congruence properties of the unramified local transfer that we establish in Section 6, it follows that, for all  $v \notin S$ :

(1) the local components  $\widetilde{\Pi}_{1,v}$  and  $\widetilde{\Pi}_{2,v}$  are unramified,

(2) the Satake parameters of  $\widetilde{\Pi}_{1,v} \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_{\ell}$  and  $\widetilde{\Pi}_{2,v} \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_{\ell}$  are integral and congruent mod the maximal ideal of  $\overline{\mathbb{Z}}_{\ell}$ .

We now apply the results of [5], which give us two continuous  $\ell$ -adic Galois representations

$$\Sigma_i : \operatorname{Gal}(\overline{\mathbb{Q}}/l) \to \operatorname{GL}_N(\overline{\mathbb{Q}}_\ell), \quad i = 1, 2,$$

such that, for any finite place v of l not dividing l, the (l-adic) Weil-Deligne representation associated with  $\widetilde{\Pi}_{i,v} |\det|_v^{(1-N)/2}$  by the local Langlands correspondence is isomorphic to the Frobenius-semisimplification of the Weil-Deligne representation associated with  $\Sigma_{i,v}$ , the restriction of  $\Sigma_i$  to a decomposition subgroup of  $\operatorname{Gal}(\overline{\mathbb{Q}}/l)$  at v. (Here  $|\cdot|_v$  denotes the absolute value of  $l_v$  normalized so that the absolute value of any uniformizer of  $l_v$  is the inverse of the cardinality of the residue field of  $l_v$ .)

Thanks to our local conditions at all  $v \notin S$ , the representations  $\Sigma_{1,v}$ ,  $\Sigma_{2,v}$  are congruent mod  $\ell$ . A density argument then implies that  $\Sigma_1$  and  $\Sigma_2$  are congruent mod  $\ell$ . In particular,  $\Sigma_{1,w}$ ,  $\Sigma_{2,w}$  are congruent mod  $\ell$ .

Associated with  $\Sigma_{i,w}$ , there is a Frobenius-semisimple Weil-Deligne representation  $(\rho_i, N_i)$ . We show in Section 7 that the fact that  $\Sigma_{1,w}$  and  $\Sigma_{2,w}$  are congruent mod  $\ell$  implies that the smooth

semi-simple representations  $\rho_1$  and  $\rho_2$  are integral and congruent mod  $\ell$ . Since  $\rho_i$  corresponds to the cuspidal support of  $\widetilde{\Pi}_{i,w} |\det|_w^{(1-N)/2} \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_{\ell}$  (thanks to the local-global compatibility at w given by [5]), it follows from the mod  $\ell$  local Langlands correspondence of Vignéras [66] that

$$\widetilde{\Pi}_{1,w} |\det|_w^{(1-N)/2} \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_{\ell}, \quad \widetilde{\Pi}_{2,w} |\det|_w^{(1-N)/2} \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_{\ell}$$

are integral and have the same mod  $\ell$  supercuspidal support (which is the supercuspidal support of any irreducible component of the reduction mod  $\ell$ ): it follows that the supercuspidal support of the generic irreducible component of the reduction mod  $\ell$  of  $\widetilde{\Pi}_{i,w} |\det|_w^{(1-N)/2} \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_{\ell}$ , denoted  $\delta_i$ , is independent of  $i \in \{1, 2\}$ . Since a generic irreducible  $\overline{\mathbb{F}}_{\ell}$ -representation is uniquely determined by its supercuspidal support, we deduce that  $\delta_1$ ,  $\delta_2$  are isomorphic. The Main Theorem 1.1 now follows from the fact that  $\widetilde{\Pi}_{i,w} |\det|_w^{(1-N)/2} \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_{\ell}$  is isomorphic to  $\mathbf{t}_{\ell}(\pi_i)$ . We refer to Sections 8 and 9 for more details.

#### 1.6.

Now let us discuss the assumptions of the main theorem.

First, the construction of  $\Pi_1$  does not require  $\pi_1$  to be cuspidal: it would be enough to assume that  $\pi_1 \otimes_{\overline{\mathbb{Q}}_{\ell}} \mathbb{C}$  is a discrete series representation of G (for one, or equivalently any, choice of the field isomorphism  $\iota$ : see Remark 9.3).

However, in order to construct the representation  $\Pi_2$  satisfying our local conditions at all places  $v \notin S$  by the method of Khare–Vignéras, we need  $\pi_2$  to be cuspidal – even more precisely, we need  $\pi_2$  to be compactly induced from some open, compact mod centre subgroup of G, which is true of any cuspidal representation of G, thanks to the work of Stevens [60] since p is odd. Consequently, both the cuspidality of  $\pi_2$  and (1.1) imply that  $\pi_1$  should be cuspidal, as the parabolic restriction functors commute with reduction mod  $\ell$ .

For the same reason, we want the auxiliary representation  $\rho$  of  $\mathbf{G}(k_u)$  to be compactly induced from an open, compact mod centre subgroup.

Moreover, as has been explained in Paragraph 1.4, we also need  $\rho$  to have a cuspidal transfer to  $\operatorname{GL}_N(k_u)$ . This is why the symplectic group requires a special treatment (see Appendix C), since no cuspidal representation of a *p*-adic symplectic group has a cuspidal transfer when *p* is odd, and the work of Stevens [**60**] is not available when p = 2.

On the other hand, we show that part (3) of Theorem 1.1 does not hold in general for noncuspidal representations: Remark 6.4 gives an example of integral unramified irreducible  $\overline{\mathbb{Q}}_{\ell}$ -representations  $\pi_1$  and  $\pi_2$  of SO<sub>5</sub>(F) such that  $\mathbf{r}_{\ell}(\pi_1) = \mathbf{r}_{\ell}(\pi_2)$  but  $\mathbf{r}_{\ell}(\mathbf{t}_{\ell}(\pi_1))$  and  $\mathbf{r}_{\ell}(\mathbf{t}_{\ell}(\pi_2))$  have no irreducible component in common.

Finally, our Assumption (1.1) is inspired from Vignéras [66] 3.5. It is tempting to conjecturate that the conclusion of Theorem 1.1 still holds when (1.1) is replaced by the weaker condition " $\mathbf{r}_{\ell}(\pi_1)$  and  $\mathbf{r}_{\ell}(\pi_2)$  have a component in common", but we have no evidence that such a conjecture should be true.

#### 1.7.

In Appendix A, we discuss the case of the local base change from  $\operatorname{GL}_n(F)$  to  $\operatorname{GL}_n(K)$  for a cyclic extension K of F, denoted  $\mathbf{b}_{K/F}$ .

As in Paragraph 1.2, choosing a field isomorphism  $\iota : \mathbb{C} \to \overline{\mathbb{Q}}_{\ell}$  gives an  $\ell$ -adic local base change map  $\mathbf{b}_{K/F,\ell}$ . By using the properties of the local Langlands correspondence for  $\mathrm{GL}_n$  with respect to conjugacy by an automorphism of  $\mathbb{C}$ , we prove that  $\mathbf{b}_{K/F,\ell}$  does not depend on the choice of  $\iota$ (see Proposition A.1). We also use certain results of Zou [71] 1.10 to prove an analogue of Theorem 1.1 for  $\mathbf{b}_{K/F,\ell}$  (see Paragraph A.4), and give an example of integral cuspidal  $\overline{\mathbb{Q}}_{\ell}$ -representations  $\pi_1$ ,  $\pi_2$  of  $\mathrm{GL}_2(F)$  such that  $\mathbf{r}_{\ell}(\pi_1) = \mathbf{r}_{\ell}(\pi_2)$  but  $\mathbf{r}_{\ell}(\mathbf{b}_{\ell}(\pi_1)) \neq \mathbf{r}_{\ell}(\mathbf{b}_{\ell}(\pi_2))$  (Paragraph A.5).

#### 1.8.

We thank R. Abdellatif, H. Atobe, R. Beuzart-Plessis, P.-H. Chaudouard, W. T. Gan, H. Grobner, G. Henniart, T. Lanard, E. Lapid, J. Mahnkopf, N. Matringe, A. Moussaoui, D. Prasad, S. W. Shin, S. Stevens, O. Taïbi and H. Yu for stimulating discussions about this work.

We thank Guy Henniart for having accepted to write Appendix C.

This work was partially supported by the Erwin Schrödinger Institute in Vienna, when we benefited from a 2020 Research in Teams grant. We thank the institute for hospitality, and for excellent working conditions.

A. Mínguez thanks the JSPS and the university of Kyoto for hospitality and excellent working conditions when the last part of this work was done. His research was partially funded by the Principal Investigator project PAT4832423 of the Austrian Science Fund (FWF).

V. Sécherre also thanks the Institut Universitaire de France for support and excellent working conditions when this work was done.

#### Notation

Throughout the paper, let p be a prime number, let  $\mathbb{Q}_p$  be the field of p-adic numbers and let  $\overline{\mathbb{Q}}_p$  be an algebraic closure of  $\mathbb{Q}_p$ . By a p-adic field, we mean a finite extension of  $\mathbb{Q}_p$  in  $\overline{\mathbb{Q}}_p$ .

#### 2. Globalizing quadratic and Hermitian forms

The purpose of this section is to prove the following result.

**Theorem 2.1.** — Let F be a p-adic field and let G be a quasi-split special orthogonal, unitary or symplectic group over F. There exist a totally real number field k and a connected reductive group G over k such that

- (1)  $\mathbf{G}$  is an inner form of a quasi-split special orthogonal, unitary or symplectic k-group,
- (2) there is a finite place w of k above p such that  $k_w = F$  and  $\mathbf{G}(F)$  is isomorphic to G,
- (3) the group  $\mathbf{G}(k_v)$  is compact for any real place v, and quasi-split for any finite place v.

This theorem will be used in Section 4 where we prove the existence of automorphic representations of  $\mathbf{G}(\mathbb{A})$  with prescribed conditions on their local components, where  $\mathbb{A}$  denotes the ring of adèles of k.

In Section 9, we will need a stronger version of Theorem 2.1: in order to transfer automorphic representations of  $\mathbf{G}(\mathbb{A})$  to a general linear group, we will need to realize  $\mathbf{G}$  as a pure inner form in the orthogonal and unitary cases, and a rigid inner form in the symplectic case. This is why,

6

rather than Theorem 2.1, we will prove the stronger Theorems 2.8 and 2.11 below. For the symplectic case, see Paragraph 2.8.

We emphasize that p may be equal to 2 in this section.

#### 2.1. Quadratic forms

In this paragraph, k denotes either a p-adic field for some prime number p, or a real Archimedean local field, or a totally real number field, and q denotes a (non-degenerate) quadratic form on a k-vector space of dimension  $d \ge 2$ . There exists a choice of non-zero scalars  $\lambda_1, \ldots, \lambda_d \in k^{\times}$ such that q is equivalent to the quadratic form  $\lambda_1 x_1^2 + \cdots + \lambda_d x_d^2$  on  $k^d$ . The quantity

$$\delta = \delta(q) = \lambda_1 \dots \lambda_d \mod k^{\times 2} \in k^{\times}/k^{\times 2}$$

does not depend on this choice. It is called the discriminant of q. In the sequel, we assume that the discriminant  $\delta$  is fixed. All quadratic forms are assumed to be non-degenerate.

If k is a p-adic field, then q is, up to equivalence, uniquely determined by its Hasse invariant

$$\varepsilon(q) = \prod_{i < j} (\lambda_i, \lambda_j) \in \{-1, 1\}$$

where  $(\cdot, \cdot)$  is the Hilbert symbol over k (see [55] IV.2.3 Theorem 7 or [29] Theorem 9.24).

If k is isomorphic to the field of real numbers, q is, up to equivalence, entirely determined by its signature (a, b) with a + b = d and  $(-1)^b = \delta$ . Its Hasse invariant is equal to  $(-1)^{b(b-1)/2}$ . If  $\delta > 0$ , then b = 2c for some  $c \in \{0, \ldots, |d/2|\}$  and the Hasse invariant is  $(-1)^c$ .

Now suppose that k is a totally real number field, and  $\delta_v > 0$  for all real places v. The Hasse principle (see [54] Theorem 6.6.6) ensures that q is uniquely determined, up to equivalence, by all its localizations  $q_v = q \otimes_k k_v$ , where v ranges over all places of k. In other words, it is determined by the Hasse invariants  $\varepsilon(q_v)$  for all finite v and the signatures  $(d - 2c(q_v), 2c(q_v))$  for all real v. Conversely, a family

$$((\varepsilon_v)_{v \text{ finite}}, (c_v)_{v \text{ real}}), \quad \varepsilon_v \in \{-1, 1\}, \quad c_v \in \{0, \dots, \lfloor d/2 \rfloor\},\$$

corresponds to a (unique) quadratic form of dimension d over k and discriminant  $\delta$  if and only if one has  $\varepsilon_v = 1$  for almost all finite places v and

(2.1) 
$$\prod_{v \text{ finite}} \varepsilon_v \cdot \prod_{v \text{ real}} (-1)^{c_v} = 1$$

(see [54] Theorem 6.6.10). We give more details in  $\S2.2$  and  $\S2.3$ , depending on the parity of d.

#### 2.2. The odd orthogonal case

If k is a p-adic field, there are two equivalence classes of quadratic forms of dimension 2n + 1and discriminant  $\delta$ , in bijection with  $\{-1, 1\}$  through the Hasse invariant. The special orthogonal groups associated with these quadratic forms are non-isomorphic. The one with Hasse invariant

(2.2) 
$$(-1,-1)^{n(n+1)/2} \cdot (-1,\delta)^n$$

(that is  $x_1x_2 + \cdots + x_{2n-1}x_{2n} + (-1)^n \delta x_{2n+1}^2$ ) is split. The other one is non-quasi-split.

If k is isomorphic to the field of real numbers, there are n + 1 equivalence classes of quadratic forms of dimension 2n + 1 and discriminant  $\delta$ . The special orthogonal groups associated with these quadratic forms are non-isomorphic. Exactly one of them is compact: this is the one with signature (2n + 1, 0) if  $\delta > 0$ , and (0, 2n + 1) if  $\delta < 0$ .

**Proposition 2.2.** — Let k be a totally real number field of degree r, and  $\delta \in k^{\times}/k^{\times 2}$ . Suppose that  $\delta_v > 0$  for all real places v. There is a quadratic form q of dimension 2n + 1 and discriminant  $\delta$  such that SO(q) is compact at all real places and quasi-split at all finite places if and only if rn(n + 1)/2 is even. When this is the case, q is unique up to equivalence.

*Proof.* — A quadratic form q over k of dimension 2n + 1 and discriminant  $\delta$  is entirely determined, up to equivalence, by the Hasse invariants  $\varepsilon(q_v) \in \{-1, 1\}$  for all finite places v and the signatures  $(2n + 1 - 2c(q_v), 2c(q_v))$  for all real places v of k. Non-equivalent quadratic forms define non-isomorphic special orthogonal groups.

For SO(q) to be compact at all real places and quasi-split at all finite places, q must have invariants  $c_v = 0$  for all real v and  $\varepsilon_v = (-1, -1)_v^{n(n+1)/2} \cdot (-1, \delta_v)_v^n$  for all finite v, where  $(\cdot, \cdot)_v$  is the Hilbert symbol with respect to  $k_v$ . By (2.1), such a q exists if and only if

$$\prod_{v \text{ finite}} (-1, -1)_v^{n(n+1)/2} \times \prod_{v \text{ finite}} (-1, \delta_v)_v^n = 1.$$

Thanks to the Hilbert reciprocity law ([49] VII), the left hand side is equal to

$$\prod_{v \text{ real}} (-1, -1)_v^{n(n+1)/2} \times \prod_{v \text{ real}} (-1, \delta_v)_v^n = (-1)^{rn(n+1)/2}$$

(since  $\delta_v > 0$  for all real v), which gives the expected result.

**Remark 2.3.** — Given any k, let q be a quadratic form of dimension 2n + 1 and discriminant 1 over k. Then, for any  $\delta \in k^{\times}$ , the quadratic form  $\delta q$  has discriminant  $\delta$  and  $SO(\delta q) = SO(q)$ .

#### 2.3. The even orthogonal case

In this paragraph, we assume that the dimension of q is 2n. It will be convenient to use the normalized discriminant  $\alpha = (-1)^n \delta$ .

Suppose first that k is a p-adic field.

• If n = 1, there is only one equivalence class of quadratic forms of dimension 2 and normalized discriminant  $\alpha = 1$ . Its Hasse invariant is 1. The special orthogonal group associated with it is isomorphic to  $GL_1(k)$ .

• Suppose that  $n \ge 2$  or  $\alpha \ne 1$ . There are two equivalence classes of quadratic forms of dimension 2n and discriminant  $\delta$ , characterized by their Hasse invariant. The special orthogonal groups associated with them are non-isomorphic if and only if  $\alpha = 1$ . When this is the case, the one with Hasse invariant  $(-1, -1)^{n(n-1)/2}$  (that is the quadratic form  $x_1x_2 + \cdots + x_{2n-1}x_{2n}$ ) is split, and the other one is non-quasi-split. Otherwise, let l be the quadratic extension of k generated by a square root of  $\alpha$ : if q is a quadratic form of dimension 2n and discriminant  $\delta$  over k, then  $\lambda q$  has same discriminant and opposite Hasse invariant for any scalar  $\lambda \in k^{\times}$  which is not an l/k-norm, and  $SO(\lambda q) = SO(q)$ .

If k is isomorphic to the field of real numbers, there are n + 1 equivalence classes of quadratic forms of dimension 2n and discriminant  $\delta$ . Quadratic forms with signatures (a, b) and (a', b')define isomorphic special orthogonal groups if and only if one has  $b' \in \{a, b\}$ . If  $\delta < 0$ , there is no compact special orthogonal group. If  $\delta > 0$ , there is exactly one compact special orthogonal group: this is the one with  $b \in \{0, 2n\}$ .

**Proposition 2.4.** — Let k be a totally real number field of degree r, and  $\delta \in k^{\times}/k^{\times 2}$ . Suppose that  $\delta_v > 0$  for all real places v.

(1) There is a quadratic form q of dimension 2n and discriminant  $\delta$  such that SO(q) is compact at all real places and quasi-split at all finite places if and only if either n is odd, or  $\delta \neq (-1)^n$ , or rn(n-1)/2 is even.

(2) Assume that  $\delta \neq (-1)^n$ . For any finite place w such that  $\delta_w \neq (-1)^n$  and any  $\varepsilon \in \{-1, 1\}$ , there is a quadratic form q as in (1) satisfying the extra condition  $\varepsilon(q \otimes k_w) = \varepsilon$ .

Proof. — A quadratic form q over k of dimension 2n and discriminant  $\delta$  is entirely determined, up to equivalence, by the Hasse invariants  $\varepsilon(q_v) \in \{-1, 1\}$  for all finite places v and the signatures  $(2n - 2c(q_v), 2c(q_v))$  for all real places v. A quadratic form f with same dimension and discriminant as q defines a special orthogonal group isomorphic to SO(q) if and only if they have the same Hasse invariants for all finite v such that  $\alpha_v = 1$ , and  $c(f_v) \in \{n - c(q_v), c(q_v)\}$  for all real places v.

For SO(q) to be compact at all real places and quasi-split at all finite places, q must have invariants  $c_v \in \{0, n\}$  for all real places v and  $\varepsilon_v = (-1, -1)_v^{n(n-1)/2}$  for all finite places v such that  $\alpha_v = 1$ . (Recall that  $\alpha = (-1)^n \delta$ .) By (2.1), such a q exists if and only if

$$\prod_{\substack{v \text{ finite} \\ \alpha_v \neq 1}} \varepsilon_v \times \prod_{\substack{v \text{ finite} \\ \alpha_v = 1}} (-1, -1)_v^{n(n-1)/2} \times (-1)^{ns} = 1$$

where s is the number of real places such that  $c_v = n$ . If n is odd, we may adjust  $s \in \{0, \ldots, r\}$  so that this product is 1. If  $\alpha \neq 1$ , we may adjust the signs  $\varepsilon_v$  for the finite v such that  $\alpha_v \neq 1$  so that this product is 1. (Since the number of such v is at least 2, we may even assume that  $\varepsilon_w$  is equal to a given sign  $\varepsilon$  for a given w as in (2).) If n is even and  $\alpha = 1$ , the condition is

$$\prod_{v \text{ finite}} (-1, -1)_v^{n(n-1)/2} = 1$$

and Hilbert's reciprocity law says that the left hand side is equal to

$$\prod_{v \text{ real}} (-1, -1)_v^{n(n-1)/2} = (-1)^{rn(n-1)/2},$$

which gives the expected result.

#### 2.4. Globalizing the base field

The following lemma will be useful in the remainder of this section.

Lemma 2.5. — Let F be a p-adic field.

(1) There exists a totally real number field k of even degree such that  $k_w = F$  for some finite place w of k dividing p.

(2) If  $p \neq 2$ , we may further assume that there exists a finite place u of k such that  $k_u \simeq \mathbb{Q}_2$ .

*Proof.* — We follow the proof of [2] Lemma 6.2.1. Let us write  $F = \mathbb{Q}_p(\beta)$  for some root  $\beta \in F$  of a monic irreducible polynomial f of degree  $r = [F : \mathbb{Q}_p]$  with coefficients in  $\mathbb{Q}_p$ . Given a field E, we identify the space of monic polynomials of degree r with coefficients in E with  $E^r$ .

By Krasner's lemma (see [51] 3.1.5), there is an open neighborhood  $U_p$  of f in  $\mathbb{Q}_p^r$  such that any  $g \in U_p$  has a root  $\beta' \in \overline{\mathbb{Q}}_p$  such that  $\mathbb{Q}_p(\beta') = F$ . Let  $U_\infty$  be the open subset of  $\mathbb{R}^r$  made of all monic polynomials with r distinct real roots. Since the diagonal image of  $\mathbb{Q}^r$  in  $\mathbb{R}^r \times \mathbb{Q}_p^r$  is dense, the intersection

$$\mathbb{Q}^r \cap (U_\infty \times U_p)$$

is non-empty. We may replace f by a polynomial in this intersection, which we still denote by f. The number field  $k = \mathbb{Q}(\beta)$  is totally real, and  $k_w = F$  for some finite place w of k dividing p. If the degree of k is even, we are done. Otherwise, we choose a monic irreducible polynomial g of degree 2 over  $\mathbb{Q}$  which splits over  $\mathbb{R}$  and  $\mathbb{Q}_p$ , whose existence can be proven in the same way as above. Then replace k by  $k(\gamma)$  where  $\gamma$  is a root of g in  $\mathbb{Q}_p$ .

Suppose now that  $p \neq 2$ , and let  $U_2$  be the open subset of  $\mathbb{Q}_2^r$  made of all monic polynomials with r distinct roots in  $\mathbb{Q}_2$ . We may replace f by a polynomial in  $\mathbb{Q}^r \cap (U_\infty \times U_p \times U_2)$ , which we still denote by f. The number field  $k = \mathbb{Q}(\beta)$  is totally real,  $k_w = F$  for some finite place wof k dividing p, and 2 is totally split in k. If the degree of k is even, we are done. Otherwise, we choose a monic irreducible polynomial g of degree 2 over  $\mathbb{Q}$  which splits over  $\mathbb{R}$ ,  $\mathbb{Q}_p$  and  $\mathbb{Q}_2$ , then replace k by  $k(\gamma)$  where  $\gamma$  is a root of g in  $\mathbb{Q}_p$ .

**Remark 2.6.** — With a similar argument, one can prove in addition to part (1) of Lemma 2.5 that, if E is a quadratic extension of F in  $\overline{\mathbb{Q}}_p$ , there is a totally imaginary quadratic extension l of k such that  $l_w = E$ .

**Remark 2.7**. — Part (2) of Lemma 2.5 will be needed in Section 9, in the symplectic case, in order to apply the results of Appendix C.

#### 2.5. Proof of Theorem 2.1 in the special orthogonal case

We prove Theorem 2.1 in the case where G is special orthogonal, that is, there is a quadratic form Q over F such that G is isomorphic to SO(Q). We will prove the following stronger result.

**Theorem 2.8**. — Let Q be a quadratic form over F such that SO(Q) is quasi-split. There exist a totally real number field k and a quadratic form q over k such that

- (1) there is a finite place w of k dividing p such that
  - (a) the field  $k_w$  is equal to F,
  - (b) the quadratic forms  $q \otimes F$  and Q are equivalent,
- (2) the group  $SO(q \otimes k_v)$  is compact for all real v, and quasi-split for all finite v.

*Proof.* — By Lemma 2.5, there exists a totally real number field k of even degree such that  $k_w$ and F are equal for some finite place w of k dividing p. Fix a  $\gamma \in F^{\times}$  such that the discriminant of Q is  $\gamma F^{\times 2}$ , and fix a  $\delta \in k^{\times}$  such that  $\gamma^{-1} \delta_w \in F^{\times 2}$  and  $\delta_v > 0$  for all real v.

By Proposition 2.2 when Q has odd dimension and Proposition 2.4 when Q has even dimension, there is a quadratic form q of discriminant  $\delta$  satisfying (2). Moreover, the quadratic forms  $q \otimes F$  and Q have the same discriminant and define quasi-split special orthogonal groups.

If Q has odd dimension, or if Q has dimension 2n and  $\gamma = (-1)^n$ , they are thus equivalent.

Otherwise, use Proposition 2.4(2) with  $\varepsilon = \varepsilon(Q)$  to ensure that  $q \otimes F$  and Q have the same Hasse invariant: they are thus equivalent. 

**Remark 2.9**. — In addition to Theorem 2.8, there is always a finite place  $u \neq w$  of k such that the group  $SO(q \otimes k_u)$  is split: one can choose

- (1) any finite place different from w in the odd orthogonal case,
- (2) any finite place  $u \neq w$  such that  $(-1)^n \delta_u \in k_u^{\times 2}$  in the even orthogonal case.

#### 2.6. Hermitian forms

In this paragraph, l is a separable quadratic k-algebra (where k is as in Paragraph 2.1) and h is a (non-degenerate) l/k-Hermitian form on an *l*-vector space of dimension  $n \ge 1$ . There exists a choice of non-zero scalars  $\lambda_1, \ldots, \lambda_n \in k^{\times}$  such that h is equivalent to the l/k-Hermitian form  $\lambda_1 N_{l/k}(x_1) + \cdots + \lambda_n N_{l/k}(x_n)$  on  $l^n$ . The quantity

$$\delta = \delta(h) = \lambda_1 \dots \lambda_n \mod \mathcal{N}_{l/k}(l^{\times}) \in k^{\times}/\mathcal{N}_{l/k}(l^{\times})$$

does not depend on this choice. It is called the discriminant of h. Fix an  $\alpha \in k^{\times}$  such that l is isomorphic to the k-algebra  $k[X]/(X^2 - \alpha)$ . The image of  $\alpha$  in  $k^{\times}/k^{\times 2}$  will still be denoted  $\alpha$ .

Up to equivalence, h is uniquely determined by its trace form t, that is, the quadratic form of dimension 2n over k obtained by seeing  $l^n$  as a k-vector space ([54] Theorem 10.1.1).

If l is split, that is, if  $l \simeq k \times k$ , then  $N_{l/k}(l^{\times}) = k^{\times}$  and we may choose  $\alpha = 1$ . There is, up to equivalence, a unique l/k-Hermitian form of dimension n. Its discriminant is trivial, and the unitary group associated with it is (non-canonically) isomorphic to  $GL_n(k)$ . More precisely, if one fixes an isomorphism  $l \simeq k \times k$  of k-algebras, h identifies with a non-degenerate bilinear form on  $k^n \times k^n$ , the group  $\operatorname{GL}_n(l)$  identifies with  $\operatorname{GL}_n(k) \times \operatorname{GL}_n(k)$  and there is an isomorphism

(2.3) 
$$\operatorname{GL}_n(k) \simeq \operatorname{U}(h)$$
$$g \mapsto (g, g^*)$$

where  $g^*$  is the contragredient of  $g \in \operatorname{GL}_n(k)$  with respect to h. (Note that changing the isomorphism  $l \simeq k \times k$  has the effect of exchanging q and  $q^*$  in (2.3).) Also, the trace form t of h is maximally isotropic, that is, it is the sum of n hyperbolic planes.

If l is a quadratic extension of k, a quadratic form of dimension 2n over k is the trace form of an l/k-Hermitian form if and only if  $q \otimes_k l$  is maximally isotropic ([54] Theorem 10.1.2).

If l/k is a quadratic extension of p-adic fields, there are two equivalence classes of l/k-Hermitian forms of dimension n, in bijection with  $k^{\times}/N_{l/k}(l^{\times})$  through the discriminant.

• If n is odd, the unitary groups associated with these Hermitian forms are isomorphic. More precisely, if  $\alpha \neq 1$  and h is a Hermitian form of odd dimension over k, then  $\delta h$  is unequivalent to h for any  $\delta \in k^{\times}$  such that  $\delta \notin N_{l/k}(l^{\times})$ , and the group  $U(\delta h) = U(h)$  is quasi-split.

• If n is even, the unitary group corresponding to the discriminant  $(-1)^{n/2}$  is quasi-split, and the other one is non-quasi-split. The trace form t of h has discriminant  $(-\alpha)^n$  and Hasse invariant

(2.4) 
$$\varepsilon(t) = (\alpha, \delta) \cdot (-\alpha, -1)^{n(n-1)/2}.$$

If l/k is isomorphic to  $\mathbb{C}/\mathbb{R}$ , the Hermitian form h is uniquely determined, up to equivalence, by its signature (a, b) with a + b = n. Its discriminant is  $(-1)^b$ . Its trace form t has discriminant 1 and signature (2a, 2b). The unitary group U(h) is compact if and only if  $b \in \{0, n\}$ .

If l is a totally imaginary quadratic extension of a totally real number field k (thus  $\alpha_v < 0$  for all real places v of k), then h is uniquely determined, up to equivalence, by any one of the following data:

- (1) the equivalence class of its trace form t,
- (2) the Hasse invariants  $\varepsilon(t_v)$  for all finite v and the integers  $b(t_v)$  for all real v,
- (3) the equivalence classes of its localizations  $h_v = h \otimes_k k_v$  for all v,
- (4) the discriminants  $\delta(h_v)$  for all finite v and the integers  $b(h_v)$  for all real v.

We have just seen that (3) and (4) are equivalent, and we have seen that (1) and (2) are equivalent in Paragraph 2.1. Now the fact that (2) and (3) are equivalent follows form the formulas

 $\varepsilon(t_v) = (\alpha_v, \delta(h_v))_v \cdot (-\alpha_v, -1)_v^{n(n-1)/2} \text{ for finite } v, \quad b(t_v) = 2b(h_v) \text{ for real } v,$ 

the first one including the case where  $l_v = l \otimes_k k_v$  splits over  $k_v$  (as  $\alpha_v = 1$  in this case) and the fact that, when  $\alpha_v \neq 1$ , the map  $x \mapsto (\alpha_v, x)_v$  is a bijection from  $k_v^{\times}/N_{l_v/k_v}(l_v^{\times})$  to  $\{-1, 1\}$ . Conversely, a family

(2.5) 
$$((\delta_v)_{v \text{ finite}}, (b_v)_{v \text{ real}}), \quad \delta_v \in k_v^{\times} / \mathcal{N}_{l_v/k_v}(l_v^{\times}), \quad b_v \in \{0, \dots, n\},$$

corresponds to an l/k-Hermitian form of dimension n if and only if there exists a  $\delta \in k^{\times}/N_{l/k}(l^{\times})$ such that  $\delta \equiv \delta_v \mod N_{l_v/k_v}(l_v^{\times})$  for all v (where we have put  $\delta_v = (-1)^{b_v}$  at all real places v), and when it is the case such a Hermitian form is unique. Indeed, this is certainly a necessary condition and, when it is satisfied, the family

(2.6) 
$$((\varepsilon_v)_v \text{ finite}, (2b_v)_v \text{ real}), \quad \varepsilon_v = (\alpha_v, \delta_v)_v \cdot (-\alpha_v, -1)_v^{n(n-1)/2},$$

satisfies  $\varepsilon_v = 1$  for almost all finite places v together with

$$\prod_{v \text{ finite}} \varepsilon_v \times \prod_{v \text{ real}} (-1)^{b_v} = \prod_{v \text{ real}} (\alpha_v, \delta_v)_v \times \prod_{v \text{ real}} (-\alpha_v, -1)_v^{n(n-1)/2} \times \prod_{v \text{ real}} (-1)^{b_v}$$
$$= \prod_{v \text{ real}} (-1, -1)_v^{b_v} \times \prod_{v \text{ real}} (-1)^{b_v}$$

which is equal to 1 (thanks to the fact that  $\alpha_v < 0$  and  $\delta_v = (-1)^{b_v}$  for all real v). Thus there is a unique quadratic form of dimension 2n over k and discriminant  $(-\alpha)^n$  with local invariants (2.6). One can verify that it is maximally isotropic over l (as it is maximally isotropic over  $l_v$  for all v). It is thus the trace form of an l/k-Hermitian form of dimension n, as expected.

 $\mathbf{12}$ 

**Proposition 2.10**. — Let k be a totally real number field of degree r and l be a totally imaginary quadratic extension of k.

(1) There is a Hermitian form h of dimension n such that U(h) is compact at all real places.

(2) There is a Hermitian form h of dimension n such that U(h) is compact at all real places and quasi-split at all finite places if and only if either n is odd, or n and rn/2 are both even.

(3) Assume that n is odd. For any finite place w and any  $\varepsilon \in k_w^{\times}/N_{l_w/k_w}(l_w^{\times})$ , there is a Hermitian form h as in (2) satisfying the extra condition  $\delta(h \otimes k_w) = \varepsilon$ .

*Proof.* — Assertion (1) is verified by any Hermitian form h of dimension n over k such that we have  $b(h_v) \in \{0, n\}$  at all real places v.

Assume now that n = 2m for some  $m \ge 1$ . A Hermitian form h of dimension n and discriminant  $\delta$  satisfies (2) if and only if  $b(h_v) \in \{0, n\}$  and  $\delta_v > 0$  for all real v, and  $\delta_v = (-1)^m$  for all finite v. Such a  $\delta \in k^{\times}/N_{l/k}(l^{\times})$  exists if and only if  $\delta_v \in N_{l_v/k_v}(l_v^{\times})$  for almost all finite v, and

$$\prod_{v} (\alpha_v, \delta_v)_v = 1$$

The first condition is satisfied since  $l_v$  is either split over  $k_v$  or an unramified extension of  $k_v$  for almost all finite v. The second condition follows from

$$\prod_{v} (\alpha_{v}, \delta_{v})_{v} = \prod_{v \text{ finite}} (\alpha_{v}, -1)_{v}^{m} = \prod_{v \text{ real}} (\alpha_{v}, -1)_{v}^{m} = (-1)^{rm}$$

thanks to the Hilbert reciprocity law and the fact that  $\alpha_v < 0$  for all real v.

Assume now that n is odd. A Hermitian form h of dimension n and discriminant  $\delta$  satisfies (2) if and only if  $b(h_v) \in \{0, n\}$  at all real places v, and satisfies (3) if and only if  $b(h_v) \in \{0, n\}$  at all real places v and  $\delta \varepsilon^{-1} \in \mathbb{N}_{l_w/k_w}(l_w^{\times})$ . Fix a finite place  $y \neq w$  and a  $\kappa \in l_y^{\times}$ . We claim that such an h exists, with the extra conditions

 $-b(h_v) = 0$  for all real places v,

 $-\delta \in \mathcal{N}_{l_v/k_v}(l_v^{\times})$  for all places  $v \notin \{w, y\}$  and  $\delta \kappa^{-1} \in \mathcal{N}_{l_u/k_u}(l_u^{\times})$ .

Arguing as in the case when n is even, it suffices to choose a  $\kappa \notin N_{l_y/k_y}(l_y^{\times})$ , which is possible as soon as y has been chosen such that  $\alpha_y \neq 1$ .

#### 2.7. Proof of Theorem 2.1 in the unitary case

We prove Theorem 2.1 in the case where G is unitary, that is, there are a quadratic extension E of F and an E/F-Hermitian form H over F such that G is isomorphic to U(H). We will prove the following more precise theorem.

**Theorem 2.11.** — Let H be an E/F-Hermitian form such that U(H) is quasi-split. There exist a totally real number field k, a totally imaginary quadratic extension l of k and an l/k-Hermitian form h such that:

- (1) there is a finite place w of k above p such that
  - (a) one has  $k_w = F$  and  $l_w = E$ ,
  - (b) the Hermitian forms  $h \otimes F$  and H are equivalent,
- (2) the group  $U(h \otimes k_v)$  is compact for all real v, and quasi-split for all finite v.

*Proof.* — By Lemma 2.5 and Remark 2.6, there are a totally real number field k of even degree and a totally imaginary quadratic extension l of k such that  $k_w = F$  and  $l_w = E$  for some finite place w of k dividing p.

By Proposition 2.10, there exists an l/k-Hermitian form h satisfying (2). Moreover, the Hermitian forms  $h \otimes F$  and H define quasi-split unitary groups.

If *H* has even dimension, they are thus equivalent. If *H* has odd dimension, one uses Proposition 2.10(3) with  $\varepsilon = \delta(H)$  to ensure that  $h \otimes F$  and *H* have the same discriminant: they are thus equivalent.

**Remark 2.12.** — In addition to Theorem 2.11, there is always a finite place  $u \neq w$  of k such that  $U(h \otimes k_u)$  is split: it suffices to choose any finite place  $u \neq w$  such that  $l_u \simeq k_u \times k_u$ .

#### 2.8. The symplectic case

We now consider the case where G is a symplectic group, that is, there exists a non-degenerate symplectic form A over F such that  $G = \operatorname{Sp}(A)$ . By Lemma 2.5, there exists a totally real number field k of even degree such that  $k_w = F$  for some finite place w of k dividing p. (Moreover, when p is odd, we may further assume that there is a finite place u of k such that  $k_u \simeq \mathbb{Q}_2$ .) In this case, Theorem 2.1 is given by [**62**] 2.1.1. See also [**63**] Proposition 3.1.2, where the inner form **G** is realized as a rigid inner form of  $\operatorname{Sp}_{2n}$  over k.

#### 3. Congruences of automorphic forms of definite groups

In this section, we fix a prime number  $\ell$ . Let  $\overline{\mathbb{Q}}_{\ell}$  be an algebraic closure of the field of  $\ell$ -adic integers,  $\overline{\mathbb{Z}}_{\ell}$  be its ring of integers and  $\overline{\mathbb{F}}_{\ell}$  be its residue field. We fix a field isomorphism

$$(3.1) \qquad \qquad \iota: \mathbb{C} \to \overline{\mathbb{Q}}_{\ell}$$

and a number field k. We denote by  $\mathbb{A} = \mathbb{A}_f \times \mathbb{A}_\infty$  the ring of adèles of k.

Given a locally compact, totally disconnected group G, an open subgroup K of G, a commutative ring R and a smooth R-representation  $\rho$  of K, we denote by

$$\mathcal{H}_R(G,\rho)$$

the endomorphism R-algebra of the compact induction of  $\rho$  to G, called the Hecke R-algebra of G relative to  $\rho$ . When  $\rho$  is the trivial R-character of K, it naturally identifies with the convolution R-algebra made of K-bi-invariant, compactly supported R-valued functions on G, and we denote it by  $\mathcal{H}_R(G, K)$ .

Let F be a p-adic field for some  $p \neq \ell$ , G be the group of F-rational points of a reductive group defined over F and  $\pi$  be an irreducible (smooth) representation of G on a  $\overline{\mathbb{Q}}_{\ell}$ -vector space V. It is said to be *integral* if V carries a G-stable  $\overline{\mathbb{Z}}_{\ell}$ -lattice. Given such a lattice L, the representation of G on the  $\overline{\mathbb{F}}_{\ell}$ -vector space  $L \otimes \overline{\mathbb{F}}_{\ell}$  (where  $\overline{\mathbb{F}}_{\ell}$  is the residue field of  $\overline{\mathbb{Z}}_{\ell}$ ) is smooth and has finite length, and its semi-simplification does not depend on the choice of L ([68] Theorem 1). This semi-simplification is denoted  $\mathbf{r}_{\ell}(\pi)$ , and called the *reduction mod*  $\ell$  of  $\pi$ . One defines similarly the reduction mod  $\ell$  of an irreducible  $\overline{\mathbb{Q}}_{\ell}$ -representation of a compact, open subgroup of G.

 $\mathbf{14}$ 

#### 3.1.

Let **G** be a connected reductive group defined over k. We assume that **G** is definite, that is, the group  $\mathbf{G}(\mathbb{A}_{\infty})$  is compact. We embed diagonally  $\mathbf{G}(k)$  in  $\mathbf{G}(\mathbb{A}_{f})$  and set

$$\mathbf{Y} = \mathbf{G}(k) \backslash \mathbf{G}(\mathbb{A}_f).$$

The quotient **Y** is compact ([50] §5) and hence **Y**/K is finite for any open subgroup K of  $\mathbf{G}(\mathbb{A}_f)$ .

We denote by  $\mathscr{A}(\mathbf{G})$  the space of functions  $\mathbf{G}(k) \setminus \mathbf{G}(\mathbb{A}) \to \mathbb{C}$  which are square integrable with respect to a Haar measure on  $\mathbf{G}(\mathbb{A})$ . It is endowed with the natural unitary  $\mathbb{C}$ -representation of  $\mathbf{G}(\mathbb{A})$ .

#### 3.2.

Given an open subgroup K of  $\mathbf{G}(\mathbb{A}_f)$ , let  $\operatorname{Alg}(\mathbf{G}, K)$  be the free  $\mathbb{Z}$ -module of finite rank made of all functions  $\mathbf{Y} \to \mathbb{Z}$  which are invariant under right translations by K ([**66**] 3.3). We consider the  $\mathbb{Z}$ -module

$$\operatorname{Alg}(\mathbf{G}) = \mathfrak{C}^{\infty}(\mathbf{Y}, \mathbb{Z}) = \bigcup_{\mathrm{K}} \operatorname{Alg}(\mathbf{G}, \mathrm{K})$$

(where K ranges over all open subgroups of  $\mathbf{G}(\mathbb{A}_f)$ ) of locally constant functions  $\mathbf{Y} \to \mathbb{Z}$ , called trivial-at-infinity algebraic automorphic forms for  $\mathbf{G}$  (see Paragraph 3.3 below). This module is endowed with the natural  $\mathbb{Z}$ -representation of  $\mathbf{G}(\mathbb{A}_f)$ . Given any commutative ring R, we write

$$\operatorname{Alg}_{R}(\mathbf{G}) = \operatorname{Alg}(\mathbf{G}) \otimes_{\mathbb{Z}} R$$
,  $\operatorname{Alg}_{R}(\mathbf{G}, K) = \operatorname{Alg}(\mathbf{G}, K) \otimes_{\mathbb{Z}} R$ .

The natural R-representation of  $\mathbf{G}(\mathbb{A}_f)$  on  $\operatorname{Alg}_{\mathbf{R}}(\mathbf{G})$  is admissible ([66] 3.3.2).

If R is the field  $\overline{\mathbb{Q}}_{\ell}$ , the representation of  $\mathbf{G}(\mathbb{A}_f)$  on  $\operatorname{Alg}_{\overline{\mathbb{Q}}_{\ell}}(\mathbf{G})$  is semi-simple and any of its irreducible components has an  $\mathcal{O}_{\mathrm{E}}$ -structure for some finite extension E of  $\mathbb{Q}_{\ell}$  in  $\overline{\mathbb{Q}}_{\ell}$  ([66] 3.3.2).

#### 3.3.

Let K be an open subgroup of  $\mathbf{G}(\mathbb{A}_f)$  and R be a commutative ring. The Hecke R-algebra of  $\mathbf{G}(\mathbb{A}_f)$  relative to K is the convolution R-algebra

$$\mathcal{H}_{\mathrm{R}}(\mathbf{G},\mathrm{K}) = \mathcal{H}_{\mathrm{R}}(\mathbf{G}(\mathbb{A}_{f}),\mathrm{K})$$

made of K-bi-invariant, compactly supported functions  $\mathbf{G}(\mathbb{A}_f) \to \mathbb{R}$ . It naturally acts on the R-module  $\mathrm{Alg}_{\mathbb{R}}(\mathbf{G}, \mathbb{K})$ .

As G is definite, there is, by [21] Proposition 8.5, an explicit isomorphism

$$\operatorname{Alg}_{\mathbb{C}}(\mathbf{G}, K) \simeq \mathscr{A}(\mathbf{G})^{K \times \mathbf{G}(\mathbb{A}_{\infty})}$$

 $\Theta \leftrightarrow \Pi$ 

of  $\mathcal{H}_{\mathbb{C}}(\mathbf{G}, \mathbf{K})$ -modules (see [21] (8.4) and Proposition 8.3). In particular, there is a bijection

between

– the irreducible subrepresentations  $\Theta$  of  $\operatorname{Alg}_{\mathbb{C}}(\mathbf{G})$  such that the space  $\Theta^{K}$  of K-fixed vectors in  $\Theta$  is non-zero,

- the irreducible automorphic representations  $\Pi = \Pi_f \otimes \Pi_\infty$  of  $\mathbf{G}(\mathbb{A}) = \mathbf{G}(\mathbb{A}_f) \times \mathbf{G}(\mathbb{A}_\infty)$ , that is, the irreducible subrepresentations of  $\mathscr{A}(\mathbf{G})$  such that  $\Pi_\infty$  is trivial and  $\Pi_f^K$  is non-zero.

**3.4**.

Let us fix an irreducible automorphic representation  $\Pi$  of  $\mathbf{G}(\mathbb{A})$  which is trivial on  $\mathbf{G}(\mathbb{A}_{\infty})$ . By (3.2), we can see  $\Pi$  as an irreducible subrepresentation of  $\operatorname{Alg}_{\mathbb{C}}(\mathbf{G})$ , thus as an irreducible factor of  $\operatorname{Alg}_{\overline{\mathbb{O}}_{\epsilon}}(\mathbf{G})$  via the isomorphism  $\iota$  fixed in (3.1).

We fix two finite places w and u of k not dividing  $\ell$  and a finite set S of finite places of k such that:

(1) the set S contains w, u and all finite places above  $\ell$ ,

(2) for any finite place  $v \notin S$ , the group **G** is unramified over  $k_v$ , and the local component  $\Pi_v$  is unramified with respect to some hyperspecial maximal compact subgroup  $K_v$  of  $\mathbf{G}(k_v)$ .

Any finite place  $v \notin S$  thus defines a character

(3.3) 
$$\chi_v : \mathcal{H}_{\overline{\mathbb{Q}}_\ell}(\mathbf{G}(k_v), \mathbf{K}_v) \to \overline{\mathbb{Q}}_\ell$$

which we call the *Satake parameter* of  $\Pi_v$ .

Recall that  $\Pi$  is admissible and has an  $\mathcal{O}_{\mathrm{E}}$ -structure for some finite extension  $\mathrm{E}$  of  $\mathbb{Q}_{\ell}$  in  $\overline{\mathbb{Q}}_{\ell}$ . Let us write  $\Pi = \Pi_{(\ell)} \otimes \Pi^{(\ell)}$ , where  $\Pi_{(\ell)}$  is the tensor product of all  $\Pi_v$  such that v divides  $\ell$  and  $\Pi^{(\ell)}$ is the restricted tensor product of all  $\Pi_v$  such that v is finite and does not divides  $\ell$ . By [**66**] A.3, both  $\Pi_{(\ell)}$  and  $\Pi^{(\ell)}$  have an  $\mathcal{O}_{\mathrm{E}}$ -structure. By applying [**66**] A.4 to  $\Pi^{(\ell)}$ , we get that each  $\Pi_v$ , for  $v \notin \mathrm{S}$  finite, has an  $\mathcal{O}_{\mathrm{E}}$ -structure. Fixing such an  $\mathcal{O}_{\mathrm{E}}$ -structure, the  $\mathcal{O}_{\mathrm{E}}$ -algebra  $\mathcal{H}_{\mathcal{O}_{\mathrm{E}}}(\mathbf{G}(k_v), \mathbf{K}_v)$ acts on it through the character  $\chi_v$ , which thus has values in  $\mathcal{O}_{\mathrm{E}}$ . It follows that the restriction of  $\chi_v$  to  $\mathcal{H}_{\overline{\mathbb{Z}}_{\ell}}(\mathbf{G}(k_v), \mathbf{K}_v)$  has values in  $\overline{\mathbb{Z}}_{\ell}$ .

#### 3.5.

We now make the following assumptions on the representation  $\Pi$  of Paragraph 3.4:

- the local component  $\Pi_w$  is cuspidal, and is compactly induced from an irreducible representation of some compact open subgroup  $K_w$  of  $\mathbf{G}(k_w)$ ,

- the local component  $\Pi_u$  is cuspidal, and is compactly induced from an irreducible representation  $\eta$  of some compact mod centre open subgroup  $K_u$  of  $\mathbf{G}(k_u)$ .

For any finite place  $v \in S$  such that  $v \notin \{u, w\}$ , we fix a compact open subgroup  $K_v$  of  $\mathbf{G}(k_v)$  such that  $\Pi_v$  has a non-zero  $K_v$ -invariant vector. Recall that, for any finite place  $v \notin S$ , we have fixed a hyperspecial maximal compact open subgroup  $K_v$  of  $\mathbf{G}(k_v)$  in Paragraph 3.4. We define

$$\mathbf{K} = \prod_{v \text{ finite}} \mathbf{K}_v.$$

This is an open subgroup of  $\mathbf{G}(\mathbb{A}_f)$ .

Given an irreducible representation  $\kappa$  of  $K_w$ , we define an irreducible representation  $\Lambda = \Lambda(\kappa)$ of K by

$$\Lambda = \bigotimes_{v \text{ finite}} \Lambda_v$$

with  $\Lambda_w = \kappa$ ,  $\Lambda_u = \eta$  and  $\Lambda_v$  is the trivial character of  $K_v$  for  $v \notin \{w, u\}$ .

We denote by  $\operatorname{Alg}_{\overline{\mathbb{Q}}_{\ell}}(\mathbf{G}, \Lambda)$  the subrepresentation of  $\operatorname{Alg}_{\overline{\mathbb{Q}}_{\ell}}(\mathbf{G})$  generated by its  $\Lambda$ -isotypic component, that is, the subrepresentation formed by the irreducible factors  $\Theta$  such that

- the local component  $\Theta_w$  contains  $\kappa$ ,

 $\mathbf{16}$ 

- the local component  $\Theta_u$  contains  $\eta$ ,

- the local component  $\Theta_v$  has a non-zero  $K_v$ -invariant vector for all finite  $v \notin \{w, u\}$ .

This amounts to considering the space

$$\mathbf{V} = \mathbf{V}(\kappa) = \operatorname{Hom}_{\mathrm{K}}\left(\Lambda, \operatorname{Alg}_{\overline{\mathbb{Q}}_{\ell}}(\mathbf{G})\right)$$

seen as a module over the endomorphism  $\overline{\mathbb{Q}}_{\ell}$ -algebra  $\mathcal{H}_{\overline{\mathbb{Q}}_{\ell}}(\mathbf{G}, \Lambda) = \mathcal{H}_{\overline{\mathbb{Q}}_{\ell}}(\mathbf{G}(\mathbb{A}_{f}), \Lambda)$  of the compact induction of  $\Lambda$  from K to  $\mathbf{G}(\mathbb{A}_{f})$ . We have an  $\mathcal{H}_{\overline{\mathbb{Q}}_{\ell}}(\mathbf{G}, \Lambda)$ -module decomposition

(3.4) 
$$\mathbf{V} = \bigoplus_{\Theta} \operatorname{Hom}_{K}(\Lambda, \Theta)$$

where  $\Theta$  ranges over the irreducible factors of  $\operatorname{Alg}_{\overline{\mathbb{Q}}_{\ell}}(\mathbf{G}, \Lambda)$ , and each  $\operatorname{Hom}_{\mathrm{K}}(\Lambda, \Theta)$  is of finite dimension as  $\Theta$  is admissible. By admissibility again, the number of  $\Theta$  contributing to the direct sum of (3.4) is finite.

Denote by X the set of finite places of k. For any non-empty subset T of X and any irreducible factor  $\Theta$  contributing to the right hand side of (3.4), we write

$$\mathbf{K}_{\mathrm{T}} = \prod_{v \in \mathrm{T}} \mathbf{K}_{v}, \quad \Lambda_{\mathrm{T}} = \bigotimes_{v \in \mathrm{T}} \Lambda_{v}, \quad \Theta_{\mathrm{T}} = \bigotimes_{v \in \mathrm{T}} \Theta_{v}.$$

We thus have  $K = K_S \times K_{X\setminus S}$ ,  $\Lambda = \Lambda_S \otimes \Lambda_{X\setminus S}$  and  $\Theta$  is isomorphic to  $\Theta_S \otimes \Theta_{X\setminus S}$ . Accordingly, we have an isomorphism of  $\overline{\mathbb{Q}}_{\ell}$ -algebras

(3.5) 
$$\mathcal{H}_{\overline{\mathbb{Q}}_{\ell}}(\mathbf{G},\Lambda) \simeq \mathcal{H}_{\overline{\mathbb{Q}}_{\ell}}\left(\mathbf{G}(\mathbb{A}_{S}),K_{S}\right) \otimes \mathcal{H}_{\overline{\mathbb{Q}}_{\ell}}\left(\mathbf{G}(\mathbb{A}_{X\setminus S}),\Lambda_{X\setminus S}\right)$$

where  $\mathbb{A}_S$  and  $\mathbb{A}_{X\setminus S}$  have their obvious meaning, and an isomorphism of  $\mathcal{H}_{\overline{\mathbb{O}}_{\ell}}(\Lambda)$ -modules

 $\operatorname{Hom}_{K}(\Lambda, \Theta) \simeq \operatorname{Hom}_{K_{S}}(\Lambda_{S}, \Theta_{S}) \otimes (\Theta_{X \setminus S})^{K_{X \setminus S}}$ 

via (3.5). The factor  $(\Theta_{X\setminus S})^{K_{X\setminus S}}$  has dimension 1 over  $\overline{\mathbb{Q}}_{\ell}$  and  $\mathcal{H}_{\overline{\mathbb{Q}}_{\ell}}(\mathbf{G}(\mathbb{A}_{X\setminus S}), K_{X\setminus S})$  acts on this line via a character denoted  $\chi_{S}(\Theta)$ . Let  $d_{S}(\Theta)$  be the dimension of  $\operatorname{Hom}_{K_{S}}(\Lambda_{S}, \Theta_{S})$ . Denoting by  $\mathbf{V}_{S}$  the restriction of  $\mathbf{V}$  to  $\mathcal{H}_{\overline{\mathbb{Q}}_{\ell}}(\mathbf{G}(\mathbb{A}_{X\setminus S}), K_{X\setminus S})$ , we therefore have an isomorphism

(3.6) 
$$\mathbf{V}_{\mathrm{S}} \simeq \bigoplus_{\Theta} d_{\mathrm{S}}(\Theta) \cdot \chi_{\mathrm{S}}(\Theta)$$

of  $\mathcal{H}_{\overline{\mathbb{Q}}_{\ell}}(\mathbf{G}(\mathbb{A}_{X\setminus S}), K_{X\setminus S})$ -modules.

#### 3.6.

Assume now that  $\eta$  is integral. Fix a K<sub>u</sub>-stable  $\overline{\mathbb{Z}}_{\ell}$ -lattice L<sub> $\eta$ </sub> of  $\eta$  with semi-simple reduction (by [15] Lemma 6.11). Since K<sub>w</sub> is compact,  $\kappa$  is integral. We also fix a K<sub>w</sub>-stable  $\overline{\mathbb{Z}}_{\ell}$ -lattice L<sub> $\kappa$ </sub> of  $\kappa$  with semi-simple reduction. If v is a finite place different from u, w, let L<sub>v</sub> =  $\overline{\mathbb{Z}}_{\ell}$  be the natural lattice of  $\Lambda_v = \overline{\mathbb{Q}}_{\ell}$ . Tensoring these  $\Lambda_v$  altogether, we obtain a K-stable  $\overline{\mathbb{Z}}_{\ell}$ -lattice L<sub> $\Lambda$ </sub> of  $\Lambda$ . Set

$$\mathbf{V}^{\circ} = \operatorname{Hom}_{K} \left( L_{\Lambda}, \operatorname{Alg}_{\overline{\mathbb{Z}}_{\ell}}(\mathbf{G}) \right)$$

It is a module over  $\mathfrak{H}_{\overline{\mathbb{Z}}_{\ell}}(\mathbf{G}(\mathbb{A}_f), \mathcal{L}_{\Lambda})$ . Set  $\overline{\Lambda} = \mathcal{L}_{\Lambda} \otimes_{\overline{\mathbb{Z}}_{\ell}} \overline{\mathbb{F}}_{\ell}$ . By [66] Lemme 3.7.3 (which we can apply since K satisfies [66] (3.5.1) by [66] Lemme 3.8) the  $\overline{\mathbb{Z}}_{\ell}$ -module  $\mathbf{V}^{\circ}$  is a  $\overline{\mathbb{Z}}_{\ell}$ -lattice of  $\mathbf{V}$  and

$$\mathbf{V}^{\circ} \otimes_{\overline{\mathbb{Z}}_{\ell}} \overline{\mathbb{F}}_{\ell} \simeq \operatorname{Hom}_{K} \left( \overline{\Lambda}, \operatorname{Alg}_{\overline{\mathbb{F}}_{\ell}}(\mathbf{G}) \right).$$

We denote the left hand side by  $\overline{\mathbf{V}}$ , and we continue to see it as a module over  $\mathcal{H}_{\overline{\mathbb{Z}}_{\ell}}(\mathbf{G}(\mathbb{A}_f), \mathcal{L}_{\Lambda})$ . Note that  $\overline{\mathbf{V}}$  is semi-simple and depends only on  $\mathbf{r}_{\ell}(\kappa)$ , the reduction mod  $\ell$  of  $\kappa$ .

#### 3.7.

We now assume that  $\mathbf{G}(k_w)$  is isomorphic to a special orthogonal, unitary or symplectic group. We do not assume that it is quasi-split, but we assume that it is not isomorphic to the split special orthogonal group  $\mathrm{SO}_2(k_w) \simeq k_w^{\times}$ . Equivalently, we assume that it has compact centre ([42] §4.2). Consequently, any cuspidal irreducible  $\overline{\mathbb{Q}}_{\ell}$ -representation of  $\mathbf{G}(k_w)$  is integral. In the rest of this section, we will prove the following theorem.

**Theorem 3.1.** — Assume that  $\mathbf{G}(k_w)$  is isomorphic to a special orthogonal, unitary or symplectic group with compact centre, and that w does not divide 2. Let  $\Pi$  be an irreducible automorphic representation of  $\mathbf{G}(\mathbb{A})$  such that

- $\Pi_{\infty}$  is trivial,
- $\Pi_w$  is cuspidal (and integral),

-  $\Pi_u$  is integral and compactly induced from a compact mod centre, open subgroup of  $\mathbf{G}(k_u)$ . Let  $\pi'$  be an (integral) irreducible cuspidal  $\overline{\mathbb{Q}}_{\ell}$ -representation of  $\mathbf{G}(k_w)$  such that

(3.7) 
$$\mathbf{r}_{\ell}(\Pi_w) \leqslant \mathbf{r}_{\ell}(\pi').$$

There is an irreducible automorphic representation  $\Pi'$  of  $\mathbf{G}(\mathbb{A})$  such that

- (1) the Archimedean component  $\Pi'_{\infty}$  is trivial,
- (2) the local component  $\Pi'_w$  is isomorphic to  $\pi'$ ,
- (3) the local components  $\Pi'_u$  and  $\Pi_u$  are isomorphic,

(4) for any finite place  $v \notin S$ , the local component  $\Pi'_v$  is  $K_v$ -unramified, with Satake parameter  $\chi'_v : \mathcal{H}_{\overline{\mathbb{Z}}_\ell}(\mathbf{G}(k_v), K_v) \to \overline{\mathbb{Z}}_\ell$ , and  $\chi_v, \chi'_v$  are congruent mod the maximal ideal of  $\overline{\mathbb{Z}}_\ell$ .

**Remark 3.2.** — Since  $\Pi$  is integral, it automatically follows from [**66**] A.3, A.4 that the representation  $\Pi_u$  is integral. However, for clarity, we added the integrality assumption in the hypotheses of Theorem 3.1.

*Proof.* — We follow the argument of Khare [**32**] and Vignéras [**66**]. We start with following lemma, which we will prove in Paragraph 3.8.

**Lemma 3.3.** — Let p be a prime number different from 2, let F be a p-adic field and G be a special orthogonal, unitary or symplectic group over F. Suppose that G has compact centre, that is, G is not isomorphic to the split special orthogonal group  $SO_2(F) \simeq F^{\times}$ . Let  $\pi$  and  $\pi'$  be (integral) cuspidal  $\overline{\mathbb{Q}}_{\ell}$ -representations of G such that

$$\mathbf{r}_{\ell}(\pi) \leq \mathbf{r}_{\ell}(\pi').$$

There are a compact open subgroup J of G and irreducible  $\overline{\mathbb{Q}}_{\ell}$ -representations  $\tau$  and  $\tau'$  of J such that  $\pi$  is isomorphic to the compact induction of  $\tau$  to G and  $\pi'$  is isomorphic to the compact induction of  $\tau$  to G' and  $\mathbf{r}_{\ell}(\tau) \leq \mathbf{r}_{\ell}(\tau')$ .

18

**Remark 3.4.** — It is known ([60]) that any cuspidal  $\overline{\mathbb{Q}}_{\ell}$ -representation of G is isomorphic to the compact induction of an (irreducible) representation of some compact open subgroup of G. The point here is that one can choose the same compact open subgroup for  $\pi$  and  $\pi'$ .

Applying Lemma 3.3 to the group  $\mathbf{G}(k_w)$  and the cuspidal  $\overline{\mathbb{Q}}_{\ell}$ -representations  $\Pi_w$  and  $\pi'$ , we obtain a compact open subgroup  $K_w$  of  $\mathbf{G}(k_w)$  and irreducible representations  $\tau$  and  $\tau'$  of  $K_w$  such that  $\Pi_w$  is isomorphic to the compact induction of  $\tau$  to  $\mathbf{G}(k_w)$  and  $\pi'$  is isomorphic to the compact induction of  $\tau$  to  $\mathbf{G}(k_w)$  and  $\pi'$  is isomorphic to the compact induction of  $\tau'$  to  $\mathbf{G}(k_w)$ . Since  $K_w$  is compact,  $\tau$  and  $\tau'$  are integral.

Let  $\eta$  be an irreducible representation of a compact mod centre, open subgroup  $K_u$  of  $\mathbf{G}(k_u)$  such that the compact induction of  $\eta$  to  $\mathbf{G}(k_u)$  is isomorphic to  $\Pi_u$ . Since  $\Pi_u$  is integral, the representation  $\eta$  is integral. Thus  $\Pi$  satisfies the conditions of Paragraphs 3.5 and 3.6.

As in Paragraph 3.5, we define  $\Lambda = \Lambda(\tau)$  and  $\mathbf{V} = \mathbf{V}(\tau)$ . Associated with a choice of  $K_w$ -stable  $\overline{\mathbb{Z}}_{\ell}$ -lattice of  $\kappa$  with semi-simple reduction, there are  $\mathbf{V}^{\circ}$  and  $\overline{\mathbf{V}}$ . Similarly, replacing  $\tau$  by  $\tau'$ , we define  $\Lambda', \mathbf{V}', \mathbf{V}'^{\circ}$  and  $\overline{\mathbf{V}}'$ . Recall that  $\overline{\mathbf{V}}$  and  $\overline{\mathbf{V}}'$  are semi-simple. The key point is that the space  $\overline{\mathbf{V}}$  is non-zero and contained in  $\overline{\mathbf{V}}'$ , since we have  $\mathbf{r}_{\ell}(\tau) \leq \mathbf{r}_{\ell}(\tau')$  by Lemma 3.3.

The character  $\chi_{S}(\Pi)$  of  $\mathcal{H}_{\overline{\mathbb{Z}}_{\ell}}(\mathbf{G}(\mathbb{A}_{X\setminus S}), K_{X\setminus S})$  defined by  $\Pi$  occurs in  $\mathbf{V}_{S}^{\circ}$ . By reduction, we get a character  $\overline{\chi}_{S}$  of  $\mathcal{H}_{\overline{\mathbb{F}}_{\ell}}(\mathbf{G}(\mathbb{A}_{X\setminus S}), K_{X\setminus S})$  occurring in  $\overline{\mathbf{V}}_{S}$ , and therefore in  $\overline{\mathbf{V}}_{S}'$ .

By Deligne-Serre's lemma ([19] Lemma 6.11), there is a character  $\chi'_S$  of  $\mathcal{H}_{\overline{\mathbb{Z}}_{\ell}}(\mathbf{G}(\mathbb{A}_{X\setminus S}), K_{X\setminus S})$  occurring in  $\mathbf{V}'_S$  such that its reduction equals  $\overline{\chi}_S$ .

Therefore, there is an irreducible factor  $\Pi'$  of  $\operatorname{Alg}_{\overline{\mathbb{Q}}_{\ell}}(\mathbf{G})$  contributing to  $\mathbf{V}'_{\mathrm{S}}$  such that  $\chi_{\mathrm{S}}(\Pi')$ and  $\chi'_{\mathrm{S}}$  coincide on  $\mathcal{H}_{\overline{\mathbb{Q}}_{\ell}}(\mathbf{G}(\mathbb{A}_{X\setminus \mathrm{S}}), \mathrm{K}_{X\setminus \mathrm{S}})$ . Such a  $\Pi'$  satisfies the conditions of the theorem.  $\Box$ 

#### 3.8.

In remains to prove Lemma 3.3.

*Proof.* — According to [60] Theorem 7.14 (and [42] Appendix A), there are a compact open subgroup J of G and an irreducible  $\overline{\mathbb{Q}}_{\ell}$ -representation  $\tau$  of J such that  $\pi$  is isomorphic to the compact induction of  $\tau$  to G. More precisely, the pair  $(J, \tau)$  can be chosen among *cuspidal types* of G in the sense of [42] Appendix A. It then has the following properties:

– There is a normal pro-*p*-subgroup  $J^1$  of J such that  $J/J^1$  is isomorphic to the group of rational points of a reductive group  $\mathscr{G}$  (whose neutral component is denoted  $\mathscr{G}^\circ$ ) defined over the residue field of F.

- The representation  $\tau$  decomposes as  $\kappa \otimes \xi$  where  $\kappa$  is a representation of J whose restriction to  $J^1$  is irreducible and  $\xi$  is an irreducible representation of J whose restriction to  $J^1$  is trivial.

– More precisely,  $\kappa$  is a *beta-extension* ([**60**] §4.1) of a *skew semisimple character*  $\theta$  ([**60**] §3.1) defined with respect to a *skew semisimple stratum* [ $\Lambda$ ,  $\beta$ ] ([**60**] §2.1) and  $\xi$  is the inflation of a representation of  $J/J^1$  whose restriction to the rational points of  $\mathscr{G}^{\circ}$  is cuspidal.

- The centre of the centralizer  $G_E$  of  $E = F[\beta]$  in G is compact, and the parahoric subgroup  $P^{\circ}(\Lambda_E)$  of  $G_E$  associated with  $[\Lambda, \beta]$  (see [60] §2.1) is a maximal parahoric subgroup of  $G_E$ .

**Lemma 3.5**. — The character  $\theta$  occurs in  $\pi'$ .

*Proof.* — By definition,  $\theta$  is a character of an open pro-*p*-subgroup  $H^1 = H^1(\Lambda, \beta)$  of *G*. Since  $\theta$  occurs in the restriction of  $\pi$  to  $H^1$ , its reduction mod  $\ell$  occurs in  $\mathbf{r}_{\ell}(\pi')|_{H^1}$ . Let *V* be an irreducible summand of  $\pi'|_{H^1}$  such that  $\mathbf{r}_{\ell}(V)$  contains  $\mathbf{r}_{\ell}(\theta)$ . Since  $H^1$  is a pro-*p*-group, *V* is isomorphic to  $\theta$ .

Now let  $\mathcal{C}$  denote the set of pairs  $([\Lambda', \beta], \theta')$  made of a skew semisimple stratum  $[\Lambda', \beta]$  and a skew semisimple character  $\theta' \in \mathcal{C}(\Lambda', \beta)$  occurring in  $\pi'$  such that

$$P^{\circ}(\Lambda'_E) \subseteq P^{\circ}(\Lambda_E)$$

and  $\mathcal{C}_{\min}$  denote the subset of  $\mathcal{C}$  made of all  $([\Lambda', \beta], \theta')$  such that  $P^{\circ}(\Lambda'_{E})$  is minimal among all parahoric subgroups of  $G_{E}$  occurring this way.

Let us prove that  $\mathcal{C}_{\min} = \mathcal{C}$ . Let  $([\Lambda', \beta], \theta') \in \mathcal{C}_{\min}$ . Then [60] §7.2 (in particular Lemma 7.4) and [42] Appendix A imply that  $\pi'$  contains a cuspidal type  $(J', \kappa' \otimes \xi')$  where  $J' = J(\Lambda', \beta)$  for some skew semisimple stratum  $[\Lambda', \beta]$  and  $\kappa'$  is any beta-extension of  $\theta'$ . By definition of a cuspidal type,  $P^{\circ}(\Lambda'_{E})$  is a maximal parahoric subgroup of  $G_{E}$ . It is thus equal to  $P^{\circ}(\Lambda_{E})$ .

It follows that  $([\Lambda, \beta], \theta) \in \mathbb{C}_{\min}$ . We thus may choose  $([\Lambda', \beta], \theta') = ([\Lambda, \beta], \theta)$  (hence J' = J) and  $\kappa' = \kappa$  in the paragraph above. Thus  $\pi'$  contains a cuspidal type  $(J, \kappa \otimes \xi')$ . It follows from [**60**] Corollary 6.19 that the compact induction of  $\tau' = \kappa \otimes \xi'$  from J to G is isomorphic to  $\pi'$ .

The representation  $\kappa$  is integral (since the group J is compact) and its reduction mod  $\ell$  is irreducible (by [**35**] Remark 6.3). Applying the functor  $\operatorname{Hom}_{J^1}(\kappa, -)$  from representations of G to representations of J which are trivial on  $J^1$ , which is compatible to reduction mod  $\ell$ , we deduce from [**35**] Corollary 8.5 that  $\mathbf{r}_{\ell}(\xi) \leq \mathbf{r}_{\ell}(\xi')$ , thus  $\mathbf{r}_{\ell}(\tau) \leq \mathbf{r}_{\ell}(\tau')$ .

#### 4. Globalizing representations

In this section, we fix a *p*-adic field F and a quasi-split special orthogonal, unitary or symplectic group G over F. Let k, w and  $\mathbf{G}$  be as in Theorem 2.1, and  $j: \mathbf{G}(F) \simeq G$  be an isomorphism of locally compact groups which we use to identify  $\mathbf{G}(F)$  with G.

Let  $\ell$  denote a prime number different from p, and fix a field isomorphism  $\iota$  as in (3.1). Let u be a finite place of k different from w, not dividing  $\ell$ .

In Paragraph 4.2 only, the prime number p will be assumed to be odd.

#### 4.1.

The next proposition is the first step towards Theorem 4.4. (See also Paragraph 1.3.)

**Proposition 4.1.** — Let  $\pi$  be a unitary cuspidal irreducible complex representation of G, and let  $\rho$  be a unitary cuspidal irreducible complex representation of  $\mathbf{G}(k_u)$ . There is an irreducible automorphic representation  $\Pi$  of  $\mathbf{G}(\mathbb{A})$  such that

- (1) the local component  $\Pi_u$  is isomorphic to  $\rho$ ,
- (2) the local component  $\Pi_w$  is isomorphic to  $\pi$ ,
- (3) the local component  $\Pi_v$  is the trivial character of  $\mathbf{G}(k_v)$  for any real place v of k.

**Remark 4.2.** — When the centre of G is compact, any cuspidal irreducible representation  $\pi$  of G is unitarizable. The only case when G has a non-compact centre is when it is isomorphic to the split special orthogonal group  $SO_2(F) \simeq F^{\times}$  (see [42] 4.2).

*Proof.* — Let  $\mathbf{Z}$  be the centre of  $\mathbf{G}$ . We start the proof by the following lemma.

Lemma 4.3. — There is a unitary automorphic character  $\Omega : \mathbf{Z}(\mathbb{A})/\mathbf{Z}(k) \to \mathbb{C}^{\times}$  such that

- (1) the local component  $\Omega_u$  is equal to the central character  $\omega_{\rho}$  of  $\rho$ ,
- (2) the local component  $\Omega_w$  is equal to the central character  $\omega_{\pi}$  of  $\pi$ ,
- (3) the local component  $\Omega_v$  is the trivial character of  $\mathbf{Z}(k_v)$  for any real place v of k.

*Proof.* — Let U denote the subgroup  $\mathbf{Z}(k_u) \times \mathbf{Z}(k_w) \times \mathbf{Z}(\mathbb{A}_\infty)$  of  $\mathbf{Z}(\mathbb{A})$ . The intersection  $U \cap \mathbf{Z}(k)$ is trivial, thus U identifies with a locally compact subgroup of  $\mathbf{Z}(\mathbb{A})/\mathbf{Z}(k)$ . By Pontryagin duality, any unitary character of U extends to  $\mathbf{Z}(\mathbb{A})/\mathbf{Z}(k)$ . (Note that  $\omega_{\rho}$  and  $\omega_{\pi}$  are unitary.) 

We now follow the proof of [24]. Let  $\Omega : \mathbf{Z}(\mathbb{A})/\mathbf{Z}(k) \to \mathbb{C}^{\times}$  be a unitary automorphic character as in Lemma 4.3. Let y be a finite place different from u and w.

Let us choose coefficients  $f_u$  and  $f_w$  of  $\rho$  and  $\pi$ , respectively, which are non-zero at 1.

For all real places v of k, let  $f_v$  be the constant function equal to 1 on  $\mathbf{G}(k_v)$ . As this group is compact,  $f_v$  is smooth and compactly supported.

For all finite places  $v \neq y$  such that **G** is unramified over  $k_v$  and  $\Omega_v$  is unramified, let  $f_v$  be the complex function on  $\mathbf{G}(k_v)$  supported on  $\mathbf{Z}(k_v)\mathbf{K}_v$  such that  $f_v(zg) = \Omega_v(z)$  for all  $z \in \mathbf{Z}(k_v)$ and all g in a fixed hyperspecial maximal compact subgroup  $K_v$  of  $\mathbf{G}(k_v)$ .

For any other place x, we choose a smooth complex function  $f_x$  on  $\mathbf{G}(k_x)$ , non-zero at 1, compactly supported modulo  $\mathbf{Z}(k_x)$  with restriction to this later group equal to  $\Omega_x$ .

We let f be the product of all these  $f_v$ . It is smooth and compactly supported on  $\mathbf{G}(\mathbb{A})$ . We may and will assume that

- the support of  $f_y$  is small enough so that

$$f(g^{-1})f(\gamma g) = 0$$
 for all  $g \in \mathbf{G}(\mathbb{A}), \gamma \in \mathbf{G}(k)$  such that  $\gamma \notin \mathbf{Z}(k)$ ,

- and  $f_v(g) = \overline{f_v(g^{-1})}$  for all places v of k and all  $g \in \mathbf{G}(k_v)$ .

We construct as in [24] the Poincaré series

$$Pf(g) = \sum_{\gamma \in \mathbf{Z}(k) \setminus \mathbf{G}(k)} f(\gamma g), \quad \text{for } g \in \mathbf{G}(\mathbb{A}).$$

We are in a particular case of the proof of [24] Appendice 1, so in particular this is well defined, non-zero, square-integrable and even cuspidal. There is thus an irreducible automorphic representation  $\Pi$  of  $\mathbf{G}(\mathbb{A})$ , with central character  $\Omega$ , such that, for each place v of k, one has

$$\int_{\mathbf{Z}(k_v)\backslash \mathbf{G}(k_v)} f_v(g^{-1}) \Pi_v(g) \, \mathrm{d}g \neq 0$$

where dg denotes a Haar measure on  $\mathbf{Z}(k_v) \setminus \mathbf{G}(k_v)$ . In particular, the local components  $\Pi_u$  and  $\Pi_w$  are isomorphic to  $\rho$  and  $\pi$ , respectively. At any real place v, the representation  $\Pi_v$  contains a vector which is  $\mathbf{G}(k_v)$ -invariant, so  $\Pi_v$  is trivial.  4.2.

We now assume that  $p \neq 2$ .

**Theorem 4.4.** — Let  $\pi_1$ ,  $\pi_2$  be integral cuspidal irreducible  $\overline{\mathbb{Q}}_{\ell}$ -representations of G such that

 $\mathbf{r}_{\ell}(\pi_1) \leq \mathbf{r}_{\ell}(\pi_2).$ 

Let  $\rho$  be a unitary cuspidal irreducible complex representation of  $\mathbf{G}(k_u)$  which is compactly induced from some compact mod centre, open subgroup of  $\mathbf{G}(k_u)$ . Assume that G is not the split special orthogonal group  $\mathrm{SO}_2(F) \simeq F^{\times}$ . There are irreducible automorphic representations  $\Pi_1$  and  $\Pi_2$  of  $\mathbf{G}(\mathbb{A})$  such that

- (1)  $\Pi_{1,u}$  and  $\Pi_{2,u}$  are both isomorphic to  $\rho$ ,
- (2)  $\Pi_{1,w} \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_{\ell}$  is isomorphic to  $\pi_1$  and  $\Pi_{1,w} \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_{\ell}$  is isomorphic to  $\pi_2$ ,
- (3)  $\Pi_{1,v}$  and  $\Pi_{2,v}$  are trivial for any real place v,
- (4) there is a finite set S of places of k, containing all real places, such that for all  $v \notin S$ :

(a) the local components  $\Pi_{1,v}$  and  $\Pi_{2,v}$  are unramified with respect to some hyperspecial maximal compact subgroup  $K_v$  of  $\mathbf{G}(k_v)$ ,

(b) the restrictions of the Satake parameters of  $\Pi_{1,v} \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_{\ell}$  and  $\Pi_{2,v} \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_{\ell}$  to the Hecke  $\overline{\mathbb{Z}}_{\ell}$ -algebra  $\mathcal{H}_{\overline{\mathbb{Z}}_{\ell}}(\mathbf{G}(k_{v}), \mathbf{K}_{v})$  are congruent mod the maximal ideal  $\mathfrak{m}$  of  $\overline{\mathbb{Z}}_{\ell}$ .

**Remark 4.5.** — The assumption on G implies that the centre of G is compact, thus any cuspidal irreducible  $\overline{\mathbb{Q}}_{\ell}$ -representation of G is integral.

*Proof.* — First, let us apply Proposition 4.1 with  $\pi = \pi_1 \otimes_{\overline{\mathbb{Q}}_{\ell}} \mathbb{C}$ . (Since the centre of G is compact, the central character of  $\pi$  has finite order, thus  $\pi$  is unitarizable.) We obtain an irreducible automorphic representation  $\Pi_1$  of  $\mathbf{G}(\mathbb{A})$  such that

- (1) the local component  $\Pi_{1,u}$  is isomorphic to  $\rho$ ,
- (2) the local component  $\Pi_{1,w} \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_{\ell}$  is isomorphic to  $\pi_1$ ,
- (3) the local component  $\Pi_{1,v}$  is the trivial character of  $\mathbf{G}(k_v)$  for any real place v.

We then choose for S a set of finite places of k as in Paragraph 3.4, that is, S contains u, w and all places dividing  $\ell$ , and, for any finite place  $v \notin S$ , the local component  $\Pi_{1,v}$  is unramified with respect to some hyperspecial maximal compact subgroup  $K_v$  of  $\mathbf{G}(k_v)$ . For such v, this defines a  $\overline{\mathbb{Z}}_{\ell}$ -character  $\chi_{1,v}$  of  $\mathcal{H}_{\overline{\mathbb{Z}}_{\ell}}(\mathbf{G}(k_v), \mathbf{K}_v)$ .

We now apply Theorem 3.1 with  $\pi' = \pi_2$ . The conditions of Paragraph 3.5 are automatically satisfied for  $\Pi_{1,w}$  thanks to [60]. We get an irreducible automorphic representation  $\Pi_2$  of the group  $\mathbf{G}(\mathbb{A})$ , trivial at infinity, such that

- (1) the local component  $\Pi_{2,w} \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_{\ell}$  is isomorphic to  $\pi_2$ ,
- (2) the local component  $\Pi_{2,u}$  is isomorphic to  $\rho$ ,

(3) for all finite places  $v \notin S$ , the local component  $\Pi_{2,v} \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_{\ell}$  is  $K_v$ -unramified with associated  $\overline{\mathbb{Z}}_{\ell}$ -character  $\chi_{2,v} : \mathcal{H}_{\overline{\mathbb{Z}}_{\ell}}(\mathbf{G}(k_v), K_v) \to \overline{\mathbb{Z}}_{\ell}$ , and  $\chi_{1,v}$  and  $\chi_{2,v}$  are congruent mod  $\mathfrak{m}$ .

This proves Theorem 4.4.

 $\mathbf{22}$ 

#### 5. Global transfer

#### 5.1. Quasi-split classical groups

Let k be either a p-adic field for some prime number p, or a real Archimedean local field, or a totally real number field. We will consider the following families of quasi-split reductive groups over k:

(1) For  $n \ge 1$ , the (split) symplectic group  $\operatorname{Sp}_{2n}$  defined as  $\operatorname{Sp}(f)$ , where f is the alternating form on  $k^{2n} \times k^{2n}$  defined by

(5.1) 
$$f(x_1, \dots, x_{2n}, y_1, \dots, y_{2n}) = x_1 y_{2n} - x_{2n} y_1 + \dots + x_n y_{n+1} - x_{n+1} y_n.$$

(2) For  $n \ge 1$  and  $\alpha \in k^{\times}$ , the (split) special orthogonal group  $SO_{2n+1}$  defined as SO(q), where q is the quadratic form on  $k^{2n+1}$  of discriminant  $(-1)^n \alpha$  defined by

(5.2) 
$$q(x_1, \dots, x_{2n+1}) = x_1 x_2 + \dots + x_{2n-1} x_{2n} + \alpha x_{2n+1}^2$$

(3) For  $n \ge 1$  and  $\alpha \in k^{\times}$ , the special orthogonal group  $SO_{2n}^{\alpha}$  defined as SO(q), where q is the quadratic form on  $k^{2n}$  of discriminant  $(-1)^n \alpha$  defined by

(5.3) 
$$q(x_1, \dots, x_{2n}) = x_1 x_2 + \dots + x_{2n-3} x_{2n-2} + x_{2n-1}^2 - \alpha x_{2n}^2$$

(4) For  $n \ge 1$  and  $\alpha \in k^{\times}$ , the unitary group  $U_n^{\alpha}$  defined as U(h), where h is the l/k-Hermitian form on  $l^n$  of discriminant  $(-1)^{n(n-1)/2}$  defined by

(5.4) 
$$h(x_1, \dots, x_n) = x_1^c x_n - x_2^c x_{n-1} + \dots + (-1)^{n-1} x_n^c x_1$$

where *l* is the *k*-algebra  $k[X]/(X^2 - \alpha)$  and *c* is the non-trivial automorphism of l/k. If  $\alpha \in k^{\times 2}$ , the *k*-group  $U_n^{\alpha}$  is thus isomorphic to  $GL_n$ .

In the even orthogonal and unitary cases, the image of  $\alpha$  in  $k^{\times}/k^{\times 2}$  will still be denoted  $\alpha$ .

#### 5.2. The dual group

In this paragraph, k is either a p-adic field or a totally real number field and  $\mathbf{G}^*$  is one of the quasi-split special orthogonal, unitary or symplectic k-groups of 5.1. We define its dual group

$$\widehat{\mathbf{G}} = \begin{cases} \operatorname{SO}_{2n+1}(\mathbb{C}) & \text{if } \mathbf{G}^* = \operatorname{Sp}_{2n}, \\ \operatorname{Sp}_{2n}(\mathbb{C}) & \text{if } \mathbf{G}^* = \operatorname{SO}_{2n+1}, \\ \operatorname{SO}_{2n}(\mathbb{C}) & \text{if } \mathbf{G}^* = \operatorname{SO}_{2n}^{\alpha}, \\ \operatorname{GL}_n(\mathbb{C}) & \text{if } \mathbf{G}^* = \operatorname{U}_n^{\alpha}. \end{cases}$$

In the even orthogonal case, the groups  $SO_{2n}(\mathbb{C}) \subseteq O_{2n}(\mathbb{C})$  are defined with respect to the symmetric bilinear form  $\langle \cdot, \cdot \rangle$  on  $\mathbb{C}^{2n}$  given by

$$\langle \mathbf{e}_i, \mathbf{e}_j \rangle = \begin{cases} 0 & \text{if } i+j \neq 2n+1, \\ 1 & \text{otherwise,} \end{cases}$$

where  $(\mathbf{e}_1, \ldots, \mathbf{e}_{2n})$  is the canonical basis of  $\mathbb{C}^{2n}$ .

#### 5.3. The local Langlands correspondence

In this paragraph, k is a p-adic field and  $\mathbf{G}^*$  is either the general linear k-group  $\operatorname{GL}_n$  for some  $n \ge 1$  (whose dual group is  $\operatorname{GL}_n(\mathbb{C})$ ) or one of the quasi-split classical k-groups of 5.1. We denote by  $W_k$  the Weil group of  $\overline{\mathbb{Q}}_p$  over k, and define the semi-direct product  ${}^L\mathbf{G} = \widehat{\mathbf{G}} \rtimes W_k$ , where

• the action of  $W_k$  on  $\hat{\mathbf{G}}$  is trivial when  $\mathbf{G}^*$  is split (that is, when  $\mathbf{G}^*$  is general linear, symplectic, odd orthogonal, even orthogonal with  $\alpha = 1$  or unitary with  $\alpha = 1$ ),

• when  $\mathbf{G}^*$  is even orthogonal and  $\alpha \neq 1$ , and if l denotes the quadratic extension of k in  $\overline{\mathbb{Q}}_p$ generated by a square root of  $\alpha$ , the action of  $W_k$  on  $\hat{\mathbf{G}}$  factors through  $\operatorname{Gal}(l/k)$ , the generator cof which acts by conjugacy by the element  $w \in \operatorname{O}_{2n}(\mathbb{C})$  fixing  $\mathbf{e}_1, \ldots, \mathbf{e}_{n-1}, \mathbf{e}_{n+2}, \ldots, \mathbf{e}_{2n}$  and exchanging  $\mathbf{e}_n$  and  $\mathbf{e}_{n+1}$  (thus  $\hat{\mathbf{G}} \rtimes \operatorname{Gal}(l/k)$  identifies with  $\operatorname{O}_{2n}(\mathbb{C})$ ),

• when  $\mathbf{G}^*$  is unitary and  $\alpha \neq 1$ , and if l denotes the quadratic extension of k in  $\overline{\mathbb{Q}}_p$  generated by a square root of  $\alpha$ , the action of  $W_k$  on  $\hat{\mathbf{G}}$  factors through the group  $\operatorname{Gal}(l/k)$  whose generator c acts by

$$g \mapsto g^* = \mathbf{J} \cdot {}^{\mathbf{t}}g^{-1} \cdot \mathbf{J}^{-1}$$

where <sup>t</sup>g denotes the transpose of  $g \in GL_n(\mathbb{C})$  and J is the antidiagonal matrix in  $GL_n(\mathbb{C})$  defined by  $J_{i,j} = 0$  if  $i + j \neq n + 1$  and  $J_{i,n+1-i} = (-1)^{i-1}$ .

Let  $WD_k = W_k \times SL_2(\mathbb{C})$  denote the Weil-Deligne group of k. A *(local) Langlands parameter* for  $\mathbf{G}(k)$  is a group homomorphism

$$\varphi: WD_k \to \widehat{\mathbf{G}} \rtimes W_k$$

such that

- its restriction to  $W_k$  is smooth,
- its restriction to  $SL_2(\mathbb{C})$  is algebraic,
- the projection of  $\varphi(\mathbf{W}_k)$  onto  $\widehat{\mathbf{G}}$  is made of semi-simple elements, and

- the projection of  $\varphi(w, x)$  onto  $W_k$  is equal to w for all  $(w, x) \in WD_k$ .

When  $\mathbf{G}^*$  is split, this is the same as a morphism  $WD_k \to \mathbf{G}$  satisfying the first three points. In the even orthogonal case with  $\alpha \neq 1$ , this is the same as a morphism  $WD_k \to O_{2n}(\mathbb{C})$  satisfying the first three points and whose determinant is the quadratic character

$$x \mapsto (\alpha, x)$$

of  $k^{\times}$ , which can be seen as a character of  $W_k$  via the Artin reciprocity map of local class field theory. We say a local Langlands parameter  $\varphi$  is bounded if  $\varphi(W_k)$  is relatively compact in  $\hat{\mathbf{G}}$ . Let

- $-\Phi(\mathbf{G}^*, k)$  be the set of  $\widehat{\mathbf{G}}$ -conjugacy classes of local Langlands parameters for  $\mathbf{G}^*$  over k,
- $-\Pi(\mathbf{G}^*(k))$  be the set of isomorphism classes of irreducible representations of  $\mathbf{G}^*(k)$ .

When  $\mathbf{G}^*$  is the general linear group  $\operatorname{GL}_n$ , the local Langlands correspondence ([23, 25]) is a bijection from  $\Pi(\operatorname{GL}_n(k))$  to  $\Phi(\operatorname{GL}_n, k)$ .

When **G**<sup>\*</sup> is classical, the local Langlands correspondence ([2] Theorem 2.2.1, [47] Theorems 2.5.1, 3.2.1, see also [4] Theorems 3.2, 3.6 and Remarks 3.3, 3.7) defines

(1) (symplectic, odd orthogonal and unitary cases) a partition

(5.5) 
$$\Pi(\mathbf{G}^*(k)) = \coprod_{\varphi \in \Phi(\mathbf{G}^*,k)} \Pi_{\varphi}(\mathbf{G}^*(k))$$

into non-empty finite sets  $\Pi_{\varphi}(\mathbf{G}^*(k))$  if  $\mathbf{G}^*$  is symplectic, odd special orthogonal or unitary,

(2) (even orthogonal case) a partition

(5.6) 
$$\Pi(\mathrm{SO}_{2n}^{\alpha}(k)) = \coprod_{\varphi \in \Phi(\mathrm{SO}_{2n}^{\alpha}, k) / \mathrm{O}_{2n}(\mathbb{C})} \Pi_{\varphi}(\mathrm{SO}_{2n}^{\alpha}(k))$$

where each  $\Pi_{\varphi}(\mathrm{SO}_{2n}^{\alpha}(k))$  is non-empty, finite and stable under  $\mathrm{O}_{2n}^{\alpha}(k)$ -conjugacy.

In each case, we have the following properties:

•  $\Pi_{\varphi}(\mathbf{G}^*(k))$  contains a tempered representation if and only if  $\varphi$  is bounded. When this is the case, all representations in  $\Pi_{\varphi}(\mathbf{G}^*(k))$  are tempered. (See for instance [2] Theorem 1.5.1 for symplectic and special orthogonal groups, and [31] Theorem 1.6.1 for unitary groups.)

•  $\Pi_{\varphi}(\mathbf{G}^*(k))$  contains a discrete series representation if and only if  $\varphi$  is bounded and the quotient of the centralizer of the image of  $\varphi$  in  $\mathbf{\hat{G}}$  by  $\mathbf{Z}(\mathbf{\hat{G}})^{W_k}$  is finite. When this is the case, all representations in  $\Pi_{\varphi}(\mathbf{G}^*(k))$  are discrete series representations. (See for instance [70] Theorem 2.2 for symplectic and special orthogonal groups, and [31] Theorem 1.6.1 for unitary groups.)

#### 5.4. The local transfer

In this paragraph, k is a p-adic field and  $\mathbf{G}^*$  is one of the quasi-split classical k-groups of 5.1. If  $\mathbf{G}^*$  is symplectic or special orthogonal, there is a morphism  $\mathrm{Std}: \hat{\mathbf{G}} \to \mathrm{GL}_N(\mathbb{C})$  with

(5.7) 
$$N = N(\mathbf{G}^*) = \begin{cases} 2n & \text{if } \mathbf{G}^* = \mathrm{SO}_{2n+1} \text{ or } \mathbf{G}^* = \mathrm{SO}_{2n}^{\alpha}, \\ 2n+1 & \text{if } \mathbf{G}^* = \mathrm{Sp}_{2n}, \end{cases}$$

given by the natural inclusion. We extend it to a morphism  $\mathrm{Std}: \widehat{\mathbf{G}} \rtimes \mathrm{W}_k \to \mathrm{GL}_N(\mathbb{C})$  as follows:

• Std is trivial on  $W_k$  when  $G^*$  is split,

• when  $\mathbf{G}^*$  is even orthogonal and  $\alpha \neq 1$ , Std is trivial on  $W_l$  and  $\operatorname{Std}(c) = w \in O_{2n}(\mathbb{C})$ , thus Std factors through

$$\mathrm{SO}_{2n}(\mathbb{C}) \rtimes \mathrm{W}_k \twoheadrightarrow \mathrm{SO}_{2n}(\mathbb{C}) \rtimes \mathrm{Gal}(l/k) \simeq \mathrm{O}_{2n}(\mathbb{C}) \subseteq \mathrm{GL}_{2n}(\mathbb{C})$$

(see also [4] 3.2).

In the unitary case ( $\mathbf{G}^* = \mathrm{U}_n^{\alpha}$ ), we need to introduce the k-group  $\mathrm{GL}_n^{\alpha}$ , the restriction of  $\mathrm{GL}_n$ with respect to l/k. Its dual group is  $\mathrm{GL}_n(\mathbb{C}) \times \mathrm{GL}_n(\mathbb{C})$ , and we define the semi-direct product

$${}^{L}\mathrm{GL}_{n}^{\alpha} = (\mathrm{GL}_{n}(\mathbb{C}) \times \mathrm{GL}_{n}(\mathbb{C})) \rtimes \mathrm{W}_{k}$$

where

• the action of  $W_k$  is trivial when l/k is split,

• otherwise, the action of  $W_k$  factors through  $\operatorname{Gal}(l/k)$  and c acts by  $(g,h) \mapsto (h,g)$ .

It will be convenient to set

$$(5.8) N = N(\mathbf{U}_n^{\alpha}) = n.$$

Let Std be the morphism  $\widehat{\mathbf{G}} \rtimes W_k \to (\operatorname{GL}_N(\mathbb{C}) \times \operatorname{GL}_N(\mathbb{C})) \rtimes W_k$  defined by  $g \rtimes w \mapsto (g, g^*) \rtimes w$ .

Given an irreducible representation  $\pi \in \Pi(\mathbf{G}^*(k))$ , let  $\varphi \in \Phi(\mathbf{G}^*, k)$  be a Langlands parameter such that  $\pi \in \Pi_{\varphi}(\mathbf{G}^*(k))$ . (In the even orthogonal case,  $\varphi$  is determined up to  $O_{2n}(\mathbb{C})$ -conjugacy only.)

If  $\mathbf{G}^*$  is symplectic or special orthogonal, then, composing with Std, we get a local Langlands parameter  $\phi = \text{Std} \circ \varphi \in \Phi(\text{GL}_N, k)$ , uniquely determined up to  $\text{GL}_N(\mathbb{C})$ -conjugacy.

If  $\mathbf{G}^*$  is unitary, then, composing with Std, we obtain a Langlands parameter

$$\operatorname{Std} \circ \varphi : \operatorname{WD}_k \to (\operatorname{GL}_N(\mathbb{C}) \times \operatorname{GL}_N(\mathbb{C})) \rtimes \operatorname{W}_k.$$

• If l is non-split, its restriction to  $WD_l$  has the form  $(w, x) \mapsto (\phi(w, x), \phi(w, x)^*) \rtimes w$  for a local Langlands parameter  $\phi \in \Phi(GL_N, l)$ , uniquely determined up to  $GL_N(\mathbb{C})$ -conjugacy.

• If l is split, it is of the form  $(w, x) \mapsto (\phi(w, x), \phi(w, x)^*) \rtimes w$  for a local Langlands parameter  $\phi \in \Phi(\operatorname{GL}_N, k)$ , uniquely determined up to  $\operatorname{GL}_N(\mathbb{C})$ -conjugacy.

**Definition 5.1.** — The local transfer of  $\pi$ , denoted  $\mathbf{t}(\pi)$ , is the isomorphism class of irreducible representations associated with  $\phi$  through the local Langlands correspondence. It is

- (1) a class of representations of  $GL_N(k)$  if  $\mathbf{G}^*$  is symplectic or special orthogonal,
- (2) a class of representations of  $GL_N(l)$  if  $\mathbf{G}^*$  is unitary,

which is uniquely determined by the isomorphism class of  $\pi$ .

**Remark 5.2.** — If  $\mathbf{G}^*$  is unitary and l is split over k, and if we fix an isomorphism of k-algebras  $l \simeq k \times k$ , which we use to identify  $U_n^{\alpha}(k)$  with  $\operatorname{GL}_n(k)$  and  $\operatorname{GL}_N(l)$  with  $\operatorname{GL}_N(k) \times \operatorname{GL}_N(k)$ , then

(5.9) 
$$\mathbf{t}(\pi) = \pi \otimes \pi^{\vee}$$

(where  $\pi^{\vee}$  is the contragredient of  $\pi$ ). This does not depend on the choice of  $l \simeq k \times k$ . Indeed, making the other choice twists the isomorphism  $U_n^{\alpha}(k) \simeq GL_n(k)$  by  $g \mapsto g^*$  (see (2.3) and the explanation thereafter) and the isomorphism  $GL_N(l) \simeq GL_N(k) \times GL_N(k)$  by  $(g,h) \mapsto (h,g)$ , which gives (5.9) again since  $g \mapsto \pi(g^*)$  is isomorphic to  $\pi^{\vee}$ .

In Section 6, we will describe explicitly the local transfer for unramified representations when  $\mathbf{G}^*$  is unramified over k, and will describe its congruence properties.

#### 5.5. Arthur parameters in the symplectic and orthogonal cases

In this paragraph, k is a totally real number field and  $\mathbf{G}^*$  is symplectic or quasi-split special orthogonal. We write A for the ring of adèles of k and  $N = N(\mathbf{G}^*)$  (see (5.7)).

**Definition 5.3**. — A discrete global Arthur parameter (for  $\mathbf{G}^*$ ) is a formal sum

(5.10) 
$$\psi = \Pi_1[d_1] \oplus \cdots \oplus \Pi_r[d_r]$$

for some integer  $r \ge 1$ , where, for each  $i \in \{1, \ldots, r\}$ ,  $d_i$  is a positive integer and  $\Pi_i$  is a self-dual cuspidal automorphic irreducible representation of  $\operatorname{GL}_{N_i}(\mathbb{A})$  for some  $N_i \ge 1$ , such that

- (1) one has  $N_1 d_1 + \dots + N_r d_r = N$ ,
- (2) if  $r \ge 2$  and  $\Pi_i \simeq \Pi_j$  for some  $i \ne j$  in  $\{1, \ldots, r\}$ , then  $d_i \ne d_j$ ,

 $\mathbf{26}$ 

(3) the self-dual representation  $\Pi_i$  has the same parity as  $\hat{\mathbf{G}}$  if  $d_i$  is odd, and has the opposite parity if  $d_i$  is even, where the parity of  $\Pi_i$  is defined to be orthogonal if  $L(s, \Pi_i, \text{Sym}^2)$  has a pole at s = 1, and symplectic if  $L(s, \Pi_i, \wedge^2)$  has a pole at s = 1,

(4) the character  $\omega_{\Pi_1}^{d_1} \dots \omega_{\Pi_r}^{d_r}$  is trivial if  $\mathbf{G}^* = \operatorname{Sp}_{2n}$  or  $\mathbf{G}^* = \operatorname{SO}_{2n+1}$ , and is equal to the quadratic character

$$\chi_{\alpha}: x \mapsto \prod_{v} (\alpha_{v}, x_{v})_{v} \in \{-1, 1\}$$

of  $\mathbb{A}^{\times}/k^{\times}$  if  $\mathbf{G}^* = \mathrm{SO}_{2n}^{\alpha}$ , where  $\omega_{\Pi_i}$  is the central character of  $\Pi_i$ .

A discrete global Arthur parameter  $\Sigma_1[e_1] \oplus \cdots \oplus \Sigma_s[e_s]$  is said to be *equivalent to* (5.10) if we have s = r and there is a permutation  $\varepsilon \in \mathfrak{S}_r$  such that  $e_i = d_{\varepsilon(i)}$  and  $\Sigma_i \simeq \prod_{\varepsilon(i)}$  for each *i*. Let

$$\widetilde{\Psi}_2(\mathbf{G}^*)$$

be the set of equivalence classes of discrete global Arthur parameters for  $\mathbf{G}^*$ .

Associated with a discrete global Arthur parameter  $\psi \in \Psi_2(\mathbf{G}^*)$  given by (5.10), there are a local Arthur parameter  $\psi_v$  and a local Arthur packet  $\Pi_{\psi_v}(\mathbf{G}^*(k_v))$  for each finite place v of k: see (5.11) and (5.12) below.

Let v be a finite place of k, and consider the local component  $\Pi_{i,v}$  for some i. It is a unitarisable irreducible representation of  $\operatorname{GL}_{N_i}(k_v)$ . Associated with it through the local Langlands correspondence for  $\operatorname{GL}_{N_i}(k_v)$ , there is a local Langlands parameter

$$\phi_{i,v}: WD_{k_v} \to GL_{N_i}(\mathbb{C}),$$

uniquely determined up to  $\operatorname{GL}_{N_i}(\mathbb{C})$ -conjugacy. Since one does not know whether  $\Pi_{i,v}$  is tempered, the parameter  $\phi_{i,v}$  might not be bounded.

We define a morphism

(5.11) 
$$\psi_v = (\phi_{1,v} \boxtimes S_{d_1}) \oplus \cdots \oplus (\phi_{r,v} \boxtimes S_{d_r}) : WD_{k_v} \times SL_2(\mathbb{C}) \to GL_N(\mathbb{C})$$

where  $S_d = Sym^{d-1}$  denotes the unique irreducible algebraic representation of  $SL_2(\mathbb{C})$  of dimension  $d \ge 1$ . Recall that we have defined a morphism Std in 5.3. By [2] Theorem 1.4.2, there is a local Arthur parameter  $\xi : WD_{k_v} \times SL_2(\mathbb{C}) \to \hat{\mathbf{G}} \rtimes W_{k_v}$  such that  $\psi_v$  is  $GL_N(\mathbb{C})$ -conjugate to  $Std \circ \xi$ . The parameter  $\xi$  is uniquely determined up to  $\hat{\mathbf{G}}$ -conjugacy, except if  $\mathbf{G}^* = SO_{2n}^{\alpha}$  and all  $N_1d_1, \ldots, N_rd_r$  are even, in which case there are two  $\hat{\mathbf{G}}$ -conjugacy classes of such  $\xi$ .

Associated with  $\psi_v$ , there is a multiset  $\Pi_{\psi_v}(\mathbf{G}^*(k_v))$  of irreducible smooth representations of  $\mathbf{G}^*(k_v)$ , that is, a map

(5.12) 
$$\Pi(\mathbf{G}^*(k_v)) \to \mathbb{Z}_{\geq 0}$$

with finite support, where  $\Pi(\mathbf{G}^*(k_v))$  is the set of isomorphism classes of irreducible smooth representations of  $\mathbf{G}^*(k_v)$ . If  $\psi_v(\mathbf{W}_{k_v})$  is relatively compact in  $\mathrm{GL}_N(\mathbb{C})$ , this comes from [2] Theorems 1.5.1, 2.2.1, 2.2.4 and (5.12) is supported in the subset  $\Pi_{\mathrm{unit}}(\mathbf{G}^*(k_v))$  of unitarisable representations. Thanks to Moeglin ([44], see also [69] Theorem 8.12), it does not take any value > 1, that is,  $\Pi_{\psi_v}(\mathbf{G}^*(k_v))$  can be regarded as a finite subset of  $\Pi_{\mathrm{unit}}(\mathbf{G}^*(k_v))$ .

When  $\psi_v(W_{k_v})$  is not relatively compact,  $\Pi_{\psi_v}(\mathbf{G}^*(k_v))$  is obtained from the relatively compact case by a parabolic induction process: see [2] 1.5 in the symplectic and orthogonal cases and [4] 6.5 in the even orthogonal case. For our purpose, it will be enough to make the following remark.

**Remark 5.4.** — Let v be a finite place of k, and assume that  $\Pi_{\psi_v}(\mathbf{G}^*(k_v))$  contains a cuspidal representation. Then  $\psi_v(\mathbf{W}_{k_v})$  is relatively compact in  $\mathrm{GL}_N(\mathbb{C})$ .

When  $\psi_v$  is trivial on the SL<sub>2</sub>( $\mathbb{C}$ )-factor, that is,  $\psi_v$  is a local Langlands parameter for  $\mathbf{G}^*(k_v)$ , the local Arthur packet  $\Pi_{\psi_v}(\mathbf{G}^*(k_v))$  coincides with the *L*-packet associated with  $\psi_v$  by the local Langlands correspondence in (5.5) and (5.6). (See [4] top of Paragraph 6.3.)

#### 5.6. Transfer

In this paragraph, k is a totally real number field,  $\mathbf{G}^*$  is one of the quasi-split special orthogonal, unitary or symplectic k-groups of 5.1 and  $\mathbf{G}$  is an inner form of  $\mathbf{G}^*$  over k such that  $\mathbf{G}(k_v)$  is compact for any real place v and quasi-split for any finite place v.

In order to state the following theorem, we need more than the group **G**. Following [63] and [31], we realize **G** as

• a rigid inner twist of  $\mathbf{G}^*$  in the symplectic case (see Paragraph 2.8),

• a pure inner twist of  $\mathbf{G}^*$  in the special orthogonal and unitary cases, that is, we fix a quadratic form q such that  $\mathbf{G} = SO(q)$  or a Hermitian form h such that  $\mathbf{G} = U(h)$ . (See for instance [33] Sections 29.D, 29.E.)

If  $\mathbf{G}^*$  is special orthogonal, let  $q^*$  be the quadratic form (5.2) or (5.3) such that  $\mathbf{G}^* = \mathrm{SO}(q^*)$ , and let  $\alpha = (-1)^{n(n-1)/2} \delta(q^*)$  be its normalized discriminant. Let v be a finite place of k:

• if  $q \otimes k_v$  is equivalent to  $q^* \otimes k_v$ , any choice of  $k_v$ -isomorphism f such that  $q = q^* \circ f$  defines a group isomorphism  $j : \mathbf{G}(k_v) \simeq \mathbf{G}^*(k_v)$ , and changing f changes j by an inner automorphism, which does not affect isomorphism classes of representations of these groups;

• if  $q \otimes k_v$  is not equivalent to  $q^* \otimes k_v$ , which can only happen when  $\mathbf{G}^* = \mathrm{SO}_{2n}^{\alpha}$  with  $\alpha \neq 1$ , there is a  $\lambda \in k_v^{\times}$  such that  $q \otimes k_v$  is equivalent to  $\lambda \cdot (q^* \otimes k_v)$ . We thus have (canonically upto an inner automorphism)  $\mathbf{G}(k_v) \simeq \mathrm{SO}(\lambda \cdot (q^* \otimes k_v)) = \mathbf{G}^*(k_v)$ .

If  $\mathbf{G}^*$  is unitary, let  $h^*$  be the l/k-Hermitian form (5.4) such that  $\mathbf{G}^* = \mathrm{U}(h^*)$ . Let v be a finite place of k:

• if  $h \otimes k_v$  is equivalent to  $h^* \otimes k_v$ , any choice of isomorphism f such that  $h = h^* \circ f$  defines a group isomorphism  $j : \mathbf{G}(k_v) \simeq \mathbf{G}^*(k_v)$ , and changing f changes j by an inner automorphism, which does not affect isomorphism classes of representations of these groups;

• if  $h \otimes k_v$  is not equivalent to  $h^* \otimes k_v$ , which can only happen when  $\mathbf{G}^* = \mathrm{U}_{2n+1}^{\alpha}$  with  $\alpha \neq 1$ , there is a  $\delta \in k_v^{\times}$  such that  $h \otimes k_v$  is equivalent to  $\delta \cdot (h^* \otimes k_v)$ . We thus have (canonically up to an inner automorphism)  $\mathbf{G}(k_v) \simeq \mathrm{U}(\delta \cdot (h^* \otimes k_v)) = \mathbf{G}^*(k_v)$ .

If  $\mathbf{G}^*$  is the symplectic group  $\operatorname{Sp}_{2n}$ , then  $\mathbf{G}(k)$  is the group made of all  $g \in \mathbf{M}_n(D)$  such that  $g^*g = 1$ , where D is a quaternion k-algebra which is split at each finite place and definite at each real place, and  $g^*$  in the matrix whose (i, j)-entry is the conjugate of  $g_{ji}$ . (See 2.8 and [62] 2.1.1.) Let v be a finite place of k, and fix an isomorphism of  $k_v$ -algebras  $u : D \otimes_k k_v \simeq \mathbf{M}_2(k_v)$ . Through u, the group  $\mathbf{G}(k_v)$  identifies with  $\operatorname{Sp}(f_v)$  for some alternating form  $f_v$  on  $k_v^{2n} \times k_v^{2n}$ . Changing u

 $\mathbf{28}$ 

changes this identification by an inner automorphism. We thus have (canonically up to an inner automorphism) a group isomorphism  $\mathbf{G}(k_v) \simeq \mathbf{G}^*(k_v)$ .

In all cases, we have explained how to canonically identify representations of  $\mathbf{G}(k_v)$  with those of  $\mathbf{G}^*(k_v)$ . This thus defines a local transfer for irreducible representations of  $\mathbf{G}(k_v)$ .

**Theorem 5.5.** — Assume that the group  $\mathbf{G}^*$  is symplectic or special orthogonal. Let  $\pi$  be an irreducible automorphic representation of  $\mathbf{G}(\mathbb{A})$  and suppose that there is a finite place u of k such that both the local component  $\pi_u$  and its local transfer to  $\operatorname{GL}_N(k_u)$  are cuspidal. There is a unique self-dual cuspidal automorphic representation  $\Pi$  of  $\operatorname{GL}_N(\mathbb{A})$  such that

- (1) for all finite places v of k, the local transfer of  $\pi_v$  to  $GL_N(k_v)$  is  $\Pi_v$ ,
- (2) for all real places v of k, the infinitesimal character of  $\Pi_v$  is algebraic regular.

*Proof.* — First note that, associated with any discrete global Arthur parameter  $\psi \in \Psi_2(\mathbf{G}^*)$  and any finite place v of k, there is a local Arthur packet  $\Pi_{\psi_v}(\mathbf{G}^*(k_v))$ . We explained how to canonically identify representations of  $\mathbf{G}(k_v)$  with those of  $\mathbf{G}^*(k_v)$ . This thus defines a local Arthur packet  $\Pi_{\psi_v}(\mathbf{G}(k_v))$ .

Now, as **G** is compact at all real places and quasi-split at all finite places, [63] Theorem 4.0.1 and Remark 4.0.2 apply. We thus get a global Arthur parameter  $\psi$  for **G**<sup>\*</sup> such that

(1)  $\pi_v \in \Pi_{\psi_v}(\mathbf{G}(k_v))$  for all finite places v of k,

(2) the infinitesimal character of  $\psi_v$  is algebraic regular for all real places v of k.

In the remainder of the proof, we follow an argument which has been suggested to us by A. Moussaoui, whom we thank for this. First, at v = u, we have

$$\pi_u \in \Pi_{\psi_u}(\mathbf{G}(k_u))$$

and it follows from Remark 5.4 that  $\psi_u(W_{k_u})$  is relatively compact in  $\operatorname{GL}_N(\mathbb{C})$ . Associated with  $\psi_u$  in [43] 4.1, there is its *extended cuspidal support* (or *infinitesimal character*), denoted  $\lambda_u$ . It is the N-dimensional representation of  $W_{k_u}$  defined by

$$\lambda_u(w) = \psi_u(w, \mathsf{d}_w, \mathsf{d}_w), \quad \mathsf{d}_w = \begin{pmatrix} |w|^{1/2} & 0\\ 0 & |w|^{-1/2} \end{pmatrix} \in \mathrm{SL}_2(\mathbb{C}), \quad w \in \mathrm{W}_{k_u},$$

where  $w \mapsto |w|$  is the character  $W_{k_u} \to \mathbb{R}^{\times}_+$  defined by  $|w| = q^{-v(w)}$ , where q is the cardinality of the residue field of  $k_u$  and  $v(w) \in \mathbb{Z}$  is the valuation of w, normalized so that any geometric Frobenius element has valuation 1. If we write explicitly

$$\psi_u = \bigoplus_{i=1}^m \sigma_i \boxtimes \mathbf{S}_{a_i} \boxtimes \mathbf{S}_{b_i}$$

for some  $m \ge 1$ , with  $a_i, b_i \ge 1$  and where  $\sigma_i$  is an irreducible representation of  $W_{k_u}$ , then

(5.13) 
$$\lambda_u = \bigoplus_{i=1}^m \bigoplus_{j=0}^{b_i - 1} \bigoplus_{k=0}^{a_i - 1} \sigma_i |\cdot|^{(b_i - 1)/2 + (a_i - 1)/2 - j - k}$$

On the other hand, by [43] 4.1 again, the extended cuspidal support (or infinitesimal character) of  $\pi_u$  is the representation  $\lambda$  of  $W_{k_u}$  defined by  $\lambda(w) = \phi(w, \mathsf{d}_w)$  for all  $w \in W_{k_u}$ , where  $\phi = \operatorname{Std} \circ \varphi$ 

and  $\varphi$  is the Langlands parameter associated with  $\pi_u$  (up to  $O_{2n}(\mathbb{C})$ -conjugacy in the even orthogonal case). Given the assumption that we made on  $\pi_u$ , the extended cuspidal support  $\lambda$  is irreducible. By [43] Proposition 4.1, the extended cuspidal supports of  $\psi_u$  and  $\pi_u$  coincide. It follows that (5.13) is irreducible, which implies that m = 1 and  $a_1 = b_1 = 1$ . Thus  $\psi$  satisfies r = 1 and  $d_1 = 1$ .

We thus have  $\psi = \Pi[1]$  for a uniquely determined self-dual cuspidal automorphic irreducible representation  $\Pi$  of  $\operatorname{GL}_N(\mathbb{A})$ . Given a finite place v of k, the local component  $\pi_v$  is in the Arthur packet  $\Pi_{\psi_v}(\mathbf{G}(k_v))$ . Since  $\psi_v$  is a Langlands parameter (as  $d_1 = 1$ ), this Arthur packet is an L-packet, thus  $\Pi_v$  is the local transfer of  $\pi_v$  to  $\operatorname{GL}_N(k_v)$ .  $\Box$ 

We now consider the case of unitary groups.

**Theorem 5.6**. — Assume that the group  $\mathbf{G}^*$  is unitary. Let  $\pi$  be an irreducible automorphic representation of  $\mathbf{G}(\mathbb{A})$ , and suppose that there is a finite place u of k such that  $\mathbf{G}(k_u)$  is split and  $\pi_u$  is cuspidal. There exists a unique conjugate-self-dual cuspidal automorphic representation  $\Pi$  of  $\mathrm{GL}_N(\mathbb{A}_l)$  such that

- (1) for all finite places v of k, the local transfer of  $\pi_v$  to  $GL_N(l_v)$  is  $\Pi_v$ ,
- (2) for all real places v of k, the infinitesimal character of  $\Pi_v$  is algebraic regular.

*Proof.* — Since **G** is compact at all real places, the assumptions of [**36**] Corollaire 5.3 are satisfied (see the paragraph following [**36**] Remarque 5.2 regarding Property (\*)). By [**36**] Corollaire 5.3, there is an integer  $r \ge 1$  and, for each  $i \in \{1, \ldots, r\}$ , there is a conjugate-self-dual discrete automorphic representation  $\Pi_i$  of  $\operatorname{GL}_{N_i}(\mathbb{A}_l)$  for some  $N_i \ge 1$ , such that

- one has  $N_1 + \cdots + N_r = N$ ,

- if  $\Pi$  is the irreducible automorphic representation of  $\operatorname{GL}_N(\mathbb{A}_l)$  obtained by parabolic induction from  $\Pi_1 \otimes \cdots \otimes \Pi_r$ , then  $\Pi_v$  is the local transfer of  $\pi_v$  for all finite places v which are either unramified or split. (The local base change of [**36**] is the same as the local transfer of Paragraph 5.4: see [**36**] 4.10.)

In particular, for v = u, the group  $\mathbf{G}(k_u)$  is split, thus  $\Pi_u$  is isomorphic to  $\pi_u \otimes \pi_u^{\vee}$  via the choice of a  $k_u$ -algebra isomorphism  $l_u \simeq k_u \times k_u$  (see Remark 5.2). Since  $\pi_u$  is cuspidal,  $\Pi_u$  is cuspidal as well. It follows that r = 1 and  $\psi$  is cuspidal. By [**36**] Théorème 5.9, we get that

•  $\Pi_v$  is the base change of the trivial character of  $\mathbf{G}(k_v)$ , thus its infinitesimal character is algebraic regular, for all real places v of k,

• and the local transfer of  $\pi_v$  to  $\operatorname{GL}_N(l_v)$  is  $\Pi_v$  for all finite places v of k.

This finishes the proof of Theorem 5.6.

#### 6. Unramified local transfer

In this section, we examine the congruence properties of the local transfer (as defined in Paragraph 5.4) for unramified representations of unramified classical groups.

6.1.

Let F be a non-Archimedean locally compact field of residue characteristic p, and G be the group of rational points of an unramified reductive group  $\mathbf{G}$  defined over F. Let  $\mathbf{S}$  be a maximal F-split torus in  $\mathbf{G}$ ,  $\mathbf{T}$  be the centralizer of  $\mathbf{S}$  in  $\mathbf{G}$  and K be a hyperspecial maximal compact subgroup of G corresponding to a hyperspecial point in the apartment associated with  $\mathbf{S}$  in the reduced Bruhat-Tits building of  $(\mathbf{G}, F)$ . Let W be the Weyl group associated with  $T = \mathbf{T}(F)$  and  $\Lambda$  be the  $\mathbb{Z}$ -lattice  $T/(T \cap K)$ . We have the Satake isomorphism ([53]) of  $\mathbb{C}$ -algebras

$$\mathbb{C}[K \backslash G/K] \to \mathbb{C}[\Lambda]^W$$

$$f \mapsto \left( t \mapsto \delta^{1/2}(t) \int_U f(tu) \, \mathrm{d}u \right)$$

where U is the group of rational points of the unipotent radical of a Borel subgroup  $\mathbf{B} = \mathbf{TU}$  of  $\mathbf{G}$ , du is the Haar measure on U giving measure 1 to  $U \cap K$  and  $\delta^{1/2}$  is the square root of the modulus character  $\delta$  of  $B = \mathbf{B}(F)$  defined with respect to the positive square root  $\sqrt{q} \in \mathbb{R}_{>0}$  of q, the cardinality of the residue field of F.

The same formula applies when one replaces  $\mathbb{C}$  by  $\overline{\mathbb{Q}}_{\ell}$ . We then get a Satake isomorphism of  $\overline{\mathbb{Q}}_{\ell}$ -algebras  $\overline{\mathbb{Q}}_{\ell}[K \setminus G/K] \to \overline{\mathbb{Q}}_{\ell}[\Lambda]^W$  depending on the choice of a square root  $q^{1/2}$  of q in  $\overline{\mathbb{Q}}_{\ell}$ . By [28] §7.10–15, as this square root and its inverse are contained in  $\overline{\mathbb{Z}}_{\ell}$ , this isomorphism induces by restriction an isomorphism

(6.1) 
$$\overline{\mathbb{Z}}_{\ell}[K \backslash G/K] \to \overline{\mathbb{Z}}_{\ell}[\Lambda]^{W}$$

of  $\overline{\mathbb{Z}}_{\ell}$ -algebras.

#### 6.2.

Let  $\pi$  be a K-unramified irreducible  $\overline{\mathbb{Q}}_{\ell}$ -representation of G, that is,  $\pi$  has a non-zero K-fixed vector. Recall that its *Satake parameter* is the character  $\chi$  of  $\overline{\mathbb{Q}}_{\ell}[K \setminus G/K]$  through which this algebra acts on the 1-dimensional space  $\pi^K$  of K-invariant vectors of  $\pi$ . Through the Satake isomorphism, it defines a character of  $\overline{\mathbb{Q}}_{\ell}[\Lambda]^W$ . Such a character is of the form

(6.2) 
$$f \mapsto \int_{T} f(t)\omega(t) \, \mathrm{d}t$$

for some unramified  $\overline{\mathbb{Q}}_{\ell}$ -character  $\omega$  of T – which we may consider as a character of  $\Lambda$  – uniquely determined up to W-conjugacy. (Here dt is the Haar measure giving measure 1 to  $T \cap K$ .) By [53], the W-conjugacy class of  $\omega$  is the cuspidal support of  $\pi$ , that is,  $\pi$  occurs as an irreducible component of the representation obtained by parabolically inducing  $\omega$  to G along B, where parabolic induction is normalized by the same square root of the  $\overline{\mathbb{Q}}_{\ell}$ -modulus  $\delta$  as the one used to define the Satake  $\overline{\mathbb{Q}}_{\ell}$ -isomorphism.

Now assume that the restriction of  $\chi$  to  $\overline{\mathbb{Z}}_{\ell}[K \setminus G/K]$  has values in  $\overline{\mathbb{Z}}_{\ell}$ . Thanks to (6.1), it defines a  $\overline{\mathbb{Z}}_{\ell}$ -character of  $\overline{\mathbb{Z}}_{\ell}[\Lambda]^W$ , still denoted  $\chi$ . Let us prove that  $\omega$  has values in  $\overline{\mathbb{Z}}_{\ell}^{\times}$ . For this, let  $\mu$  be the  $\overline{\mathbb{Q}}_{\ell}$ -character of  $\overline{\mathbb{Q}}_{\ell}[\Lambda]$  defined by (6.2). Its restriction to  $\overline{\mathbb{Z}}_{\ell}[\Lambda]^W$  is equal to  $\chi$ . According

to [8] Chapter 5, §1, n°9, Proposition 22, the ring  $\overline{\mathbb{Z}}_{\ell}[\Lambda]$  is integral over  $\overline{\mathbb{Z}}_{\ell}[\Lambda]^{W}$ . As  $\chi$  takes values in  $\overline{\mathbb{Z}}_{\ell}$  on  $\overline{\mathbb{Z}}_{\ell}[\Lambda]^{W}$ , and as  $\overline{\mathbb{Z}}_{\ell}$  is integrally closed, it follows that  $\mu$  takes values in  $\overline{\mathbb{Z}}_{\ell}$  on  $\overline{\mathbb{Z}}_{\ell}[\Lambda]$ . By evaluating  $\mu$  at the characteristic function of any  $\lambda \in \Lambda$ , we get  $\omega(\lambda) \in \overline{\mathbb{Z}}_{\ell}$ . So far, we proved the following result.

**Proposition 6.1.** — Let G be the group of rational points of an unramified group defined over F, let K be a hyperspecial maximal compact subgroup of G and  $\pi$  be a K-unramified  $\overline{\mathbb{Q}}_{\ell}$ -representation of G with Satake parameter  $\chi$ . Then  $\pi$  is integral if and only if  $\chi$  is integral (that is, it takes integral values on  $\overline{\mathbb{Z}}_{\ell}[K \setminus G/K]$ ).

*Proof.* — Indeed, using the notation above, the cuspidal support of  $\pi$  is the *W*-conjugacy class of the unramified character  $\omega$  of *T*, and  $\pi$  is integral if and only if  $\omega$  is. (For this latter fact, see [18] Corollary 1.6.)

Finally, assume that  $\chi_1$  and  $\chi_2$  are congruent  $\overline{\mathbb{Z}}_{\ell}$ -characters of  $\overline{\mathbb{Z}}_{\ell}[K \setminus G/K]$ . One can see them via (6.1) as congruent  $\overline{\mathbb{Z}}_{\ell}$ -characters of  $\overline{\mathbb{Z}}_{\ell}[\Lambda]^W$ , still denoted  $\chi_1$  and  $\chi_2$ . For i = 1, 2, let  $\mu_i$  be a character of  $\overline{\mathbb{Z}}_{\ell}[\Lambda]$  extending  $\chi_i$ . It takes the form (6.2) for a uniquely determined unramified character  $\omega_i$  of T, which is integral thanks to the previous paragraph. Reducing mod the maximal ideal of  $\overline{\mathbb{Z}}_{\ell}$ , the characters  $\mu_1$  and  $\mu_2$  define  $\overline{\mathbb{F}}_{\ell}$ -characters  $\overline{\mu}_1$  and  $\overline{\mu}_2$  of  $\overline{\mathbb{F}}_{\ell}[\Lambda]$  which, by assumption, coincide on  $\overline{\mathbb{F}}_{\ell}[\Lambda]^W$ . Applying the corollary of [8] Chapter 5, §2, n°2, Theorem 2, it follows that the characters  $\mathbf{r}_{\ell}(\omega_1)$  and  $\mathbf{r}_{\ell}(\omega_2)$  are W-conjugate. We thus proved:

**Proposition 6.2.** Let G be the group of rational points of an unramified group defined over F, let K be a hyperspecial maximal compact subgroup of G, let  $\pi_1$  and  $\pi_2$  be K-unramified irreducible  $\overline{\mathbb{Q}}_{\ell}$ -representations of G whose Satake parameters  $\chi_1$  and  $\chi_2$  define congruent  $\overline{\mathbb{Z}}_{\ell}$ -characters of  $\overline{\mathbb{Z}}_{\ell}[K \setminus G/K]$  and let  $\omega_1$  and  $\omega_2$  be unramified  $\overline{\mathbb{Q}}_{\ell}$ -characters of T such that  $\pi_i$  occurs in the parabolic induction of  $\omega_i$  to G along B, for i = 1, 2. Then  $\mathbf{r}_{\ell}(\omega_1)$  and  $\mathbf{r}_{\ell}(\omega_2)$  are W-conjugate.

#### 6.3.

From now on and until the end of this section, we assume that **G** is an unramified special orthogonal, unitary or symplectic group among the groups of Paragraph 5.1. The associated dual group  $\hat{\mathbf{G}}$  has been defined in Paragraph 5.2. Recall that  $G = \mathbf{G}(F)$ .

Let  $\pi$  be an integral K-unramified  $\overline{\mathbb{Q}}_{\ell}$ -representation of G. Its cuspidal support is the W-orbit of an unramified  $\overline{\mathbb{Z}}_{\ell}$ -character  $\omega$  of T. Its Satake parameter is a character  $\chi : \overline{\mathbb{Z}}_{\ell}[K \setminus G/K] \to \overline{\mathbb{Z}}_{\ell}$ . They are related through the Satake isomorphism by the formula (6.2).

Restriction from T to  $S = \mathbf{S}(F)$  induces an isomorphism  $\Lambda \simeq S/(S \cap K)$ , thus between unramified characters of T and unramified characters of S. The later is the dual group  $\widehat{\mathbf{S}}(\overline{\mathbb{Q}}_{\ell})$ .

Let  $\Phi$  be a Frobenius element in the Weil group  $W_F$ . By [6] 6.4, 6.5, the surjection of  $\widehat{\mathbf{T}}(\overline{\mathbb{Q}}_{\ell})$  onto  $\widehat{\mathbf{S}}(\overline{\mathbb{Q}}_{\ell})$  induces a bijection between

- N-conjugacy classes in  $\widehat{\mathbf{T}}(\overline{\mathbb{Q}}_{\ell}) \rtimes \Phi$ , and

- W-conjugacy classes in  $\widehat{\mathbf{S}}(\overline{\mathbb{Q}}_{\ell})$ ,

where N is the inverse image of W in the normalizer of  $\widehat{\mathbf{T}}(\overline{\mathbb{Q}}_{\ell})$  in  $\widehat{\mathbf{G}}(\overline{\mathbb{Q}}_{\ell})$ , and the embedding of of  $\widehat{\mathbf{T}}(\overline{\mathbb{Q}}_{\ell})$  in  $\widehat{\mathbf{G}}(\overline{\mathbb{Q}}_{\ell})$  induces a bijection between

- N-conjugacy classes in  $\widehat{\mathbf{T}}(\overline{\mathbb{Q}}_{\ell}) \rtimes \Phi$ , and
- $-\widehat{\mathbf{G}}(\overline{\mathbb{Q}}_{\ell})$ -conjugacy classes of semi-simple elements in  $\widehat{\mathbf{G}}(\overline{\mathbb{Q}}_{\ell}) \rtimes \Phi$ .

The W-orbit of  $\omega$  thus determines the W-conjugacy class of a point  $s \in \widehat{\mathbf{S}}(\overline{\mathbb{Z}}_{\ell})$ , then the  $\widehat{\mathbf{G}}(\overline{\mathbb{Q}}_{\ell})$ conjugacy class of a semi-simple element  $t \rtimes \Phi \in \widehat{\mathbf{G}}(\overline{\mathbb{Q}}_{\ell}) \rtimes \Phi$ . We are going to prove that  $t \rtimes \Phi$ may be chosen in  $\widehat{\mathbf{T}}(\overline{\mathbb{Z}}_{\ell}) \rtimes \Phi \subseteq \widehat{\mathbf{G}}(\overline{\mathbb{Z}}_{\ell}) \rtimes \Phi$ . Let us fix a uniformizer  $\varpi$  of F.

When **G** is split, we have  $\mathbf{T} = \mathbf{S}$ , thus t = s is in  $\widehat{\mathbf{T}}(\overline{\mathbb{Z}}_{\ell}) \subseteq \widehat{\mathbf{G}}(\overline{\mathbb{Z}}_{\ell})$ . Explicitly, if we identify T with  $(F^{\times})^m$  for some integer  $m \ge 1$ , then  $\omega$  identifies with the tensor product of m unramified characters  $\omega_1, \ldots, \omega_m$  of  $F^{\times}$  and  $t \rtimes \Phi$  is  $\widehat{\mathbf{G}}(\overline{\mathbb{Q}}_{\ell})$ -conjugate to

diag(ω<sub>1</sub>(ϖ),...,ω<sub>m</sub>(ϖ),1,ω<sub>m</sub>(ϖ)<sup>-1</sup>,...,ω<sub>1</sub>(ϖ)<sup>-1</sup>) ∈ GL<sub>2n+1</sub>(ℤ<sub>ℓ</sub>) if G = Sp<sub>2n</sub>,
diag(ω<sub>1</sub>(ϖ),...,ω<sub>m</sub>(ϖ),ω<sub>m</sub>(ϖ)<sup>-1</sup>,...,ω<sub>1</sub>(ϖ)<sup>-1</sup>) ∈ GL<sub>2n</sub>(ℤ<sub>ℓ</sub>) if G = SO<sub>2n+1</sub> or G = SO<sup>1</sup><sub>2n</sub>, with m = n in all cases.

Now assume that **G** is non-split, thus either  $\mathbf{G} = SO_{2n}^{\alpha}$  or  $\mathbf{G} = U_n^{\alpha}$ , with  $\alpha \neq 1$ .

• In the even orthogonal case, we have  $S \simeq (F^{\times})^m$  and  $T \simeq S \times \mathrm{SO}_2^{\alpha}(F)$  with m = n - 1 (see [7] §23.4), thus  $\widehat{\mathbf{T}}(\overline{\mathbb{Q}}_{\ell}) \simeq \overline{\mathbb{Q}}_{\ell}^{m+1}$  surjects onto  $\widehat{\mathbf{S}}(\overline{\mathbb{Q}}_{\ell}) \simeq \overline{\mathbb{Q}}_{\ell}^{m}$  through

$$(t_1, t_2, \ldots, t_{m+1}) \mapsto (t_1, t_2, \ldots, t_m)$$

and  $\widehat{\mathbf{T}}(\overline{\mathbb{Q}}_{\ell}) \rtimes W_F$  embeds into  $\widehat{\mathbf{G}}(\overline{\mathbb{Q}}_{\ell}) \rtimes W_F$  through

$$(t_1, t_2, \dots, t_n) \rtimes w \mapsto \operatorname{diag}(t_1, \dots, t_n, t_n^{-1}, \dots, t_1^{-1}) \rtimes w$$

thus the image of  $t \rtimes \Phi$  in  $\widehat{\mathbf{G}}(\overline{\mathbb{Q}}_{\ell}) \rtimes \Phi$  is  $\widehat{\mathbf{G}}(\overline{\mathbb{Q}}_{\ell})$ -conjugate to

diag
$$(\omega_1(\varpi),\ldots,\omega_m(\varpi),1,1,\omega_m(\varpi)^{-1},\ldots,\omega_1(\varpi)^{-1})$$
  $\rtimes \Phi \in \operatorname{GL}_{2n}(\overline{\mathbb{Z}}_{\ell}) \rtimes \Phi.$ 

• In the unitary case, we have  $S \simeq (F^{\times})^m$  and  $T \simeq (E^{\times})^m$  where E is the quadratic extension of F generated by a square root of  $\alpha$  (note that it is unramified since G is assumed to be unramified), and  $m = \lfloor n/2 \rfloor$  is the Witt index of **G** (see [7] §23.9), thus  $\widehat{\mathbf{T}}(\overline{\mathbb{Q}}_{\ell}) \simeq \overline{\mathbb{Q}}_{\ell}^{2m}$  surjects onto  $\widehat{\mathbf{S}}(\overline{\mathbb{Q}}_{\ell}) \simeq \overline{\mathbb{Q}}_{\ell}^m$  through

$$(t_1, t_2, \dots, t_{2m}) \mapsto (t_1 t_{2m}, t_2 t_{2m-1}, \dots, t_m t_{m+1})$$

and  $\widehat{\mathbf{T}}(\overline{\mathbb{Q}}_{\ell}) \rtimes W_F$  embeds into  $\widehat{\mathbf{G}}(\overline{\mathbb{Q}}_{\ell}) \rtimes W_F$  through

$$(t_1, t_2, \dots, t_{2m}) \rtimes w \mapsto \begin{cases} \operatorname{diag}(t_1, \dots, t_m, t_{m+1}^{-1}, \dots, t_{2m}^{-1}) \rtimes w & \text{if } n = 2m \text{ is even,} \\ \operatorname{diag}(t_1, \dots, t_m, 1, t_{m+1}^{-1}, \dots, t_{2m}^{-1}) \rtimes w & \text{if } n = 2m + 1 \text{ is odd,} \end{cases}$$

thus the image of  $t \rtimes \Phi$  in  $\widehat{\mathbf{G}}(\overline{\mathbb{Q}}_{\ell}) \rtimes \Phi$  is  $\widehat{\mathbf{G}}(\overline{\mathbb{Q}}_{\ell})$ -conjugate to

$$\begin{cases} \operatorname{diag}(\omega_1(\varpi)^{1/2},\ldots,\omega_m(\varpi)^{1/2},\omega_m(\varpi)^{-1/2},\ldots,\omega_1(\varpi)^{-1/2}) \rtimes \Phi & \text{if } n = 2m \text{ is even,} \\ \operatorname{diag}(\omega_1(\varpi)^{1/2},\ldots,\omega_m(\varpi)^{1/2},1,\omega_m(\varpi)^{-1/2},\ldots,\omega_1(\varpi)^{-1/2}) \rtimes \Phi & \text{if } n = 2m+1 \text{ is odd,} \end{cases}$$

which is in  $\operatorname{GL}_n(\overline{\mathbb{Z}}_\ell) \rtimes \Phi$  in both cases.

We now define an unramified local Langlands parameter  $\varphi : WD_F \to \widehat{\mathbf{G}}(\overline{\mathbb{Z}}_{\ell}) \rtimes W_F$  by

- $-\varphi(\Phi) = t \rtimes \Phi$ , and
- $-\varphi$  is trivial on the inertia subgroup  $I_F$  of  $W_F$  and on  $SL_2(\mathbb{C})$ .

It is uniquely determined by the K-unramified representation  $\pi$ , or equivalently by its Satake parameter  $\chi$ . Composing with Std (or just restricting to WD<sub>E</sub> in the unitary case), we get an unramified Langlands parameter  $\phi \in \Phi(\text{GL}_N, E)$ , where E = F in the symplectic and orthogonal cases and E is the quadratic extension of F generated by a square root of  $\alpha$  in the unitary case. This  $\phi$  uniquely determines an unramified  $\overline{\mathbb{Q}}_{\ell}$ -representation of  $\text{GL}_N(E)$ , denoted  $\mathbf{t}_{\ell}(\pi)$ .

In the symplectic and special orthogonal cases,  $\mathbf{t}_{\ell}(\pi)$  is the unique unramified irreducible component of

$$\begin{cases} \omega_1 \times \cdots \times \omega_n \times \omega_n^{-1} \times \cdots \times \omega_1^{-1} & \text{if } \mathbf{G} = \mathrm{SO}_{2n+1} \text{ or } \mathbf{G} = \mathrm{SO}_{2n}^1, \\ \omega_1 \times \cdots \times \omega_n \times 1 \times \omega_n^{-1} \times \cdots \times \omega_1^{-1} & \text{if } \mathbf{G} = \mathrm{Sp}_{2n}, \\ \omega_1 \times \cdots \times \omega_{n-1} \times 1 \times 1 \times \omega_{n-1}^{-1} \times \cdots \times \omega_1^{-1} & \text{if } \mathbf{G} = \mathrm{SO}_{2n}^{\alpha} \text{ with } \alpha \neq 1, \end{cases}$$

where  $\times$  denotes the parabolic induction to  $\operatorname{GL}_N(F)$  normalized with respect to  $q^{1/2}$ .

In the unitary case, the Weil group  $W_E$  is generated by  $\Phi^2$  and  $I_F$  (since E/F is unramified), thus  $\phi$  is uniquely determined by  $\phi(\Phi^2) = tt^* \rtimes \Phi^2$ , with

$$tt^* = \begin{cases} \operatorname{diag}(\omega_1(\varpi), \dots, \omega_m(\varpi), \omega_m(\varpi)^{-1}, \dots, \omega_1(\varpi)^{-1}) & \text{if } n = 2m \text{ is even,} \\ \operatorname{diag}(\omega_1(\varpi), \dots, \omega_m(\varpi), 1, \omega_m(\varpi)^{-1}, \dots, \omega_1(\varpi)^{-1}) & \text{if } n = 2m + 1 \text{ is odd,} \end{cases}$$

which gives  $\mathbf{t}_{\ell}(\pi)$  explicitly. Namely,  $\mathbf{t}_{\ell}(\pi)$  is the unique unramified irreducible component of

$$\begin{cases} \omega_1 \times \cdots \times \omega_m \times \omega_m^{-1} \times \cdots \times \omega_1^{-1} & \text{if } \mathbf{G} = \mathbf{U}_{2m}^{\alpha}, \\ \omega_1 \times \cdots \times \omega_m \times 1 \times \omega_m^{-1} \times \cdots \times \omega_1^{-1} & \text{if } \mathbf{G} = \mathbf{U}_{2m+1}^{\alpha} \end{cases}$$

where  $\times$  denotes the parabolic induction to  $\operatorname{GL}_n(E)$  normalized with respect to  $(q^{1/2})^2 = q$  (as E is quadratic and unramified over F). (See also [38].) We have:

**Proposition 6.3.** — Let G be the group of rational points of an unramified special orthogonal, unitary or symplectic F-group among the groups of Paragraph 5.1. Let K be a hyperspecial maximal compact subgroup of G and let  $\pi_1$ ,  $\pi_2$  be K-unramified irreducible  $\overline{\mathbb{Q}}_{\ell}$ -representations of G whose Satake parameters  $\chi_1$ ,  $\chi_2$  define congruent  $\overline{\mathbb{Z}}_{\ell}$ -characters of  $\overline{\mathbb{Z}}_{\ell}[K\backslash G/K]$ . Then

- (1) the representations  $\mathbf{t}_{\ell}(\pi_1)$  and  $\mathbf{t}_{\ell}(\pi_2)$  of  $\operatorname{GL}_N(E)$  are integral,
- (2) their Langlands parameters are integral and congruent.

**Remark 6.4.** — Note that the reductions mod  $\ell$  of  $\mathbf{t}_{\ell}(\pi_1)$  and  $\mathbf{t}_{\ell}(\pi_2)$  may not have any irreducible component in common. However, if  $\tau_i$  denotes the unique unramified irreducible component of  $\mathbf{r}_{\ell}(\mathbf{t}_{\ell}(\pi_i))$  for i = 1, 2, then  $\tau_1$  and  $\tau_2$  have the same cuspidal support.

For instance, assume that  $\mathbf{G} = \mathrm{SO}_5$  and  $\ell$  divides  $q^2 - 1$ . Let  $\pi_1$  (resp.  $\pi_2$ ) be the unramified representation of  $G = \mathrm{SO}_5(F)$  with respect to some hyperspecial maximal compact group, with cuspidal support the *W*-conjugacy class of  $|\cdot|^{-1/2} \otimes |\cdot|^{1/2}$  (resp.  $|\cdot|^{3/2} \otimes |\cdot|^{1/2}$ ), where  $|\cdot|$  is the absolute value of  $F^{\times}$ . By assumption, these cuspidal supports are congruent (for  $|\cdot|^{3/2}$  and  $|\cdot|^{-1/2}$  have the same reduction mod  $\ell$ ). Then  $\mathbf{t}_{\ell}(\pi_1)$  is the unique unramified irreducible component of

$$|\cdot|^{-1/2} \times |\cdot|^{1/2} \times |\cdot|^{-1/2} \times |\cdot|^{1/2}$$

that is,  $\mathbf{t}_{\ell}(\pi_1) = 1_2 \times 1_2$  (where  $1_2$  is the trivial character of  $\operatorname{GL}_2(F)$ ). Similarly,  $\mathbf{t}_{\ell}(\pi_2)$  is equal to  $|\det| \times |\det|$  (where det is the determinant of  $\operatorname{GL}_2(F)$ ). Now assume further that  $\ell \neq 2$  and

 $\mathbf{34}$ 

 $\ell$  divides q + 1, thus  $\ell$  does not divide q - 1. Then  $\mathbf{r}_{\ell}(\mathbf{t}_{\ell}(\pi_1))$  and  $\mathbf{r}_{\ell}(\mathbf{t}_{\ell}(\pi_2))$  are both irreducible and twists of each other by the non-trivial character  $|\cdot|$ , thus non-isomorphic.

**Remark 6.5.** — Let  $\iota$  be an isomorphism of fields  $\mathbb{C} \to \overline{\mathbb{Q}}_{\ell}$  taking the positive square root of q in  $\mathbb{R}$  to the square root  $q^{1/2} \in \overline{\mathbb{Q}}_{\ell}$  of Paragraph 6.1. According to Arthur [2] 6.1 (see p. 304) and Mok [47] 7.1 (see also Labesse [36] p. 38-39), which describe the local transfer map **t** of Definition 5.1 for unramified representations of unramified groups, we have

$$\mathbf{t}(\pi) \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_{\ell} = \mathbf{t}_{\ell}(\pi \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_{\ell})$$

for any K-unramified complex representation  $\pi$  of G, where tensor products are taken with respect to  $\iota$ . Proposition 6.7 and Remark 6.8 will describe the dependency of  $\mathbf{t}_{\ell}$  in  $q^{1/2}$ .

#### 6.4.

In this paragraph, G is an unramified classical group and K is a hyperspecial maximal compact subgroup of G as in Paragraph 6.3.

Unlike Paragraph 6.3 however, we will consider *complex* representations rather than  $\overline{\mathbb{Q}}_{\ell}$ -representations. We examine the dependency of the unramified transfer from G to  $\operatorname{GL}_N(E)$  with respect to the choice of a square root of q, and to the action of  $\operatorname{Aut}(\mathbb{C})$ . This will be useful in Paragraph 9.2.

Let  $\pi$  be a K-unramified irreducible complex representation of G. Associated with  $\pi$ , there is the W-conjugacy class of an unramified character  $\omega$  of T, such that  $\pi$  is the unique K-unramified irreducible component of the normalized parabolic induction of  $\omega$  to G along B. Writing  $\text{Ind}_B^G$ for unnormalized parabolic induction from T to G along B and  $i_B^G$  for normalized parabolic induction, we have

$$i_B^G(\omega) = \operatorname{Ind}_B^G(\delta^{1/2}\omega).$$

Let us write  $T \simeq E^{\times m} \times T_0$ , where

- $T_0$  is trivial if G is split or  $G \simeq U_{2m}^{\alpha}(F)$  with  $\alpha \neq 1$ ,
- $T_0 = \mathrm{SO}_2^{\alpha}(F)$  if  $G \simeq \mathrm{SO}_{2n}^{\alpha}(F)$  with  $\alpha \neq 1$ ,
- $T_0 = \mathrm{U}_1^{\alpha}(F)$  if  $G \simeq \mathrm{U}_{2m+1}^{\alpha}(F)$  with  $\alpha \neq 1$ .

The character  $\omega$  can thus be written  $\omega_1 \otimes \cdots \otimes \omega_m \otimes 1$ , where each  $\omega_i$  is an unramified character of  $E^{\times}$  and 1 is the trivial character of  $T_0$ . By [46] IV.4 p. 69, the modulus character of the parabolic subgroup  $P \supseteq B$  with Levi component  $\operatorname{GL}_m(E) \times T_0$  is equal to

$$\nu_m^{d-m-e} \otimes 1$$

where  $\nu_m$  is the unramified character "absolute value of the determinant" of  $GL_m(E)$  and

- d = 2n and e = -1 if  $G = \text{Sp}_{2n}(F)$ ,
- d = 2n + 1 and e = 1 if  $G = SO_{2n+1}(F)$ ,
- d = 2n and e = 1 if  $G = SO_{2n}^{\alpha}(F)$ ,
- d = n and e = 0 if  $G = U_n^{\alpha}(F)$  with  $\alpha \neq 1$ .

By using the transivity property of parabolic induction, we deduce that

$$\begin{split} \delta^{1/2} &= |\cdot|_{E}^{(d-m-e)/2 + (m-1)/2} \otimes \cdots \otimes |\cdot|_{E}^{(d-m-e)/2 - (m-1)/2} \otimes 1 \\ &= |\cdot|_{E}^{(d-e-1)/2} \otimes \cdots \otimes |\cdot|_{E}^{(d-e-1)/2 - m+1} \otimes 1 \end{split}$$

(where  $|\cdot|_E$  is the absolute value of E). Replacing  $\sqrt{q}$  by the opposite square root changes  $|\cdot|_E$  to  $\eta|\cdot|_E$ , where  $\eta$  is the unramified character of  $E^{\times}$  of order 2. It thus changes  $\omega_i$  to  $\omega_i \eta^{(d-e-1)f}$ , where f is the residual degree of E over F (which is 1 or 2 depending whether E = F or not).

Similarly, replacing  $\sqrt{q}$  by its opposite square root has the effect of twisting normalized parabolic induction from  $E^{\times N}$  to  $\operatorname{GL}_N(E)$  (along the Borel subgroup made of upper triangular matrices) by  $\eta^{(1-N)f}$ .

Consequently, considering the explicit formulas of Paragraph 6.3, replacing  $\sqrt{q}$  by the opposite square root has the effect of twisting  $\mathbf{t}(\pi)$  by the character  $\eta^{(d-e-N)f}$ . We have

- d e N = 2n + 1 (2n + 1) = 0 if  $G = \operatorname{Sp}_{2n}(F)$ ,
- d e N = (2n + 1) 1 2n = 0 if  $G = SO_{2n+1}(F)$ ,
- d e N = 2n 1 2n = -1 if  $G = SO_{2n}^{\alpha}(F)$ ,

and f = 2 if G is unitary. The integer (d - e - N)f is thus even, except if G is even orthogonal.

**Example 6.6.** — If  $G = SO_2^1(F) \simeq F^{\times}$ , the transfer of any unramified character  $\omega$  of  $F^{\times}$  is the unique unramified irreducible component of  $\omega \times \omega^{-1}$ . If G is the compact group  $SO_2^{\alpha}(F)$  with  $\alpha \neq 1$ , the transfer of the trivial character of G is  $1 \times 1$ . In both cases, the transfer depends on the choice of a square root of q.

Now consider an automorphism  $\gamma \in \operatorname{Aut}(\mathbb{C})$ . Given a representation  $\pi$  of a group H on a complex vector space V, we write  $\pi^{\gamma}$  for the representation of H on  $V \otimes_{\mathbb{C}} \mathbb{C}_{\gamma}$ , where  $\mathbb{C}_{\gamma}$  is the field  $\mathbb{C}$ considered as a  $\mathbb{C}$ -algebra via  $\gamma$ . Consider the map  $\pi \mapsto \mathbf{t}(\pi^{\gamma^{-1}})^{\gamma}$  from K-unramified irreducible representations of G to irreducible representations of  $\operatorname{GL}_N(E)$ . It is the unramified local transfer map from G to  $\operatorname{GL}_N(E)$  with respect to the square root  $\gamma(\sqrt{q})$ . We thus have:

**Proposition 6.7.** — Let G and K be as above, and let  $\pi$  be a K-unramified irreducible representation of G. Let  $\gamma \in Aut(\mathbb{C})$ .

(1) If G is not even orthogonal, then  $\mathbf{t}(\pi^{\gamma}) = \mathbf{t}(\pi)^{\gamma}$ .

(2) If G is even orthogonal, then  $\mathbf{t}(\pi^{\gamma}) = \mathbf{t}(\pi)^{\gamma} \cdot \varepsilon_{\gamma}$ , where  $\varepsilon_{\gamma}$  is the unramified character

$$x \mapsto \left(\frac{\gamma(\sqrt{q})}{\sqrt{q}}\right)^{\operatorname{val}_F(x)}$$

of  $F^{\times}$ .

**Remark 6.8.** — We now go back to  $\overline{\mathbb{Q}}_{\ell}$ -representations. We deduce that the map

(6.3) 
$$\pi \mapsto \mathbf{t}_{\ell}(\pi)$$

from (isomorphism classes of) unramified  $\overline{\mathbb{Q}}_{\ell}$ -representations of G to those of  $\operatorname{GL}_N(E)$  is insensitive to the choice of a square root of q in  $\overline{\mathbb{Q}}_{\ell}$ , except when G is even orthogonal, in which case changing this square root to its opposite has the effect of twisting (6.3) by  $\eta$ .

36

#### 7. Representations of local Galois and Weil groups

In this section, F is a p-adic field. We write  $\Gamma$  for the Galois group  $\operatorname{Gal}(\overline{\mathbb{Q}}_p/F)$  and W for the associated Weil group, considered as a subgroup of  $\Gamma$ . It is endowed with a smooth character  $w \mapsto |w|$  with kernel I, the inertia subgroup of W, taking any geometric Frobenius element to  $q^{-1}$ , where q is the cardinality of the residue field of F.

All representations of  $\Gamma$  and W considered in this section will be finite-dimensional. Let  $\ell$  be a prime number different from p.

#### 7.1.

For this paragraph, the reader may refer to [10] Chapter 7 and [61] 4.2.

If  $\sigma$  is a smooth representation of  $\Gamma$ , then its restriction  $\sigma|_{W}$  to W is smooth.

Restriction from  $\Gamma$  to W induces an injection from isomorphism classes of irreducible smooth representations of  $\Gamma$  to isomorphism classes of irreducible smooth representations of W. The image is made of those representations of W whose determinant has finite order (see [10] 28.6 Proposition).

If  $\rho$  is a smooth  $\ell$ -adic representation (that is,  $\overline{\mathbb{Q}}_{\ell}$ -representation) of W on a vector space V, and if  $\Phi \in W$  is a Frobenius element, the following assertions are equivalent:

- (1)  $\rho$  is semi-simple,
- (2)  $\rho(\Phi)$  is a semi-simple element in GL(V),
- (3)  $\rho(w)$  is a semi-simple element in GL(V) for any  $w \in W$

(see **[10]** 28.7 Proposition).

If  $\sigma$  is a continuous  $\ell$ -adic representation of  $\Gamma$ , its restriction to W is a continuous  $\ell$ -adic representation of W, which is irreducible if and only if  $\sigma$  is irreducible.

Fix a continuous surjective group homomorphism

$$(7.1) t: I \to \mathbb{Z}_{\ell}.$$

For the following proposition, see [10] 32.5 Theorem and [56] Appendix.

**Proposition 7.1.** — Let  $\sigma$  be a finite-dimensional continuous  $\ell$ -adic representation of W on a  $\overline{\mathbb{Q}}_{\ell}$ -vector space V.

(1) There is a unique nilpotent endomorphism  $N \in End(V)$  such that there is an open subgroup U of the inertia subgroup I such that

$$\sigma(x) = e^{t(x)N}, \quad x \in U.$$

(2) We have  $\sigma(w)N\sigma(w)^{-1} = |w| \cdot N$  for all  $w \in W$ .

Note that N = 0 if and only if  $\sigma$  is smooth.

The subspaces  $\operatorname{Ker}(N^i)$ ,  $i \ge 0$  of V are stable by  $\sigma$ . Thus, if  $\sigma$  is irreducible, then N = 0 and  $\sigma$  is smooth. More generally, a semi-simple representation of  $\Gamma$  is smooth, and its restriction to W is smooth semi-simple (see also [61] 4.2.3).

Fix a Frobenius element  $\Phi \in W$ . Associated with  $\sigma$ , there is a smooth  $\ell$ -adic representation  $\rho$  of W defined by

$$\rho(\Phi^a x) = \sigma(\Phi^a x) e^{-t(x)N}, \quad a \in \mathbb{Z}, \quad x \in I.$$

The pair  $(\rho, N)$  is called the *Deligne representation* of W associated with  $\sigma$ . Up to isomorphism, it does not depend on the choices of t and  $\Phi$  (see [10] 32.6 Theorem).

The element  $\rho(\Phi) = \sigma(\Phi)$  decomposes uniquely in GL(V) as su = us, with s semi-simple and u unipotent. Define a smooth  $\ell$ -adic representation  $\rho^*$  of W by

$$\rho^*(\Phi^a x) = s^a \rho(x), \quad a \in \mathbb{Z}, \quad x \in I.$$

This defines a Deligne representation  $(\rho^*, N)$ , called the *Frobenius-semi-simplification* of  $(\rho, N)$ . By Paragraph 7.1, the representation  $\rho^*$  is a semi-simple smooth representation of W.

#### 7.2.

In this paragraph, if  $\kappa$  is a continuous  $\ell$ -adic representation of W or  $\Gamma$ , we will write  $\kappa$ <sup>ss</sup> for its semisimplification.

**Lemma 7.2.** — If  $\rho$  is a (finite-dimensional) smooth  $\ell$ -adic representation of W such that  $\rho^{ss}$  is integral, then  $\rho$  is integral.

*Proof.* — We prove it by induction on the dimension n of  $\rho$ . If  $\rho$  is irreducible, there is nothing to prove. Otherwise, let  $\tau$  be an irreducible subrepresentation of  $\rho$ , of dimension  $k \ge 1$ , and let  $\lambda$  be the quotient of  $\rho$  by  $\tau$ , which is of dimension l = n - k. Since  $\rho^{ss} = \tau \oplus \lambda^{ss}$ , we may apply the inductive hypothesis to  $\lambda$ , from which we deduce that  $\lambda$  is integral. We therefore fix a basis of the vector space of  $\rho$  such that

$$\rho(w) = \begin{pmatrix} \tau(w) & \alpha(w) \\ 0 & \lambda(w) \end{pmatrix} \in \operatorname{GL}_n(\overline{\mathbb{Q}}_\ell), \quad \tau(w) \in \operatorname{GL}_k(\overline{\mathbb{Z}}_\ell), \quad \lambda(w) \in \operatorname{GL}_l(\overline{\mathbb{Z}}_\ell), \quad w \in W,$$

and  $\alpha : W \to \mathbf{M}_{k,l}(\overline{\mathbb{Q}}_{\ell})$  satisfies  $\alpha(xy) = \tau(x)\alpha(y) + \alpha(x)\lambda(y)$  for all  $x, y \in W$ .

Since  $\rho$  is smooth, we may consider it as a representation of the discrete group W/U for some open subgroup U of W. Since this quotient is a finitely generated group, we may consider  $\rho$  as a representation of the free group F with r generators  $f_1, \ldots, f_r$  for some  $r \ge 1$ . Assume  $\alpha$  is not identically zero, and let -v denote the minimum of the  $\ell$ -adic valuations of all the entries of all the  $\alpha(f_i)$ . Conjugating  $\rho$  by diag( $\ell^v \cdot id_t, id_l$ ), we may and will assume that v = 0.

We are going to prove that  $\alpha$  takes values in  $\mathbf{M}_{t,l}(\overline{\mathbb{Z}}_{\ell})$ . We prove it by induction on the length of the words in F. Given  $x \in \mathsf{F}$ , write it yf with  $f = \mathsf{f}_i$  for some  $i \in \{1, \ldots, r\}$  and the length of y is smaller than that of x. Then

$$\alpha(x) = \tau(y)\alpha(f) + \alpha(y)\lambda(f) \in \mathbf{M}_{k,l}(\overline{\mathbb{Z}}_{\ell})$$

thanks to the inductive hypothesis.

**Lemma 7.3.** — Let  $\sigma$  be a (finite-dimensional) continuous  $\ell$ -adic representation of  $\Gamma$ , with associated Deligne representation ( $\rho$ , N). The restriction of  $\sigma^{ss}$  to W is equal to  $\rho^{ss}$ .

38

*Proof.* — Note that semi-simplification and restriction from  $\Gamma$  to W commute, that is

$$\sigma^{\rm ss}|_{\rm W} = (\sigma|_{\rm W})^{\rm ss}$$

If N is zero, then  $\sigma$  is smooth and  $\rho$  is the restriction of  $\sigma$  to W, thus  $\sigma^{ss}$  is smooth semi-simple, and its restriction to W is smooth semi-simple as well. Otherwise, if  $n = \dim(\sigma)$ , there is a basis of  $\overline{\mathbb{Q}}_{\ell}^{n}$  such that

$$\sigma(g) = \begin{pmatrix} \alpha(g) & \gamma(g) \\ 0 & \beta(g) \end{pmatrix} \in \operatorname{GL}_n(\overline{\mathbb{Q}}_{\ell}), \quad g \in \Gamma, \quad \text{and} \quad N = \begin{pmatrix} 0 & C \\ 0 & M \end{pmatrix},$$

where

- $-\alpha$  is a smooth  $\ell$ -adic representation of  $\Gamma$  of dimension  $k = \dim(\operatorname{Ker} N)$ ,
- $\beta$  is a continuous  $\ell$ -adic representation of  $\Gamma$  of dimension l = n k,
- $\gamma$  is a continuous map from  $\Gamma$  to  $\mathbf{M}_{k,l}(\mathbb{Q}_{\ell})$ ,
- M is nilpotent in  $\mathbf{M}_l(\overline{\mathbb{Q}}_\ell)$  and C is a matrix in  $\mathbf{M}_{k,l}(\overline{\mathbb{Q}}_\ell)$ .

We have

$$e^{t(x)N} = \begin{pmatrix} \mathrm{id} & \varepsilon(x) \\ 0 & e^{t(x)M} \end{pmatrix} \quad \mathrm{with} \quad \varepsilon(x) = \sum_{i \ge 1} \frac{t(x)^i}{i!} C M^{i-1}, \quad x \in I.$$

Writing  $(\rho_1, N_1)$  for the Deligne representation associated with  $\beta$ , we get  $N_1 = M$  and

$$\rho(w) = \begin{pmatrix} \alpha(w) & \delta(w) \\ 0 & \rho_1(w) \end{pmatrix} \in \operatorname{GL}_n(\overline{\mathbb{Q}}_\ell), \quad w \in \mathbf{W},$$

for some smooth map  $\delta$  from W to  $\mathbf{M}_{k,l}(\overline{\mathbb{Q}}_{\ell})$  which can be explicitly described by

$$\delta(w) = (\gamma(w) - \alpha(w)\varepsilon(w))e^{-t(x)M}, \quad w = \Phi^a x, \quad a \in \mathbb{Z}, \quad x \in I.$$

By the inductive hypothesis, we get

$$\sigma^{\rm ss}|_{\rm W} = (\alpha|_{\rm W}) \oplus (\beta^{\rm ss}|_{\rm W}) = (\alpha|_{\rm W}) \oplus \rho_1^{\rm ss} = \rho^{\rm ss}.$$

This proves the lemma.

**Corollary 7.4.** — Let  $\sigma$  be a (finite-dimensional) continuous  $\ell$ -adic representation of  $\Gamma$ , with associated Deligne representation  $(\rho, N)$ . Then  $\rho$  is (smooth) integral and  $\mathbf{r}_{\ell}(\rho) = \mathbf{r}_{\ell}(\sigma)|_{W}$ .

*Proof.* — Since  $\sigma$  is integral (for  $\Gamma$  is compact),  $\sigma^{ss}|_W$  is integral. We deduce from Lemma 7.3 that  $\rho^{ss}$  is integral, then from Lemma 7.2 that  $\rho$  is integral. Now write

$$\mathbf{r}_{\ell}(\rho) = \mathbf{r}_{\ell}(\rho^{\mathrm{ss}}) = \mathbf{r}_{\ell}(\sigma^{\mathrm{ss}}|_{\mathrm{W}}) = \mathbf{r}_{\ell}(\sigma^{\mathrm{ss}})|_{\mathrm{W}} = \mathbf{r}_{\ell}(\sigma)|_{\mathrm{W}}$$

This proves the corollary.

#### 8. Galois representations associated with automorphic representations

Recall that we have fixed an isomorphism of fields  $\iota : \mathbb{C} \to \overline{\mathbb{Q}}_{\ell}$ . Fix a positive integer N.

Let k be a totally real number field, and l be either k or a quadratic totally imaginary extension of k in an algebraic closure  $\overline{\mathbb{Q}}$  of  $\mathbb{Q}$ . For any place v of l, let  $l_v$  denote the completion of l at v.

For any finite place v, fix a decomposition subgroup  $\Gamma_v$  of  $\operatorname{Gal}(\overline{\mathbb{Q}}/l)$  at v and write  $W_v$  for the associated Weil group. For any finite place v not dividing  $\ell$ , write

- WD( $\sigma$ ) for the Deligne representation of W<sub>v</sub> associated with a continuous  $\ell$ -adic representation  $\sigma$  of  $\Gamma_v$  and WD<sup>\*</sup>( $\sigma$ ) for its Frobenius-semi-simplification,

- rec<sub>v</sub> for the local Langlands correspondence ([23] Theorem A) between irreducible smooth complex representations of  $\operatorname{GL}_N(l_v)$  and N-dimensional Frobenius-semi-simple complex Deligne representations of  $W_v$ .

#### 8.1.

A cuspidal irreducible automorphic representation  $\Pi$  of  $\operatorname{GL}_N(\mathbb{A}_l)$  is said to be

- *polarized* if its contragredient  $\Pi^{\vee}$  is isomorphic to  $\Pi^c$ , where c is the generator of  $\operatorname{Gal}(l/k)$  (thus  $\Pi^c = \Pi$  when l = k),

- algebraic regular if the Harish-Chandra module  $\Pi_{\infty}$  associated with  $\Pi$  has the same infinitesimal character as some irreducible algebraic representation of the restriction of scalars from lto  $\mathbb{Q}$  of  $\operatorname{GL}_N$ .

Recall the following result of Barnet-Lamb–Geraghty–Harris–Taylor ([5] Theorems 1.1, 1.2).

**Theorem 8.1.** — Let  $\Pi$  be an algebraic regular, polarized, cuspidal irreducible automorphic representation of  $\operatorname{GL}_N(\mathbb{A}_l)$ . There is a continuous semi-simple  $\ell$ -adic representation

$$\Sigma : \operatorname{Gal}(\overline{\mathbb{Q}}/l) \to \operatorname{GL}_N(\overline{\mathbb{Q}}_\ell)$$

such that, for any finite place v of l not dividing l, we have

$$\mathrm{WD}^*(\Sigma|_{\Gamma_v}) \simeq \mathrm{rec}_v(\Pi_v \otimes |\det|_v^{(1-N)/2}) \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_\ell.$$

Note that the representation  $\Sigma$  depends on the choice of  $\iota$ .

#### 8.2.

The main result of this section is the following. Let  $\mathfrak{m}$  denote the maximal ideal of  $\overline{\mathbb{Z}}_{\ell}$ .

**Theorem 8.2.** — Let  $\Pi_1$  and  $\Pi_2$  be algebraic regular, polarized, cuspidal irreducible automorphic representations of  $\operatorname{GL}_n(\mathbb{A}_l)$ . Suppose that there is a finite set S of places of l, containing all infinite places, such that for all  $v \notin S$ :

(1) the local components  $\Pi_{1,v}$  and  $\Pi_{2,v}$  are unramified,

(2) the characteristic polynomials of the conjugacy classes of semisimple elements in  $\operatorname{GL}_n(\overline{\mathbb{Q}}_\ell)$ associated with  $\Pi_{1,v} \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_\ell$  and  $\Pi_{2,v} \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_\ell$  have coefficients in  $\overline{\mathbb{Z}}_\ell$  and are congruent mod  $\mathfrak{m}$ .

Then, for any finite place v of l not dividing  $\ell$ , the representations  $\Pi_{1,v} \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_{\ell}$  and  $\Pi_{2,v} \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_{\ell}$  are integral, their reductions mod  $\mathfrak{m}$  share a common generic irreducible component, and such a generic component is unique.

*Proof.* — Applying Theorem 8.1 to  $\Pi_1$  and  $\Pi_2$ , we get continuous  $\ell$ -adic representations

$$\Sigma_i : \operatorname{Gal}(\overline{\mathbb{Q}}/l) \to \operatorname{GL}_N(\overline{\mathbb{Q}}_\ell), \quad i = 1, 2,$$

such that, for any finite place v of l not dividing  $\ell$ , we have

$$\mathrm{WD}^*(\Sigma_{i,v}) \simeq \mathrm{rec}_v(\Pi_{i,v} \otimes |\det|_v^{(1-N)/2}) \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_\ell$$

where  $\Sigma_{i,v}$  denotes the restriction of  $\Sigma_i$  to  $\Gamma_v$  and the tensor product over  $\mathbb{C}$  is taken with respect to  $\iota$ . For all  $v \notin S$ , the  $\ell$ -adic representation  $\prod_{i,v} \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_{\ell}$  is unramified, generic and integral, thus

$$\operatorname{rec}_{v}(\Pi_{i,v} \otimes |\det|_{v}^{(1-N)/2}) \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_{\ell} \simeq (\phi_{i,v}, 0)$$

where  $\phi_{i,v}$  is an integral semi-simple  $\ell$ -adic representation of  $W_v$  trivial on  $I_v$ . It is thus entirely determined by the semi-simple matrix

$$\phi_{i,v}(\Phi_v) \in \mathrm{GL}_N(\overline{\mathbb{Q}}_\ell)$$

where  $\Phi_v$  is a Frobenius element in  $W_v$ . By assumption, its characteristic polynomial has coefficients in  $\overline{\mathbb{Z}}_{\ell}$ , hence, as  $\phi_{i,v}(\Phi_v)$  is semi-simple, its eigenvalues are in  $\overline{\mathbb{Z}}_{\ell}^{\times}$ . That the nilpotent operator is 0 implies that  $\Sigma_{i,v}$  is smooth, thus

$$\phi_{i,v} = (\Sigma_{i,v}|_{\mathbf{W}_v})^{\mathrm{ss}}.$$

Thus  $\Sigma_{i,v}$  is trivial on the inertia subgroup  $I_v$ , that is,  $\Sigma_{i,v}$  is unramified.

Given  $v \notin S$  and  $i \in \{1, 2\}$ , let  $P_{i,v}(T)$  be the characteristic polynomial of  $\Sigma_{i,v}(\Phi_v) \otimes \overline{\mathbb{F}}_{\ell}$ , that is, the characteristic polynomial of  $\phi_{i,v}(\Phi_v) \otimes \overline{\mathbb{F}}_{\ell}$ . By assumption,  $P_{1,v}(T) = P_{2,v}(T)$  at all  $v \notin S$ . Applying Deligne-Serre [19] Lemma 3.2 to the semi-simple  $\overline{\mathbb{F}}_{\ell}$ -representations  $(\Sigma_1 \otimes \overline{\mathbb{F}}_{\ell})^{ss}$  and  $(\Sigma_2 \otimes \overline{\mathbb{F}}_{\ell})^{\mathrm{ss}}$ , which at  $v \notin S$  give  $\phi_{1,v} \otimes \overline{\mathbb{F}}_{\ell}$  and  $\phi_{2,v} \otimes \overline{\mathbb{F}}_{\ell}$  respectively, we deduce that  $(\Sigma_1 \otimes \overline{\mathbb{F}}_{\ell})^{\mathrm{ss}}$ and  $(\Sigma_2 \otimes \overline{\mathbb{F}}_{\ell})^{ss}$  are isomorphic. In particular, we deduce that

$$(\Sigma_1 \otimes \overline{\mathbb{F}}_\ell)^{\mathrm{ss}}|_{\Gamma_w} \simeq (\Sigma_2 \otimes \overline{\mathbb{F}}_\ell)^{\mathrm{ss}}|_{\Gamma_v}$$

thus the continuous  $\ell$ -adic representations  $\Sigma_{1,w}$  and  $\Sigma_{2,w}$  of  $\Gamma_w$  are congruent mod  $\ell$ .

Now write  $WD^*(\Sigma_{i,w}) = (\rho_i, N_i)$  for i = 1, 2. Thanks to Corollary 7.4, we know that  $\rho_1$  and  $\rho_2$  are integral and have same reduction mod  $\ell$ . By [66] Theorem 1.6, we deduce that

$$\mu_1 = (\Pi_{1,w} \otimes |\det|_w^{(1-N)/2}) \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_\ell, \quad \mu_2 = (\Pi_{2,w} \otimes |\det|_w^{(1-N)/2}) \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_\ell,$$

are integral and have the same mod  $\ell$  supercuspidal support, that is, the supercuspidal support of any irreducible component v of  $\mathbf{r}_{\ell}(\mu_i)$  is independent of i (and of the choice of v).

Since  $\mu_i$  is generic (as  $\Pi_{i,w}$  is a local component of a cuspidal automorphic representation of  $\operatorname{GL}_n(\mathbb{A}_l)$ , the  $\overline{\mathbb{F}}_{\ell}$ -representation  $\mathbf{r}_{\ell}(\mu_i)$  contains a generic irreducible component  $\delta_i$ . It occurs in  $\mathbf{r}_{\ell}(\mu_i)$  with multiplicity 1, and any generic irreducible representation occurring in  $\mathbf{r}_{\ell}(\mu_i)$  is isomorphic to  $\delta_i$ . Since  $\delta_i$  only depends on the mod  $\ell$  supercuspidal support of  $\mu_i$  ([66] III.5.10), we deduce that  $\delta_1$  and  $\delta_2$  are isomorphic. 

**Remark 8.3.** — We expect Theorem 8.2 to hold without assuming that  $\Pi_1$ ,  $\Pi_2$  are polarized.

#### 9. Proof of the main theorem

#### 9.1.

We prove our main theorem 1.1.

Let p be a prime number different from 2, let F be a p-adic field and G be a quasi-split special orthogonal, unitary or symplectic group over F. We thus have

- either G = SO(Q) for some non-degenerate quadratic form Q over F,
- or G = U(H) for some non-degenerate E/F-Hermitian form H,
- or G = Sp(A) for some non-degenerate symplectic form A over F.

As usual, we write E = F in the symplectic and orthogonal cases.

In this paragraph and the next one, we assume that the group G is not the split special orthogonal group  $SO_2(F) \simeq F^{\times}$ . The case of split  $SO_2(F)$  will be treated in Paragraph 9.3.

Let  $\pi_1, \pi_2$  be integral cuspidal irreducible  $\overline{\mathbb{Q}}_{\ell}$ -representations of G such that

$$\mathbf{r}_{\ell}(\pi_1) \leq \mathbf{r}_{\ell}(\pi_2).$$

First, let  $k, w, \mathbf{G}$  be as in Theorem 2.1. More precisely, we have

- either  $\mathbf{G} = \mathrm{SO}(q)$  for a quadratic form q as in Theorem 2.8 if G is special orthogonal,
- or  $\mathbf{G} = \mathrm{U}(h)$  for an l/k-Hermitian form h as in Theorem 2.11 if G is unitary,
- or **G** is as in Paragraph 2.8 if G is symplectic (see also Paragraph 5.6).

In particular, we have  $k_w = F$  and  $l_w = E$ , and the group  $\mathbf{G}(F)$  naturally identifies with G. As usual, we write l = k in the symplectic and orthogonal cases.

Let  $\mathbf{G}^*$  be the quasi-split inner form of  $\mathbf{G}$  over k, and write  $N = N(\mathbf{G}^*)$ . We thus have:

- either  $\mathbf{G}^* = \mathrm{SO}(q^*)$  where  $q^*$  is a quadratic form over k as in (5.2) or (5.3),
- or  $\mathbf{G}^* = \mathrm{U}(h^*)$  where  $h^*$  is an l/k-Hermitian form as in (5.4),
- or  $\mathbf{G}^* = \operatorname{Sp}(f^*)$  where  $f^*$  is a symplectic form over k as in (5.1).

Let **t** be the local transfer from  $\mathbf{G}^*(F)$  to  $\operatorname{GL}_N(E)$  given by Definition 5.1. We explained how to canonically identify representations of  $\mathbf{G}(F)$  with those of  $\mathbf{G}^*(F)$  in Paragraph 5.6. (In the symplectic case, we identified  $\mathbf{G}(F)$  with  $\operatorname{Sp}(f_w)$  for some symplectic form  $f_w$  over  $k_w = F$ .) This gives us a local transfer from  $\mathbf{G}(F) = G$  to  $\operatorname{GL}_N(E)$ , still denoted **t**.

**Lemma 9.1.** — There is a finite place u of k different from w, not dividing  $\ell$ , such that there is a unitary cuspidal irreducible complex representation  $\rho$  of  $\mathbf{G}(k_u)$  with the following properties:

- (1)  $\rho$  is compactly induced from some compact mod centre, open subgroup of  $\mathbf{G}(k_u)$ ,
- (2) the local transfer of  $\rho$  to  $\operatorname{GL}_N(l_u)$  is cuspidal.

*Proof.* — If G is special orthogonal, it suffices to choose  $u \neq w$  such that  $\mathbf{G}(k_u)$  is split (Remark 2.9), and then apply Proposition B.1.

If G is unitary, it suffices to choose  $u \neq w$  such that  $\mathbf{G}(k_u)$  is split (Remark 2.12).

If G is symplectic, it suffices to choose a place u such that  $k_u$  is isomorphic to  $\mathbb{Q}_2$  (see Lemma 2.5) and then apply Theorem C.1.

Now fix an isomorphism of fields  $\iota : \mathbb{C} \to \overline{\mathbb{Q}}_{\ell}$ . As in Paragraph 1.2, let  $\mathbf{t}_{\ell}$  denote the  $\ell$ -adic local transfer from G to  $\operatorname{GL}_N(E)$  obtained from  $\mathbf{t}$  thanks to  $\iota$ , that is,  $\mathbf{t}_{\ell}(V \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_{\ell}) = \mathbf{t}(V) \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_{\ell}$ 

 $\mathbf{42}$ 

for any complex representation V of G, where tensor products are taken with respect to  $\iota$ . Let u and  $\rho$  be as in Lemma 9.1. By Theorem 4.4, there are irreducible automorphic representations  $\Pi_1$  and  $\Pi_2$  of  $\mathbf{G}(\mathbb{A})$  such that

- (1)  $\Pi_{1,u}$  and  $\Pi_{2,u}$  are both isomorphic to  $\rho$ ,
- (2)  $\Pi_{1,w} \otimes_{\mathbb{C}} \mathbb{Q}_{\ell}$  is isomorphic to  $\pi_1$  and  $\Pi_{1,w} \otimes_{\mathbb{C}} \mathbb{Q}_{\ell}$  is isomorphic to  $\pi_2$ ,
- (3)  $\Pi_{1,v}$  and  $\Pi_{2,v}$  are trivial for any real place v,
- (4) there is a finite set S of places of k, containing all real places, such that for all  $v \notin S$ :
  - (a) the group **G** is unramified over  $k_v$ , and the local components  $\Pi_{1,v}$  and  $\Pi_{2,v}$  are unramified with respect to some hyperspecial maximal compact subgroup  $K_v$  of  $\mathbf{G}(k_v)$ ,
  - (b) the restrictions of the Satake parameters of  $\Pi_{1,v} \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_{\ell}$  and  $\Pi_{2,v} \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_{\ell}$  to the Hecke  $\overline{\mathbb{Z}}_{\ell}$ -algebra  $\mathcal{H}_{\overline{\mathbb{Z}}_{\ell}}(\mathbf{G}(k_v), \mathbf{K}_v)$  are congruent mod the maximal ideal  $\mathfrak{m}$  of  $\overline{\mathbb{Z}}_{\ell}$ .

Applying Theorems 5.5 and 5.6 to  $\Pi_1$ ,  $\Pi_2$ , we get algebraic regular, polarized, cuspidal irreducible automorphic representations  $\widetilde{\Pi}_1$ ,  $\widetilde{\Pi}_2$  of  $\operatorname{GL}_N(l)$  such that, for i = 1, 2 and all finite places v of k, the local transfer of  $\Pi_{i,v}$  to  $\operatorname{GL}_N(l_v)$  is  $\widetilde{\Pi}_{i,v}$ . Writing  $\mathbf{t}_v$  for the local transfer over  $k_v$ , we thus have  $\widetilde{\Pi}_{i,v} = \mathbf{t}_v(\Pi_{i,v})$ , or equivalently  $\widetilde{\Pi}_{i,v} \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_{\ell} = \mathbf{t}_{v,\ell}(\Pi_{i,v} \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_{\ell})$  where  $\mathbf{t}_{v,\ell}$  is obtained from  $\mathbf{t}_v$ thanks to  $\iota$ .

In particular, for all  $v \notin S$ , it follows from Proposition 6.3 that  $\Pi_{1,v}$  and  $\Pi_{2,v}$  are unramified and that the characteristic polynomials of the conjugacy classes of semisimple elements in  $\operatorname{GL}_n(\overline{\mathbb{Q}}_\ell)$ associated with  $\Pi_{1,v} \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_\ell$  and  $\Pi_{2,v} \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_\ell$  have coefficients in  $\overline{\mathbb{Z}}_\ell$  and are congruent mod  $\mathfrak{m}$ .

Now apply Theorem 8.2 at w: the representations  $\Pi_{1,w} \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_{\ell}$  and  $\Pi_{2,w} \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_{\ell}$  are integral, their reductions mod  $\ell$  share a common generic irreducible component, and such a generic component is unique. The result now follows from the fact that  $\Pi_{i,w} \otimes_{\mathbb{C}} \overline{\mathbb{Q}}_{\ell} \simeq \pi_i$  for i = 1, 2.

#### 9.2.

We now describe how the map  $\mathbf{t}_{\ell}$  depends on the choice of  $\iota$ . Equivalently, since any two isomorphisms  $\iota, \iota'$  between  $\mathbb{C}$  and  $\overline{\mathbb{Q}}_{\ell}$  give rise to a field automorphism  $\iota^{-1} \circ \iota'$  of  $\mathbb{C}$ , we will describe the behavior of  $\mathbf{t}$  under the action of Aut( $\mathbb{C}$ ). More precisely, we prove the following result.

**Proposition 9.2.** — Let  $\pi$  be a cuspidal complex representation of G. Let  $\gamma \in Aut(\mathbb{C})$ .

- (1) If G is not even orthogonal, then  $\mathbf{t}(\pi^{\gamma}) = \mathbf{t}(\pi)^{\gamma}$ .
- (2) If G is even orthogonal, then  $\mathbf{t}(\pi^{\gamma}) = \mathbf{t}(\pi)^{\gamma} \cdot \varepsilon_{\gamma}$ , where  $\varepsilon_{\gamma}$  is the unramified character

(9.1) 
$$x \mapsto \left(\frac{\gamma(\sqrt{q})}{\sqrt{q}}\right)^{\operatorname{val}_F(x)}$$

of  $F^{\times}$ , where q is the cardinality of the residue field of F.

Let  $\pi$  be a cuspidal complex representation of G. As in Lemma 9.1, let u be a finite place of k different from w, not dividing  $\ell$ , and  $\rho$  be a unitary cuspidal irreducible complex representation of  $\mathbf{G}(k_u)$  with cuspidal transfer. By Proposition 4.1, we have an irreducible automorphic representation  $\Pi$  of  $\mathbf{G}(\mathbb{A})$  such that

- (1) the local component  $\Pi_u$  is isomorphic to  $\rho$ ,
- (2) the local component  $\Pi_w$  is isomorphic to  $\pi$ ,

(3) the local component  $\Pi_v$  is the trivial character of  $\mathbf{G}(k_v)$  for any real place v of k.

Associated with  $\Pi$  by Theorems 5.5 and 5.6, there is an algebraic regular, polarized, cuspidal irreducible automorphic representation  $\widetilde{\Pi}$  of  $\operatorname{GL}_N(\mathbb{A}_l)$  such that  $\widetilde{\Pi}_v = \mathbf{t}_v(\Pi_v)$  for all finite places v of k, where  $\mathbf{t}_v$  is as in Paragraph 9.1. Now let  $\gamma \in \operatorname{Aut}(\mathbb{C})$ . Then  $\Pi^{\gamma}$  satisfies

- (1) the local component  $\Pi_u^{\gamma}$  is isomorphic to  $\rho^{\gamma}$ ,
- (2) the local component  $\Pi_w^{\gamma}$  is isomorphic to  $\pi^{\gamma}$ ,
- (3) the local component  $\Pi_v^{\gamma}$  is the trivial character of  $\mathbf{G}(k_v)$  for any real place v of k.

Associated with it by Theorems 5.5 and 5.6, there is an algebraic regular, polarized, cuspidal irreducible automorphic representation  $\widetilde{\Pi}'$  of  $\operatorname{GL}_N(\mathbb{A}_l)$  such that  $\widetilde{\Pi}'_v = \mathbf{t}_v(\Pi_v^{\gamma})$  for all finite v.

Let S be a finite set of places of k, containing all real places, such that for all  $v \notin S$  the group **G** is unramified over  $k_v$  and the local component  $\Pi_v$  is unramified with respect to some hyperspecial maximal compact subgroup of  $\mathbf{G}(k_v)$ .

Assume first that G is not an even special orthogonal group. For  $v \notin S$ , Proposition 6.7 gives us  $\mathbf{t}_v(\Pi_v^{\gamma}) = \mathbf{t}_v(\Pi_v)^{\gamma}$ , thus  $\widetilde{\Pi}'$  and  $\widetilde{\Pi}^{\gamma}$  coincide at almost all finite places. By strong multiplicity 1, we deduce that  $\widetilde{\Pi}' = \widetilde{\Pi}^{\gamma}$ . It follows that  $\mathbf{t}_w(\Pi_w^{\gamma}) = \mathbf{t}_w(\Pi_w)^{\gamma}$ , that is,  $\mathbf{t}(\pi^{\gamma}) = \mathbf{t}(\pi)^{\gamma}$ .

Assume now that G is even special orthogonal (thus l = k). For all finite places v of k, let  $\mathbf{t}_v^*$  be the map  $\pi \mapsto \mathbf{t}_v(\pi) |\det|_v^{1/2}$ , where  $|\cdot|_v$  is the absolute value of  $k_v^{\times}$  and  $|\cdot|_v^{1/2}$  is its square root with respect to  $q_v^{1/2}$ , where  $q_v$  is the cardinality of the residue field of  $k_v$ . An argument similar to that of the non even orthogonal case gives us  $\widetilde{\Pi}' |\det|^{1/2} = (\widetilde{\Pi} |\det|^{1/2})^{\gamma}$  where  $|\cdot|$  is the absolute value of  $\mathbb{A}^{\times}$ . Looking at the local component at w, we deduce that  $\mathbf{t}(\pi^{\gamma}) = \mathbf{t}(\pi)^{\gamma} \cdot \varepsilon_{\gamma}$ , where  $\varepsilon_{\gamma}$  is defined as in (9.1). We have proved Proposition 9.2.

**Remark 9.3.** — The same argument shows that Proposition 9.2 holds for all discrete series representations  $\pi$  of G (it suffices to replace Proposition 4.1 by [58] Theorem 5.13). Let us explain how this implies that the set of isomorphism classes of discrete series representations of G is stable under Aut( $\mathbb{C}$ ). Let  $\varphi$  be the local Langlands parameter of a discrete series representation  $\pi$  (up to  $O_{2n}(\mathbb{C})$ -conjugacy in the even orthogonal case) and let  $\phi = \text{Std} \circ \varphi$  be the Langlands parameter of  $\mathbf{t}(\pi)$ . On the one hand, the fact that  $\pi$  is a discrete series representation implies that the quotient of the centralizer of the image of  $\phi$  in  $\hat{\mathbf{G}}$  by  $\mathbf{Z}(\hat{\mathbf{G}})^{W_k}$  is finite (see the end of Paragraph 5.3). On the other hand, the Langlands parameter of  $\mathbf{t}(\pi^{\gamma})$  is  $\phi' = \phi^{\gamma} \cdot \eta \chi$  (where  $\eta$  is the unramified character of  $F^{\times}$  of order 2 and  $\chi$  is either the character  $\varepsilon_{\gamma}$  defined by (9.1) if G is even orthogonal, or the trivial character otherwise), which has the same finiteness property. Thus the *L*-packet of  $\pi^{\gamma}$  is discrete. Thus  $\pi^{\gamma}$  is a discrete series representation.

**Remark 9.4.** — Let us examine how the local transfer map behaves under automorphisms of the base field F, for discrete series representations. Let  $\pi$  be a discrete series representation of G, and let  $\phi$  be the Langlands parameter of its transfer  $\mathbf{t}(\pi)$ . By Moeglin [45], an irreducible Langlands parameter  $\sigma \boxtimes S_a$ , where  $\sigma$  is an irreducible representation of dimension  $k \ge 1$  of  $W_F$ and a is a positive integer, occurs in  $\phi$  if and only if:

(1) the cuspidal representation  $\rho$  of  $\operatorname{GL}_k(E)$  associated with  $\sigma$  by the Langlands correspondence is *c*-selfdual,

44

(2) if s is the unique non-negative real number such that the normalized parabolically induced representation  $\rho\nu^s \rtimes \pi$  is reducible, then 2s - 1 is a positive integer and 2s - 1 - a is a non-negative even integer.

Now let  $\varkappa \in \operatorname{Aut}(F)$ , which extends to an automorphism of E still denoted  $\varkappa$ . Then

- the cuspidal representation  $\rho^{\varkappa}$  is  $\varkappa^{-1}c\varkappa$ -selfdual,

– the irreducible representation of  $W_E$  associated with it by the Langlands correspondence is  $\sigma^{\varkappa}$  (see [26] Propriété 1),

- the normalized parabolically induced representation  $\rho^{\varkappa}\nu^{s} \rtimes \pi^{\varkappa}$  is reducible,

- the representation  $\sigma \boxtimes S_a$  occurs in  $\phi$  if and only if  $\sigma^{\varkappa} \boxtimes S_a$  occurs in  $\phi^{\varkappa}$ .

It follows that the Langlands parameter of  $\mathbf{t}(\pi^{\varkappa})$  is  $\phi^{\varkappa}$ . Applying [26] Propriété 1 again,  $\phi^{\varkappa}$  is the Langlands parameter of  $\mathbf{t}(\pi)^{\varkappa}$ . Thus  $\mathbf{t}(\pi^{\varkappa})$  is equal to  $\mathbf{t}(\pi)^{\varkappa}$ .

#### 9.3.

In this paragraph, we discuss the case of the split special orthogonal group  $SO_2(F) \simeq F^{\times}$ . Let  $\chi$  be a  $\overline{\mathbb{Q}}_{\ell}$ -character of this group. Its transfer to  $GL_2(F)$  is

– either the normalized parabolically induced representation  $\chi \times \chi^{-1}$  when the character  $\chi^2$  is different from the absolute value  $|\cdot|$  and its inverse  $|\cdot|^{-1}$ ,

- or the unique character occurring as a component of  $\chi \times \chi^{-1}$  when  $\chi^2 \in \{|\cdot|, |\cdot|^{-1}\}$ .

Properties (1) and (2) of Theorem 1.1 thus hold, since

– an irreducible  $\overline{\mathbb{Q}}_{\ell}$ -representation of  $\operatorname{GL}_2(F)$  is integral if and only if its cuspidal support is integral (see [65] II.4.14 and [15] Proposition 6.7),

- if  $\chi$  is integral, the supercuspidal support of any irreducible component of  $\mathbf{r}_{\ell}(\chi \times \chi^{-1})$  is the  $\operatorname{GL}_2(F)$ -conjugacy class of the cuspidal pair  $(F^{\times} \times F^{\times}, \chi \otimes \chi^{-1})$ .

However, if  $\xi$  is any non-trivial character of  $F^{\times}$  with values in  $1 + \mathfrak{m}$  (where  $\mathfrak{m}$  is the maximal ideal of  $\overline{\mathbb{Z}}_{\ell}$ ) such that  $\xi^2 \notin \{1, |\cdot|^{-2}\}$ , the characters  $|\cdot|^{1/2}$  and  $\xi |\cdot|^{1/2}$  are congruent, but the transfer of the first one is the trivial character of  $\operatorname{GL}_2(F)$ , which is not generic. Property (3) thus does not hold. Also, the transfer of the second one is  $\xi |\cdot|^{1/2} \times \xi^{-1} |\cdot|^{-1/2}$ , whose reduction mod  $\ell$  contains the trivial character with multiplicity 1 (if  $\ell \neq 2$ ) or 2 (if  $\ell = 2$ ) by [**64**] Théorème 3.

Assume further that q has order 2 mod  $\ell$ , that is,  $\ell$  divides  $q^2 - 1$  but not q - 1, and let  $\eta$  be the unique unramified  $\overline{\mathbb{Q}}_{\ell}$ -character of order 2 of  $F^{\times}$ . Then the transfer of  $\eta |\cdot|^{-1/2}$  (which is congruent to  $|\cdot|^{1/2}$ ) is  $\eta \circ \det$ , whose reduction mod  $\ell$  is a character of order 2. We thus have two congruent characters of  $F^{\times}$  whose transfers to  $\operatorname{GL}_2(F)$  have reductions mod  $\ell$  with no component in common.

#### $\mathbf{A}$

#### Cyclic base change

Let F be a p-adic field, and let K be a cyclic finite extension of F of degree d. Fix an integer  $n \ge 1$  and write  $G = \operatorname{GL}_n(F)$  and  $H = \operatorname{GL}_n(K)$ . By [3], there exists a map from isomorphism

classes of irreducible (smooth) complex representations of G to those of H called the local *base* change, denoted  $\mathbf{b} = \mathbf{b}_{K/F}$ .

Now let us fix a prime number  $\ell$  different from p and an isomorphism of fields  $\iota$  between  $\mathbb{C}$  and  $\overline{\mathbb{Q}}_{\ell}$ . Replacing  $\mathbb{C}$  by  $\overline{\mathbb{Q}}_{\ell}$  thanks to  $\iota$ , one obtains a local base change  $\mathbf{b}_{K/F,\ell}$  for irreducible smooth  $\overline{\mathbb{Q}}_{\ell}$ -representations.

In this appendix, we investigate the dependency of  $\mathbf{b}_{K/F,\ell}$  in the choice of  $\iota$ , or equivalently the behavior of  $\mathbf{b}_{K/F}$  with respect to automorphisms of  $\mathbb{C}$ .

#### A.1.

Let  $\mathbf{a}_F$  denote the local Langlands correspondence from the set of isomorphism classes of irreducible complex representations of G to the set  $\Phi(G)$  of  $\operatorname{GL}_n(\mathbb{C})$ -conjugacy classes of local Langlands parameters for G ([23, 25]).

Replacing  $\mathbb{C}$  by  $\overline{\mathbb{Q}}_{\ell}$  thanks to  $\iota$ , one obtains a local Langlands correspondence  $\mathbf{a}_{F,\ell}$  for irreducible  $\overline{\mathbb{Q}}_{\ell}$ -representations. The dependency of  $\mathbf{a}_{F,\ell}$  in  $\iota$ , or equivalently the behavior of  $\mathbf{a}_F$  with respect to automorphisms of  $\mathbb{C}$ , has been studied in [26, 14]: the map  $\pi \mapsto \mathbf{a}_F(\pi |\det|^{(1-n)/2})$  is insensitive to automorphisms of  $\mathbb{C}$ . It follows that

(A.1) 
$$\mathbf{a}_F(\pi^{\gamma}) = \mathbf{a}_F(\pi)^{\gamma} \cdot \eta_{F\gamma}^{1-n}$$

for all  $\gamma \in \operatorname{Aut}(\mathbb{C})$  and all irreducible complex representations  $\pi$  of G, where

(A.2) 
$$\eta_{F,\gamma}(w) = \left(\frac{\gamma(\sqrt{q})}{\sqrt{q}}\right)^{v_F(w)}$$

for all  $w \in W_F$ , where  $v_F$  is the valuation map taking any Frobenius element to 1.

#### A.2.

Let  $\operatorname{res}_{K/F}$  be the map from  $\Phi(G)$  to  $\Phi(H)$  defined by restricting local Langlands parameters from WD<sub>F</sub> to WD<sub>K</sub>. The local base change  $\mathbf{b}_{K/F}$  is characterized by the identity

$$\mathsf{a}_K \circ \mathsf{b}_{K/F} = \mathsf{res}_{K/F} \circ \mathsf{a}_F.$$

Now let us prove that  $\mathbf{b} = \mathbf{b}_{K/F}$  is insensitive to the action of  $\operatorname{Aut}(\mathbb{C})$ .

**Proposition A.1.** — For all  $\gamma \in Aut(\mathbb{C})$  and all irreducible complex representations  $\pi$  of G, we have  $\mathbf{b}_{K/F}(\pi^{\gamma}) = \mathbf{b}_{K/F}(\pi)^{\gamma}$ .

*Proof.* — Let  $\pi$  be an irreducible complex representation of G. We have

$$\begin{aligned} \mathbf{a}_{K}(\mathbf{b}_{K/F}(\pi^{\gamma})) &= \operatorname{res}_{K/F}(\mathbf{a}_{F}(\pi^{\gamma})) \\ &= \operatorname{res}_{K/F}(\mathbf{a}_{F}(\pi)^{\gamma} \cdot \eta_{F,\gamma}^{1-n}) \\ &= \mathbf{a}_{K}(\mathbf{b}_{K/F}(\pi))^{\gamma} \cdot (\eta_{F,\gamma}|_{W_{K}})^{1-n} \\ &= \mathbf{a}_{K}(\mathbf{b}_{K/F}(\pi)^{\gamma}) \cdot (\eta_{K,\gamma} \cdot \eta_{F,\gamma}|_{W_{K}})^{1-n} \,. \end{aligned}$$

We are thus reduced to compare  $\eta_{F,\gamma}|_{W_K}$  with  $\eta_{K,\gamma}$ . Using the explicit formula (A.2), we get

$$\eta_{F,\gamma}|_{W_K} = \left(\frac{\gamma(\sqrt{q})}{\sqrt{q}}\right)^{\upsilon_F|_{W_K}}, \quad \eta_{K,\gamma} = \left(\frac{\gamma(\sqrt{q'})}{\sqrt{q'}}\right)^{\upsilon_K}$$

where q' is the cardinality of the residue field of K. Since  $q' = q^{f_{K/F}}$  and  $\upsilon_F|_{W_K} = f_{K/F}\upsilon_K$ , we deduce that  $\eta_{F,\gamma}|_{W_K} = \eta_{K,\gamma}$ , thus  $\mathbf{b}_{K/F}(\pi^{\gamma}) = \mathbf{b}_{K/F}(\pi)^{\gamma}$ .

#### A.3.

The map  $\mathbf{b}_{K/F,\ell}$  preserves the fact of being integral: this follows from the fact that an irreducible  $\overline{\mathbb{Q}}_{\ell}$ -representation  $\pi$  of G is integral if and only if the restriction of  $\mathbf{a}_{F,\ell}(\pi)$  to  $W_F$  is integral ([66] 1.4) and that the restriction to  $W_K$  of an integral  $\overline{\mathbb{Q}}_{\ell}$ -representation of  $W_F$  is integral.

#### A.4.

We now review the congruence properties of  $\mathbf{b}_{K/F,\ell}$ , after J. Zou's PhD thesis [71] 1.10.

Associated with an irreducible representation  $\tau$  of  $\operatorname{GL}_n(K)$ , with coefficients in  $\overline{\mathbb{Q}}_{\ell}$  or  $\overline{\mathbb{F}}_{\ell}$ , there is a partition

$$\lambda(\tau) = (k_1 \ge k_2 \ge \dots)$$

of *n* defined inductively as follows. Let  $k_1$  denote the largest integer  $k \in \{1, \ldots, n\}$  such that the *k*th derivative  $\tau^{(k)}$  is non-zero. If  $k_1 = n$ , then  $\lambda(\tau) = (n)$ . Otherwise,  $(k_2 \ge \ldots)$  is the partition of  $n - k_1$  associated with the representation  $\tau^{(k_1)}$  of  $\operatorname{GL}_{n-k_1}(K)$ .

By [66] V.9.2, if  $\tau$  is an integral irreducible  $\overline{\mathbb{Q}}_{\ell}$ -representation of  $\operatorname{GL}_n(K)$ , its reduction mod  $\ell$ has a unique irreducible component  $\pi$  such that  $\lambda(\pi) = \lambda(\tau)$ . This component is denoted  $\mathbf{j}_{\ell}(\tau)$ .

**Theorem A.2** ([71] Theorem 1.10.17). — Let  $\pi_1$  and  $\pi_2$  be integral irreducible  $\overline{\mathbb{Q}}_{\ell}$ -representations of  $\operatorname{GL}_n(F)$ . If  $\mathbf{j}_{\ell}(\pi_1) = \mathbf{j}_{\ell}(\pi_2)$ , then  $\mathbf{j}_{\ell}(\mathbf{b}_{K/F,\ell}(\pi_1)) = \mathbf{j}_{\ell}(\mathbf{b}_{K/F,\ell}(\pi_2))$ .

In particular, if  $\pi_1$ ,  $\pi_2$  are cuspidal, which implies that  $\lambda(\pi_1) = \lambda(\pi_2) = (n)$ , their base changes  $\mathbf{b}_{K/F,\ell}(\pi_1)$  and  $\mathbf{b}_{K/F,\ell}(\pi_2)$  are generic. This theorem thus says that, if  $\mathbf{r}_{\ell}(\pi_1) = \mathbf{r}_{\ell}(\pi_2)$ , then  $\mathbf{r}_{\ell}(\mathbf{b}_{K/F,\ell}(\pi_1))$  and  $\mathbf{r}_{\ell}(\mathbf{b}_{K/F,\ell}(\pi_2))$  have a unique generic irreducible component in common. This can be seen as an analogue of Theorem 1.1 for the cyclic base change from G to H.

#### A.5.

In this paragraph, we give an example of congruent integral cuspidal  $\overline{\mathbb{Q}}_{\ell}$ -representations  $\pi_1, \pi_2$  of G such that  $\mathbf{b}_{K/F,\ell}(\pi_1)$  and  $\mathbf{b}_{K/F,\ell}(\pi_2)$  are not congruent.

First, assume that  $\pi$  is an integral cuspidal irreducible  $\overline{\mathbb{Q}}_{\ell}$ -representation of G. Let m denote the cardinality of the set of isomorphism classes of  $\pi\chi$ , where  $\chi$  runs over the characters of  $F^{\times}$  trivial on  $N_{K/F}(K^{\times})$ , and set e = d/m. Then there exists a cuspidal irreducible representation  $\rho$  of  $\operatorname{GL}_{n/e}(K)$  such that

$$\mathbf{b}_{K/F}(\pi) = \rho \times \rho^{\alpha} \times \cdots \times \rho^{\alpha^{e-1}}$$

where  $\alpha$  is a generator of  $\operatorname{Gal}(K/F)$  and  $\times$  denotes normalized parabolic induction with respect to a choice of square root of q, the cardinality of the residue field of F (see [3] Chapter 1, §6.4).

Now assume that n = 2 and that K is a ramified quadratic extension of F, and let  $\omega_{K/F}$  be the character of  $F^{\times}$  with kernel  $N_{K/F}(K^{\times})$ . Let  $\pi_1$  be an integral cuspidal  $\overline{\mathbb{Q}}_{\ell}$ -representation of  $G = \operatorname{GL}_2(F)$  of level 0. By [13], it is compactly induced from a representation  $\lambda_1$  of  $F^{\times}\operatorname{GL}_2(\mathcal{O}_F)$ whose restriction to  $\operatorname{GL}_2(\mathcal{O}_F)$  is the inflation of a cuspidal irreducible representation  $\sigma_1$  of the group  $\operatorname{GL}_2(\mathbf{k})$ , where  $\mathbf{k}$  is the residue field of F. Associated with  $\sigma_1$ , there is ([20]) a character

(A.3) 
$$\xi_1: l^{\times} \to \mathbb{Z}_{\ell}^{\times}$$

such that  $\xi_1^q \neq \xi_1$ , where l is a quadratic extension of k and q is the cardinality of k.

The representation  $\pi_1 \omega_{K/F}$  is isomorphic to  $\pi_1$  if and only if  $\lambda_1 \omega_{K/F}$  is isomorphic to  $\lambda_1$ . As these representations all have the same central character, this is equivalent to  $\sigma_1 \eta \simeq \sigma_1$ , where  $\eta$ is the unique character of order 2 of  $\mathbf{k}^{\times}$  (note that the restriction of  $\omega_{K/F}$  to  $\mathcal{O}_F^{\times}$  is the inflation of  $\eta$ ), which is equivalent to  $\xi_1(\eta \circ N_{l/k}) = \xi_1^q$ , that is,  $\xi_1^{q-1}$  has order 2. Assume that this is the case. Thus  $e_1 = 2$  and we may write  $\mathbf{b}_{K/F}(\pi_1) = \rho_1 \times \rho_1^{\alpha}$  for some (tamely ramified, integral) character  $\rho_1$  of  $K^{\times}$ .

Assume further that  $\ell$  is a prime divisor of  $q^2 - 1$  not dividing q - 1, that is,  $\ell$  is an odd prime divisor of q + 1. Let  $\mu$  be a character of  $l^{\times}$  of order  $\ell$  and set  $\xi_2 = \xi_1 \mu$ . Since  $\xi_2^q \neq \xi_2$ , there is a cuspidal  $\overline{\mathbb{Q}}_{\ell}$ -representation  $\sigma_2$  of  $\operatorname{GL}_2(\mathbf{k})$  associated with  $\xi_2$ . Since  $\xi_2$  and  $\xi_1$  are congruent,  $\sigma_2$  and  $\sigma_1$  are congruent (see for instance [40] 2.6). Let us inflate and extend  $\sigma_2$  to a representation  $\lambda_2$  of  $F^{\times}\operatorname{GL}_2(\mathbb{O}_F)$  which is congruent to  $\lambda_1$ , then compactly induce  $\lambda_2$  to a representation  $\pi_2$  of  $\operatorname{GL}_2(F)$ . This is an integral cuspidal representation of level 0 which is congruent to  $\pi_1$ .

Since  $\mu^q \neq \mu$ , we have  $e_2 = 1$ , thus  $\mathbf{b}_{K/F}(\pi_1)$  is a cuspidal representation  $\rho_2$  of  $\operatorname{GL}_2(K)$ . Its reduction mod  $\ell$  is an irreducible cuspidal  $\overline{\mathbb{F}}_{\ell}$ -representation of  $\operatorname{GL}_2(K)$ . It is the unique generic component of  $\mathbf{r}_{\ell}(\rho_1 \times \rho_1^{\alpha})$ .

#### В

# Cuspidal representations of split *p*-adic orthogonal groups with irreducible Galois parameter

#### B.1.

Let F be a p-adic field with  $p \neq 2$ , and let G be a split special orthogonal group over F, that is, G = SO(Q) where Q is a maximally isotropic quadratic form over F. Let n be the dimension of Q. In this section, we assume that  $n \neq 2$ . Let  $m = \lfloor n/2 \rfloor$  be the Witt index of Q. With the notation of Paragraph 5.1, we have  $G = SO_{2m+1}(F)$  if n is odd,  $G = SO_{2m}^1(F)$  if n is even. We will prove the following result.

**Proposition B.1.** — There exists a cuspidal representation of level 0 of G whose transfer to  $GL_N(F)$  is cuspidal.

#### **B.2**.

In this paragraph, we refer to [37] §2 (see p. 1090 in particular). Let V be the n-dimensional F-vector space on which Q is defined. Write

$$V = V^{\rm an} \oplus V^{\rm iso}$$

48

where  $V^{\text{an}}$  is anisotropic (thus dim $(V^{\text{an}}) \leq 1$ ) and  $V^{\text{iso}}$  is a sum of *m* hyperbolic planes.

Let q denote the cardinality of the residue field of F. The anisotropic group  $G^{an} = SO(V^{an})$ has a unique (up to conjugacy) maximal parahoric subgroup. Its finite reductive quotient  $\mathcal{G}^{an}$ has neutral component the finite special orthogonal group

$$SO_a(q)$$

with  $a = \dim(V^{\mathrm{an}})$ .

For any choice of integers  $m_1, m_2 \ge 0$  such that  $m_1 + m_2 = m$ , there is a maximal parahoric subgroup  $J = J_{m_1,m_2}$  whose finite reductive quotient  $\mathfrak{G} = \mathfrak{G}_{m_1,m_2}$  has neutral component

$$SO_{a+m_1,m_1}(q) \times SO_{m_2,m_2}(q)$$

where  $SO_{u,v}(q)$  is the special orthogonal group over  $\mathbb{F}_q$  associated with a quadratic space of dimension u + v and Witt index v. Choose  $m_2 = 0$ , so that  $\mathcal{G}$  has neutral component  $SO_{m,m}(q)$  if n = 2m, and  $SO_{m+1,m}(q)$  if n = 2m + 1. In other words,  $\mathcal{G}^{\circ}$  is split.

#### B.3.

Let  $\sigma$  be a self-dual cuspidal irreducible representation of  $\operatorname{GL}_{2r}(q)$  and  $s \in \mathbb{F}_{q^{2r}}^{\times}$  be a parameter corresponding to  $\sigma$ . In particular, s has degree 2r over  $\mathbb{F}_q$  and  $s^{-1} = s^{q^r}$ . Its characteristic polynomial P(X) is thus irreducible, of degree 2r, and self-dual (that is, reciprocal).

The parameter s can be seen in the dual group  $\mathcal{G}^{\circ,*} \subseteq \operatorname{GL}_{2r}(q)$ . It then defines a Lusztig series  $\mathcal{E}(\mathcal{G}^{\circ}, s)$ .

#### **Lemma B.2.** — The Lusztig series $\mathcal{E}(\mathcal{G}^\circ, s)$ contains a cuspidal representation.

*Proof.* — If *m* is odd, see [**37**] §7.2 (p. 1098). Assume now that *m* is even. We follow [**37**] §7.3. Consider the group with connected centre  $\tilde{\mathcal{G}} = \text{GSO}_m^{\pm}$  of which  $\mathcal{G}^{\circ}$  is a subgroup. The scalars 1 and -1 are not eigenvalues of *s*. The centralizer of *s* is thus connected and the two Lusztig series associated with *s* are the same. A cuspidal representation of  $\mathcal{G}^{\circ}$  associated with *s* is an irreducible component of the restriction to  $\mathcal{G}^{\circ}$  of a cuspidal representation of  $\tilde{\mathcal{G}}$  associated with a semi-simple element  $\tilde{s} \in \tilde{\mathcal{G}}^*$  lifting *s*. To prove the lemma, it thus suffices to prove that the Lusztig series  $\mathcal{E}(\tilde{\mathcal{G}}, \tilde{s})$  contains a cuspidal representation.

The two groups  $\tilde{\mathfrak{G}}$  and  $\mathfrak{G}^{\circ}$  act naturally on the same space, thus  $\tilde{s}$  and s have the same characteristic polynomial P(X). It follows from [**37**] §7.2 (p. 1098) that  $\mathcal{E}(\tilde{\mathfrak{G}}, \tilde{s})$  contains a cuspidal representation.

#### **B.4**.

Let  $\tau$  be a cuspidal representation in the Lusztig series  $\mathcal{E}(\mathcal{G}^{\circ}, s)$ . Let  $\lambda$  be an irreducible representation of J whose restriction to  $J^{\circ}$  (the preimage of  $\mathcal{G}^{\circ}$  in J) is a direct sum of conjugates (under J) of the inflation of  $\tau$ . Let  $\pi$  be the representation obtained by compactly inducing  $\lambda$ to G. It is a cuspidal irreducible representation of level 0 of G.

As G is split, it follows from Moeglin [45] that the Langlands parameter  $\varphi$  associated with  $\pi$  is described by the reducibility set  $\text{Red}(\pi)$  and the Jordan set  $\text{Jord}(\pi)$  (see for instance the introduction of [37] for a definition).

In our situation, it follows from [37] §8 that the sets  $\operatorname{Red}(\pi)$  and  $\operatorname{Jord}(\pi)$  are equal and both reduced to a single element  $(\rho, 1)$ , where  $\rho$  is a selfdual cuspidal representation of  $\operatorname{GL}_N(F)$  (with N = n - 1 if n is odd and N = n if n is even), which proves Proposition B.1.

**Remark B.3.** — More precisely,  $\rho$  has level 0, and is obtained by compactly inducing a representation of  $F^{\times} \mathrm{GL}_N(\mathcal{O}_F)$  which is trivial on  $1 + \mathbf{M}_N(\mathfrak{p}_F)$  and whose restriction to  $\mathrm{GL}_N(\mathcal{O}_F)$  is the inflation of  $\sigma$ .

### C Cuspidal representations of $\text{Sp}_{2n}(\mathbb{Q}_2)$ with irreducible Galois parameter (by Guy HENNIART at ORSAY)

#### C.1.

Let p be a prime number and F a finite extension of  $\mathbb{Q}_p$ . Let  $\overline{F}$  be an algebraic closure of Fand  $W_F$  the Weil group of  $\overline{F}/F$ . Let n be a positive integer, and  $\pi$  a cuspidal (complex) representation of  $\operatorname{Sp}_{2n}(F)$ . Let  $\sigma$  be the Galois parameter attached to  $\pi$  by Arthur [2], which one sees as an orthogonal representation of  $W_F \times \operatorname{SL}_2(\mathbb{C})$ , of dimension 2n + 1. The following result is used in the main text, in Section 9.

**Theorem C.1.** — Assume that  $F = \mathbb{Q}_2$ , and take for  $\pi$  the (unique) simple supercuspidal representation of  $\operatorname{Sp}_{2n}(F)$ . Then  $\sigma$  is an irreducible representation of  $W_F$ .

Here simple is in the sense of Gross and Reeder [22]. The point of the result is that  $\pi$  is compactly induced from a compact open subgroup of  $\operatorname{Sp}_{2n}(F)$ , as we describe below. Indeed when p = 2 there is at least one irreducible orthogonal representation  $\sigma$  of W<sub>F</sub> of dimension 2n + 1[11], only one if  $F = \mathbb{Q}_2$ , and by [2] it is the parameter of a cuspidal representation  $\pi$  of  $\operatorname{Sp}_{2n}(F)$ , but it is not clear a priori that  $\pi$  is compactly induced.

Our method is inspired by work of Oi [48]. When p is odd, Oi determines the parameter  $\sigma$  of a simple cuspidal representation  $\pi$  of  $\operatorname{Sp}_{2n}(F)$ . In his case  $\sigma$  is always reducible, but a number of techniques and results remain valid when p = 2, and, with extra information given by Adrian and Kaplan [1] when  $F = \mathbb{Q}_2$ , that is enough for us. It is quite likely that one can describe  $\sigma$ explicitly whenever  $\pi$  is simple cuspidal, not only when p is odd or  $F = \mathbb{Q}_2$ . Indeed many of our arguments work more generally, and until C.6 we make no special assumption on F, except that in C.3 we start assuming that<sup>(1)</sup> p = 2.

#### C.2.

We now proceed. We use customary notation,  $\mathcal{O}_F$  for the ring of integers of F,  $\mathfrak{p}_F$  for the maximal ideal of  $\mathcal{O}_F$ . We fix a uniformizer  $\varpi$  of F, and write k for the residue field  $\mathcal{O}_F/\mathfrak{p}_F$  and q for its cardinality. We also fix a non-trivial character  $\psi$  of k. If **H** is an algebraic group over F, we usually put  $H = \mathbf{H}(F)$ .

 $\mathbf{50}$ 

<sup>&</sup>lt;sup>(1)</sup>Oi and the author ([27]) can now extend Theorem C.1 to any 2-adic field F.

We use the usual explicit model of  $\mathbf{G} = \mathrm{Sp}_{2n}$ , see [48] §2.4, so elements of  $G = \mathrm{Sp}_{2n}(F)$  are symplectic  $2n \times 2n$  matrices. By cuspidal representation of G we mean an irreducible smooth complex cuspidal representation. We are interested in simple cuspidal representations of G, in the sense of Gross and Reeder [22]. Let us describe them.

The choice in [22] of a root basis and an affine root basis determines an Iwahori subgroup I of G, with its first two congruence subgroups  $I^+$  and  $I^{++}$ . The Iwahori subgroup I is the subgroup of  $\operatorname{Sp}_{2n}(\mathcal{O}_F)$  made out of the matrices which are upper triangular modulo  $\mathfrak{p}_F$ ,  $I^+$  is made out of the matrices which are further upper unipotent modulo  $\mathfrak{p}_F$ , and  $I^{++}$  is made out of the matrices  $(x_{i,j})$  in  $I^+$  with  $x_{i,i+1} \in \mathfrak{p}_F$  for  $i = 1, \ldots, 2n-1$ , and  $x_{2n,1} \in \mathfrak{p}_F^2$ . The quotient  $I^+/I^{++}$ is isomorphic to a product of n + 1 copies of k, via the surjective homomorphism

$$(x_{i,j}) \mapsto (x_{1,2} \mod \mathfrak{p}_F, \dots, x_{n,n+1} \mod \mathfrak{p}_F, x_{2n,1}/\varpi \mod \mathfrak{p}_F)$$

from  $I^+$  to  $k^{n+1}$ .

A character of  $I^+$  is simple if it is trivial on  $I^{++}$ , and is the inflation of a character of  $k^{n+1}$ which is non-trivial on each factor k. The normalizer in G of a simple character  $\theta$  of  $I^+$  is  $ZI^+$ . where Z is the centre of G, and  $ZI^+$  is also the intertwining of  $\theta$  in G, so that any extension of  $\theta$  to  $ZI^+$  gives by compact induction to G a cuspidal representation of G: see [48] §2.4, Proposition 2.6. Note that when p is 2, the centre Z of G is actually contained in  $I^{++}$ . The cuspidal representations of G thus obtained are the *simple cuspidal* representations of [22].

The normalizer of  $I^+$  in G is ZI, and I acts on  $I^+/I^{++}$  via  $I/I^+$ ; identifying  $I/I^+$  with  $k^{\times n}$ via

$$(x_{i,j}) \mapsto (x_{1,1} \mod \mathfrak{p}_F, \dots, x_{n,n} \mod \mathfrak{p}_F),$$

the conjugation action of  $(\chi_1, \ldots, \chi_n) \in k^{\times n}$  on  $I^+/I^{++}$  (identified with  $k^{n+1}$ ) sends the family  $(u_1, \ldots, u_{n+1}) \in k^{n+1}$  to

$$(u_1\chi_1\chi_2^{-1}, u_2\chi_2\chi_3^{-1}, \dots, u_{n-1}\chi_{n-1}\chi_n^{-1}, u_n\chi_n^2, u_{n+1}\chi_1^{-2}).$$

In particular, when p = 2, a given simple character  $\theta$  of  $I^+$  can always be conjugated in I to the character

$$\theta(a): (u_1, \dots, u_{n+1}) \mapsto \psi(u_1 + \dots + u_n + au_{n+1})$$

for some a in  $k^{\times}$ , uniquely determined by  $\theta$ . More precisely if  $\theta$  sends  $(u_1, \ldots, u_{n+1})$  to  $\psi(a_1u_1 + \cdots + a_{n+1})$  $\cdots + a_n u_n + a_{n+1} u_{n+1}$  for some  $a_i$ 's in  $k^{\times}$ , then a is equal to  $(a_1 \cdots a_{n-1})^2 \cdot a_n \cdot a_{n+1}$ . Thus<sup>(2)</sup> when p = 2 there are only q - 1 isomorphism classes of simple cuspidal representations of G, whereas, by a similar analysis ([48] §2.4), there are 4(q-1) such classes when p is odd. Note that when q = 2 all that is obvious since k has only one non-trivial character.

<sup>&</sup>lt;sup>(2)</sup>The referee remarks that one needs to know the "intertwining implies conjugacy" result that says that if two simple characters  $\theta$  and  $\theta'$  of  $I^+$  intertwine in G, then they are actually conjugate. The arguments are the same as for proving that the construction of simple cuspidals does give irreducible representations. The reader can consult [52] for the more general cases of epipelagic representations.

C.3.

The group  $\operatorname{Sp}_{2n}$  is split, and its dual group is  $\operatorname{SO}_{2n+1}(\mathbb{C})$ . To a cuspidal representation  $\pi$  of G, Arthur attaches the conjugacy class of a discrete parameter, that is (the conjugacy class of) a continuous homomorphism from  $W_F \times \operatorname{SL}_2(\mathbb{C})$  into  $\operatorname{SO}_{2n+1}(\mathbb{C})$  which, as a representation of dimension 2n + 1, is a direct sum of inequivalent irreducible orthogonal representations  $\sigma_1, \ldots, \sigma_r$ with the product det  $\sigma_1 \cdots \det \sigma_r$  trivial. What our theorem says is that when  $F = \mathbb{Q}_2$  and  $\pi$  is simple cuspidal, then r = 1 and  $\sigma_1$  is trivial on  $\operatorname{SL}_2(\mathbb{C})$ , *i.e.* is in fact a representation of  $W_F$ . Note that [11] shows that when p is odd, there is no irreducible orthogonal representation of  $W_F$  of odd dimension > 1, contrary to the case p = 2, where [11] gives a complete classification.

From now on we assume p = 2. For a in  $k^{\times}$  let us denote by  $\pi(a)$  the isomorphism class of the representation of G compactly induced from the character  $\theta(a)$  of  $I^+$ . We let  $\phi(a)$  be the parameter of  $\pi(a)$ , r(a) the number of irreducible components of  $\phi(a)$ , and  $\Pi(a)$  the L-packet of  $\pi(a)$ , that is the set of isomorphism classes of tempered (in fact, discrete series) representations of G with parameter  $\phi(a)$ ; it is known that  $\Pi(a)$  has  $2^{r(a)-1}$  elements, so one of our goals is to show that r(a) = 1. Let  $\mathbf{G}_{ad}$  be the adjoint group of  $\mathbf{G}$ , and  $\iota$  the quotient map from  $\mathbf{G}$  to  $\mathbf{G}_{ad}$ .

**Lemma C.2.** —  $\pi(a)$  is stable under the action of  $G_{ad}$ .

*Proof.* — We follow the proof of [48] Proposition 5.2. As there, one gets a description of the quotient  $G_{\rm ad}/\iota(G)$ . It is isomorphic to  $\operatorname{Hom}(F^{\times}, \mu_2)$ , itself isomorphic, by Kummer theory, to  $F^{\times}/F^{\times 2}$ . More concretely if T is the diagonal torus of G made out of elements

$$t(b) = (b, b, \dots, b, b^{-1}, \dots, b^{-1})$$

(with n times b and n times  $b^{-1}$ ), then for any b in  $\overline{F}$  with  $b^2$  in  $F^{\times}$  the image of t(b) in  $\mathbf{G}_{\mathrm{ad}}(\overline{F})$  is actually in  $G_{\mathrm{ad}}$ , and the set of such t(b)'s covers  $G_{\mathrm{ad}}/\iota(G)$ .

If  $b^2$  is a unit in F then t(b) actually normalizes I and its congruence subgroups, and sends  $\theta(a)$  to the character given by

$$(u_1, \dots, u_{n+1}) \mapsto \psi(u_1 + \dots + u_{n-1} + b^2 u_n + (a/b^2)u_{n+1})$$

conjugate in I to  $\theta(a)$ . If  $b^2$  is the uniformizer  $\varpi$ , t(b) conjugates I to another Iwahori subgroup, but if s is the matrix in G with four blocks of size n, first line  $(0, I_n)$  and second line  $(-I_n, 0)$ , then st(b) normalizes I and its congruence subgroups, and sends  $\theta(a)$  to the character given by

$$(u_1, \ldots, u_{n+1}) \mapsto \psi(u_1 + \cdots + u_{n-1} + au_n + u_{n+1})$$

(recall that p = 2, so -1 = 1 in k), which is conjugate to  $\theta(a)$ . Since the stabilizer in  $G_{ad}$  of  $\pi(a)$  is a subgroup containing all of  $\iota(G)$ , it follows that it is all of  $G_{ad}$ .

An important point is the genericity of simple cuspidal representations. We fix the same Whittaker datum as Oi [48] §6.3(2) to define genericity. By [30] Proposition 5.1, the  $G_{ad}$ -orbit of  $\pi(a)$  contains a single generic representation, so by the previous lemma the representation  $\pi(a)$  is generic. Reasoning as in [48] Corollary 4.9 and Corollary 5.7, we get:

**Proposition C.3.** — The parameter  $\phi(a)$  is trivial on  $SL_2(\mathbb{C})$ , every element of  $\Pi(a)$  is cuspidal, and among them only  $\pi(a)$  is a simple cuspidal representation.

 $\mathbf{52}$ 

It only remains to prove that r(a) = 1.

#### C.4.

Still following [48] we prove:

#### **Proposition C.4.** — $\Pi(a)$ does not contain any level 0 cuspidal representation.

*Proof.* — By [57] Corollary 9.10, all elements of  $\Pi(a)$  have the same formal degree. If dg is a Haar measure on G/Z, then the formal degree of  $\pi(a)$  is  $dg/vol(I^+/Z, dg)$  (by [48] Lemma 5.10), whereas the formal degree of a level 0 cuspidal representation of G is strictly smaller, by the following reasoning inspired by *loc. cit.*, Proposition 5.11. A level 0 cuspidal representation  $\pi'$  of G is compactly induced from an irreducible representation  $\rho$  of a maximal parahoric subgroup P of G, trivial on the pro-p radical  $P^+$  of P, and coming via inflation from a cuspidal representation of the finite (connected here) reductive group  $\overline{P} = P/P^+$ . The formal degree of  $\pi'$  is

$$\frac{\dim(\rho)}{\operatorname{vol}(P/Z,\mathrm{d}g)}\mathrm{d}g$$

One can assume that P contains I and  $I^+$  contains  $P^+$ . Since p = 2, the group  $P^+$  contains Z, so what we have to prove is that  $\dim(\rho) < \operatorname{card}(P/I^+)$ . But  $I^+/P^+$  is the unipotent radical  $\overline{U}$  of the Borel subgroup  $\overline{B} = I/P^+$  of  $\overline{P}$ , and obviously  $\dim(\rho)^2$  is at most  $\operatorname{card}(\overline{P})$ , so it is enough to check  $\operatorname{card}(\overline{P}) < \operatorname{card}(\overline{P}/\overline{U})^2$  or  $\operatorname{card}(\overline{U})^2 < \operatorname{card}(\overline{P})$ , which is a consequence of the existence of the big cell  $\overline{BwU}$  in the Bruhat decomposition for  $\overline{P}$ .

**Remark C.5.** — It is highly plausible that for a cuspidal representation  $\pi'$  of G which is not of level 0 and is not a simple cuspidal either, the formal degree of  $\pi'$  is bigger than the formal degree of  $\pi(a)$ . But nothing explicit is known about such  $\pi'$ .

#### C.5.

Now we compute the character  $\xi(a)$  of  $\pi(a)$  at an affine generic element g of  $I^+$ , where g generic means that modulo  $I^{++}$ , g gives an (n+1)-tuple  $(u_1, \ldots, u_{n+1})$  in  $k^{n+1}$  with all coordinates non-zero. As in [48] Lemma 2.5, we see that an element y conjugating g into  $I^+$  belongs to I, so that by the usual formula for the character of compactly induced representations (see e.g. *loc. cit.* Theorem 3.2), the character  $\xi(a)$  of  $\pi(a)$  at g is the sum

$$\sum_{(\chi_1,\dots,\chi_n)\in k^{\times n}}\psi\left(u_1\chi_1\chi_2^{-1}+u_2\chi_2\chi_3^{-1}+\dots+u_{n-1}\chi_{n-1}\chi_n^{-1}+u_n\chi_n^2+au_{n+1}\chi_1^{-2}\right),$$

which is a kind of Kloosterman sum, the sum

$$\sum_{(\eta_1,\dots,\eta_n)\in k^{\times n}}\psi\left(u_1\eta_1+u_2\eta_2+\dots+u_{n-1}\eta_{n-1}+u_n\eta_n^2+au_n\eta_{n+1}\right)$$

with  $\eta_{n+1}$  given by  $(\eta_1 \dots \eta_{n-1})^2 \eta_n \eta_{n+1} = 1$ . Noting that  $\psi$  takes only the values 1 and -1, we conclude:

**Proposition C.6.** — The value of  $\xi(a)$  at a generic element g of  $I^+$  is an odd integer depending only on g modulo  $I^{++}$ .

#### C.6.

Still following [48] 5.3, we now show:

**Proposition C.7.** — r(a) = 1 or 2, and, seen as a representation of  $W_F$  of dimension 2n + 1,  $\phi(a)$  is either irreducible or the direct sum of a character  $\omega$  with  $\omega^2 = 1$  and an irreducible (orthogonal) representation with determinant  $\omega$ .

Proof. — Put  $s = 2^{r(a)-1}$  and enumerate the elements of  $\pi(a)$  as  $\pi_1 = \pi(a), \ldots, \pi_s$ , and let  $\xi_i$  be the character of  $\pi_i$ . Let g be a generic element of  $I^+$ . Choose  $\varepsilon(i) = 1$  or -1 for  $i = 1, \ldots, s$ . Exactly as in the proof of Claim in *loc. cit.*, we get that  $\varepsilon(1)\xi_1 + \cdots + \varepsilon(s)\xi_s$  does not vanish at g. Using that the characteristic polynomial of g is irreducible of degree 2n (*loc. cit.*, Lemma 7.5, still valid when p = 2), the proofs of Theorem 5.1 and Corollary 5.13 in *loc. cit.* give the result.

#### C.7.

To get the remaining assertion that r(a) is in fact 1, we use new information given by Adrian and Kaplan [1]. Unfortunately that information is only available presently when  $F = \mathbb{Q}_2$ , hence the restriction in our main result, but we expect that the computation in *loc. cit.* can be carried over to the general case. When  $F = \mathbb{Q}_2$  there is only a = 1, so we put  $\pi = \pi(1)$ . In [1] Theorem 3.13, the authors compute the Rankin-Selberg  $\gamma$ -factor  $\gamma(\pi \times \tau, \psi')$  (a rational function in  $2^s$  for a complex parameter s) for any tame character  $\tau$  of  $\mathbb{Q}_2^{\times}$  with  $\tau^2 = 1$  and a character  $\psi'$  of  $\mathbb{Q}_2$ trivial on  $2\mathbb{Z}_2$  but not on  $\mathbb{Z}_2$ . They find

(C.1) 
$$\gamma(\pi \times \tau, \psi') = \tau(2)2^{1/2-s}$$

On the other hand if  $\phi$  is the parameter of  $\pi$ , seen as a representation of  $W_F$  of dimension 2n+1, and  $\lambda$  the character of  $W_F$  corresponding to  $\tau$  via class field theory, then

(C.2) 
$$\gamma(\pi \times \tau, \psi') = \gamma(\phi \otimes \lambda, \psi')$$

where the right-hand side is the Deligne-Langlands factor<sup>(3)</sup>.

That gives new information on  $\phi$  which, we recall, is by Proposition C.7 either irreducible or the direct sum of a character  $\omega$  with  $\omega^2 = 1$  and an irreducible representation, say  $\alpha$ , with

$$\omega \det \alpha = 1.$$

But the factor  $\gamma(\pi \times \tau, \psi')$  has no zero nor pole, so is equal to the factor  $\varepsilon(\pi \times \tau, \psi') = \varepsilon(\phi \otimes \lambda, \psi')$ which has the form

$$u \cdot 2^{\operatorname{Art}(\phi \otimes \lambda) - \dim(\phi \otimes \lambda))(1/2 - s)}$$

for some non-zero complex number u: the exact value of the exponent comes from the fact that  $\psi'$  is trivial on  $2\mathbb{Z}_2$  but not on  $\mathbb{Z}_2$ . This implies that  $\operatorname{Art}(\phi \otimes \lambda) = 2n + 2$ , and taking  $\lambda$  trivial yields  $\operatorname{Art}(\phi) = 2n + 2$ .

Assume we are in the case where  $\phi = \omega \oplus \alpha$ . Taking  $\lambda = \omega$  gives a pole to  $\gamma(\phi \otimes \omega, \psi)$  which contradicts (C.1) if  $\omega$  is tame (that is, since  $F = \mathbb{Q}_2$ , unramified). Thus  $\omega$  is wildly ramified, so

 $\mathbf{54}$ 

<sup>&</sup>lt;sup>(3)</sup>That is to say, Arthur's correspondence is compatible with Rankin-Selberg  $\gamma$ -factors. It can be proved by a local-global argument. Detail will appear in joint work with Oi [27].

its Artin exponent is at least 2, and the Artin exponent of  $\alpha$  is at most 2*n*. That implies that  $\alpha$  is tamely ramified, and in fact  $\operatorname{Art}(\alpha) = 2n$ ,  $\operatorname{Art}(\omega) = 2$ . But then det  $\alpha$  is also tamely ramified, which contradicts  $\omega \det \alpha = 1$ . That contradiction shows that  $\phi$  is irreducible, as desired.

#### C.8.

One can describe  $\phi$  explicitly. By the main result of [11] an orthogonal irreducible representation of  $W_{\mathbb{Q}_2}$  is induced from an order 2 wildly ramified character  $\beta$  of  $W_K$ , where K is a totally ramified extension of  $\mathbb{Q}_2$  degree 2n + 1. Such an extension is unique up to isomorphism, generated by a uniformizer z with  $z^{2n+1} = 2$ . Let  $\tilde{\beta}$  be the character of  $K^{\times}$  corresponding to  $\beta$  via class field theory. Since  $\operatorname{Art}(\phi) = 2n + 2$ , we have  $\operatorname{Art}(\tilde{\beta}) = 2$ , and moreover  $\det(\phi) = 1$  is the restriction of  $\tilde{\beta}$  to  $\mathbb{Q}_2^{\times}$  times the determinant of the representation of  $W_{\mathbb{Q}_2}$  induced from the trivial character of  $W_K$ . That determinant is an unramified quadratic character of  $W_K$ , computed in [9] as the unramified character taking value at Frobenius elements the Jacobi symbol of 2 modulo 2n + 1. That imposes  $\tilde{\beta}(z)$ , and with  $\operatorname{Art}(\tilde{\beta}) = 2$  and  $\tilde{\beta}(1+z) = -1$  it determines  $\tilde{\beta}$  hence  $\beta$ .

#### References

- M. Adrian and E. Kaplan, The Langlands parameter of a simple supercuspidal representation: symplectic groups, Ramanujan J. 50 (2019), n°3, 589–619.
- J. Arthur, The endoscopic classification of representations. Orthogonal and symplectic groups, American Mathematical Society Colloquium Publications 61, American Mathematical Society, Providence, RI, 2013.
- 3. J. Arthur and L. Clozel, Simple algebras, base change, and the advanced theory of the trace formula, Annals of Mathematics Studies **120**, Princeton University Press, Princeton, NJ, 1989.
- H. Atobe and W. T. Gan, On the local Langlands correspondence and Arthur conjecture for even orthogonal groups, Represent. Theory 21 (2017), 354–415.
- T. Barnet-Lamb, D. Geraghty, M. Harris and R. Taylor, A family of Calabi-Yau varieties and potential automorphy II, Publ. Res. Inst. Math. Sci. 47 (2011), n°1, 29–98.
- A. Borel, Automorphic L-functions, in Automorphic forms, representations and L-functions, Proc. Sympos. Pure Math. 33, Part 2, 27–61, Amer. Math. Soc., Providence, R.I., 1979.
- Linear algebraic groups, Second edition, Graduate Texts in Mathematics 126, Springer-Verlag, New York, 1991.
- 8. N. Bourbaki, Éléments de mathématique, Algèbre commutative, Chap. 5 à 7, Hermann, Paris, 1975.
- C. Bushnell and A. FrAPhlich, Gauss sums and p-adic division algebras, Lecture Notes in Mathematics 987, Springer-Verlag, New York, 1983.
- C. Bushnell and G. Henniart, *The local Langlands conjecture for* GL(2), Grundlehren der mathematischen Wissenschaften **335**, Springer-Verlag, Berlin, 2006.
- 11. \_\_\_\_\_, Self-dual representations of some dyadic groups, Math. Ann. 351 (2011), n°1, 67–80.
- 12. \_\_\_\_\_, Modular local Langlands correspondence for  $GL_n$ , Int. Math. Res. Not. 15 (2014), 4124–4145.
- C. Bushnell and P. Kutzko, The admissible dual of GL(N) via compact open subgroups, Annals of Mathematics Studies 129, Princeton University Press, Princeton, NJ, 1993.
- L. Clozel, Motifs et formes automorphes: applications du principe de fonctorialité, in Automorphic forms, Shimura varieties, and L-functions, Vol. I, 77–159, Perspect. Math. 10, Academic Press, Boston, MA, 1990.

#### ALBERTO MÍNGUEZ & VINCENT SÉCHERRE

- J. F. Dat, ν-tempered representations of p-adic groups. I. l-adic case, Duke Math. J. 126 (2005), n°3, 397–469.
- 16. \_\_\_\_\_, Théorie de Lubin-Tate non-abélienne l-entière, Duke Math. J. 161 (2012), n°6, 951–1010.
- 17. \_\_\_\_\_, Un cas simple de correspondance de Jacquet-Langlands modulo ℓ, Proc. London Math. Soc. 104 (2012), 690–727.
- J. F. Dat, D. Helm, R. Kurinczuk and G. Moss, *Finiteness for Hecke algebras of p-adic group*, J. Amer. Math. Soc. (2023).
- P. Deligne and J.-P. Serre, Formes modulaires de poids 1, Ann. Sci. Ecole Norm. Sup. (4) 7 (1974), 507–530 (1975).
- J. Green, The characters of the finite general linear groups, Trans. Amer. Math. Soc. 80 (1955), 402–447.
- 21. B. Gross, Algebraic modular forms, Israel J. Math. 113 (1999), 61–93.
- B. Gross and M. Reeder, Arithmetic invariants of discrete Langlands parameters, Duke Math. J. 154 (2010), n°3, 431–508.
- M. Harris and R. Taylor, *The geometry and cohomology of some simple Shimura varieties*, with an appendix by Vladimir G. Berkovich, Annals of Mathematics Studies 151, Princeton University Press, Princeton, NJ, 2001.
- G. Henniart, La conjecture de Langlands locale pour GL(3), Mém. Soc. Math. France (N.S.) n°11-12 (1984).
- 25. \_\_\_\_\_, Une preuve simple des conjectures de Langlands pour GL(n) sur un corps p-adique, Invent. Math. **139** (2000), n°2, 439–455.
- 26. \_\_\_\_, Sur la conjecture de Langlands locale pour GL(n), J. Théor. Nombres Bordeaux 13 (2001), n°1, 167–187.
- G. Henniart and M. Oi, Simple supercuspidal L-packets of symplectic groups over dyadic fields, arXiv:2207.12985
- G. Henniart and M.-F. Vignéras, A Satake isomorphism for representations modulo p of reductive groups over local fields, J. Reine Angew. Math. 701 (2015), 33–75.
- 29. N. Jacobson, Basic algebra II, Second edition, W. H. Freeman and Company, New York, 1989.
- 30. T. Kaletha, Simple wild L-packets, J. Inst. Math. Jussieu 12 (2013), n°1, 43-75.
- T. Kaletha, A. Mínguez, S. W. Shin and P.-J. White, Endoscopic Classification of Representations: Inner Forms of Unitary Groups, arXiv:1409.3731
- 32. C. Khare, On the local Langlands correspondence mod l, J. Number Theory 88 (2001), n°2, 357–365.
- M.-A. Knus, A. Merkurjev, M. Rost and J.-P. Tignol, *The book of involutions*, American Mathematical Society Colloquium Publications 44, American Mathematical Society, Providence, RI, 1998.
- R. Kurinczuk, D. Skodlerack and S. Stevens, Endoparameters for p-adic classical groups, Invent. Math. 223 (2021), n° 2, 597–723.
- R. Kurinczuk and S. Stevens, Cuspidal l-modular representations of p-adic classical groups, J. Reine Angew. Math. 764 (2020), 23–69.
- 36. J.-P. Labesse, Changement de base CM et séries discrètes, in On the stabilization of the trace formula, 429–470, Int. Press, Somerville, MA, 2011.
- J. Lust and S. Stevens, On depth zero L-packets for classical groups, Proc. Lond. Math. Soc. (3) 121 (2020), n°5, 1083–1120.
- A. Mínguez, Unramified representations of unitary groups, in On the stabilization of the trace formula, 389–410, Int. Press, Somerville, MA, 2011.
- 39. A. Mínguez and V. Sécherre, *Représentations lisses modulo*  $\ell$  *de*  $GL_m(D)$ , Duke Math. J. **163** (2014), 795–887.

- 40. \_\_\_\_\_, Classification des représentations modulaires de  $GL_n(q)$  en caractéristique non naturelle, Trends in number theory, 163–183, Contemp. Math. 649, Amer. Math. Soc., Providence, RI, 2015.
- 41. \_\_\_\_\_, Correspondance de Jacquet-Langlands locale et congruences modulo ℓ, Invent. Math. 208 (2017), n°2, 553-631.
- M. Miyauchi and S. Stevens, Semisimple types for p-adic classical groups, Math. Ann. 358 (2014), n°1-2, 257–288.
- C. Mœglin, Comparaison des paramètres de Langlands et des exposants à l'intérieur d'un paquet d'Arthur, J. Lie Theory 19 (2009), n°4, 797–840.
- 44. \_\_\_\_\_, Multiplicité 1 dans les paquets d'Arthur aux places p-adiques, in On certain L-functions, 333–374, Clay Math. Proc. 13, Amer. Math. Soc., Providence, RI, 2011.
- 45. \_\_\_\_\_, Paquets stables des séries discrètes accessibles par endoscopie tordue; leur paramètre de Langlands, in Automorphic forms and related geometry: assessing the legacy of I. I. Piatetski-Shapiro, 295–336, Contemp. Math. 614, Amer. Math. Soc., Providence, RI, 2014.
- 46. C. Mœglin, J.-L. Waldspurger and M.-F. Vignéras, *Correspondances de Howe sur un corps p-adique*, Lecture Notes in Mathematics **1291**, Springer-Verlag, Berlin, 1987.
- C. P. Mok, Endoscopic classification of representations of quasi-split unitary groups, Mem. Amer. Math. Soc. 235 (2015), n°1108.
- 48. M. Oi, Simple supercuspidal L-packets of quasi-split classical groups, arXiv:1805.01400, to appear in Mem. Amer. Math. Soc.
- 49. O. T. O'Meara, *Hilbert's Reciprocity Law*, in *Introduction to Quadratic Forms*, Classics in Mathematics **117**, Springer, Berlin, Heidelberg.
- V. Platonov and A. Rapinchuk, Algebraic groups and number theory, Pure and Applied Mathematics 139, Academic Press, Inc., Boston, MA, 1994.
- 51. A. Robert, A course in p-adic analysis, Graduate Texts in Mathematics 198, Springer, Berlin, 2000.
- 52. B. Romano, On input and Langlands parameters for epipelagic representations, Representation Theory 28 (2024), pp. 90 – 111.
- I. Satake, Theory of spherical functions on reductive algebraic groups over p-adic fields, Publ. Math. Inst. Hautes Études Sci. 18 (1963), 5–69.
- W. Scharlau, *Quadratic and Hermitian forms*, Grundlehren der mathematischen Wissenschaften 270, Springer-Verlag, Berlin, 1985.
- 55. J.-P. Serre, Cours d'arithmétique, Presses Universitaires de France, Paris, 1977.
- 56. J.-P. Serre and J. Tate, Good reduction of abelian varieties, Ann. of Math. (2) 88 (1968), 492–517.
- F. Shahidi, A proof of Langlands' conjecture on Plancherel measures; complementary series for p-adic groups, Ann. of Math. (2) 132 (1990), n°2, 273–330.
- 58. S. W. Shin, Automorphic Plancherel density theorem, Israel J. Math. 192 (2012), n°1, 83–120.
- 59. T. Shintani, On an explicit formula for class-1 Whittaker functions on  $GL_n$  over  $\mathfrak{p}$ -adic fields, Proc. Japan Acad. **52** (1976), 180–182.
- S. Stevens, The supercuspidal representations of p-adic classical groups, Invent. Math. 172 (2008), n°2, 289–352.
- J. Tate, Number theoretic background, in Automorphic forms, representations and L-functions, Proc. Sympos. Pure Math. 33, Part 2, 3–26, Amer. Math. Soc., Providence, R.I., 1979.
- O. Taïbi, Eigenvarieties for classical groups and complex conjugations in Galois representations, Math. Res. Lett. 23 (2016), n°4, 1167–1220.
- 63. \_\_\_\_\_, Arthur's multiplicity formula for certain inner forms of special orthogonal and symplectic groups, J. Eur. Math. Soc. **21** (2019), n°3, 839–871.
- 64. M.-F. Vignéras, Représentations modulaires de GL(2, F) en caractéristique l, F corps p-adique,  $p \neq l$ , Compositio Math. **72** (1989), n°1, 33–66.

- 65. \_\_\_\_\_, Représentations l-modulaires d'un groupe réductif p-adique avec  $l \neq p$ , Progress in Mathematics **137**, Birkhäuser Boston Inc., Boston, MA, 1996.
- 66. \_\_\_\_\_, Correspondance de Langlands semi-simple pour  $\operatorname{GL}_n(F)$  modulo  $\ell \neq p$ , Invent. Math 144 (2001), 177–223.
- 67. \_\_\_\_\_, Induced R-representations of p-adic reductive group, Selecta Math. (N.S.) 4 (1998), n°4, 549–623. With an appendix by Alberto Arabia.
- 68. \_\_\_\_\_, On highest Whittaker models and integral structures, Contributions to Automorphic forms, Geometry and Number theory: Shalikafest 2002, John Hopkins Univ. Press, 2004, 773–801.
- B. Xu, On Moeglin's parametrization of Arthur packets for p-adic quasisplit Sp(N) and SO(N), Canad. J. Math. 69 (2017), n°4, 890–960.
- 70. \_\_\_\_\_, On the cuspidal support of discrete series for p-adic quasisplit Sp(N) and SO(N), Manuscripta Math. **154** (2017), n°3-4, 441–502.
- 71. J. Zou, Représentations supercuspidales de GL(n) sur un corps local non archimédien : distinction par un sous-groupe unitaire ou orthogonal, changement de base et induction automorphe, thèse de l'université Paris-Saclay, 2021.

ALBERTO MÍNGUEZ, University of Vienna, Fakultät für Mathematik, Oskar-Morgenstern-Platz 1, 1090 Wien *E-mail* : alberto.minguez@univie.ac.at

VINCENT SÉCHERRE, Laboratoire de Mathématiques de Versailles, UVSQ, CNRS, Université Paris-Saclay, 78035 Versailles, France, Institut Universitaire de France • *E-mail* : vincent.secherre@uvsq.fr